MULTICHIP MODULE TECHNOLOGIES AND ALTERNATIVES:

THE BASICS

Elm Exhibit 2162 Samsung, Micron, SK hynix v. Elm IPR2016-00387

Cover Photo

Courtesy of Unisys photographer, Paul Robinson. The top MCM is a hermetic processor module utilizing the *latest* MCM-D technology with fine pitch, high lead count, flip TAB connections. The bottom three packages are MCM-C modules with conventional wire bond connections. These modules are described in Chapter 14.

MULTICHIP MODULE TECHNOLOGIES AND ALTERNATIVES:

THE BASICS

Edited by: Daryl Ann Doane Paul D. Franzon



msgalica@mintz.com

Copyright © 1993 by Springer Science+Business Media New York Originally published by Van Nostrand Reinhold in 1993 Softcover reprint of the hardcover 1st edition 1993 Library of Congress Catalog Card Number 92-2779 ISBN 978-0-442-01236-6

All rights reserved. No part of this work covered by the copyright hereon may be reproduced or used in any form by any means—graphic, electronic, or mechanical, including photocopying, recording, taping, or information storage and retrieval systems—without written permission of the publisher.

I TP Van Nostrand Reinhhold is a division of International Thomson Publishing. ITP logo is a trademark under license.

16 15 14 13 12 11 10 9 8 7 6 5 4 3 2

Library of Congress Cataloging-in-Publication Data

Multichip module technologies and alternatives : the basics / [edited] by Daryl Ann Doane and Paul D. Franzon.

p. cm. Includes Index. ISBN 978-0-442-01236-6 ISBN 978-1-4615-3100-5 (eBook) DOI 10.1007/978-1-4615-3100-5 1. Microelectronic packaging. 2. Integrated circuits — Very large scale integration — Design and construction. I. Doane, Daryl Ann. II. Franzon, Paul D. TK7874.M864 1993 621.39'5 — dc20 92-2779

Elm Exhibit 2162, Page 4

msgalica@mintz.com

DEDICATION

This book is dedicated with sincere gratitude to the 42 authors who contributed their knowledge, suggestions and many hours of writing to form the chapters we requested. Without their diligence, timeliness of response and patience with our requests for additional or alternative information, there would be no book.

ABOUT THE EDITORS

Daryl Ann Doane, President, DAD Technologies, Inc., is a specialist in the manufacturing of IC devices and packages, and in the marketing of materials, equipment and processes for such manufacturing. She has published more than 40 reports, papers and reviews documenting new process development and yield optimization. DAD Technologies, Inc. is a technology-based consulting organization serving both producers and users. Clients have included both government and commercial organizations connected with IC fabrication and packaging.

Dr. Doane previously served as Director, Microelectronics R&D for Hunt Chemical (now Olin Hunt Specialty Products). Prior to that she was a Member of the Technical Staffs of AT&T Bell Laboratories, RCA and Hewlett-Packard Laboratories.

Dr. Doane is a member of the American Physical Society (APS), Materials Research Society (MRS), International Society for Hybrid Microelectronics (ISHM), International Electronics Packaging Society (IEPS), and is a Senior Member of the IEEE. She received the 1980 Distinguished New Engineer Award from the Society of Women Engineers. She also served as President of the New Jersey Section of that Society.

Dr. Doane presently serves as Director of Technical Marketing for the IEEE-CHMT (Components, Hybrids & Manufacturing Technology) Society. She also is an elected member of the Board of Governors of the National CHMT Society. She served previously as Chairman of the San Diego Chapter of the CHMT Society.

Dr. Doane received the BA with Distinction in Chemistry and Physics, an MA in Chemistry, and an MS and PhD in Metallurgy and Materials Science from M.I.T. and the University of Pennsylvania, respectively.

Paul D. Franzon is currently an Assistant Professor in the Department of Electrical and Computer Engineering at North Carolina State University. He has over eight years experience in electronic systems design and design methodology research and development. During that time, in addition to his current position, he has worked at AT&T Bell Laboratories in Holmdel NJ, at the Australian Defense Science and Technology Organization, as a founding member of a successful Australian technology start-up company, and as a consultant to industry, including positions on Technical Advisory Boards.

Dr. Franzon's current research interests include design sciences/methodology for high speed packaging and interconnections systems and also for high speed and low power chip design. In the past, he has worked on problems and projects in wafer scale integration, IC yield modeling, VLSI chip design and communications systems design. He has published over 45 articles and reports.

His teaching interests focus on microelectronic systems building including package and interconnection design, circuit design, processor design and the gaining of hands-on systems experience for students.

Dr. Franzon is a member of the IEEE, ACM and ISHM. He serves as the Chairman of the Education Committee for the National IEEE-CHMT Society.

Dr. Franzon received a BS in Physics and Mathematics, a BE with First Class Honors in Electrical Engineering and a PhD in Electrical Engineering all from the University of Adelaide, Adelaide, Australia.

TABLE OF CONTENTS

Fo	preword	xxiii
Preface		xxv
Ac	knowledgments	xxxi
Pa	urt A: The Framework	1
1	Introduction	3
	1.1 Background and Definitions	3
	1.1.1 Purpose and Perspective of the Book	3
	1.1.2 Architecture - Building a	
	Multichip Module Structure	5
	1.1.3 First Level Connection and Common	
	Circuit Base Alternatives	9
	1.1.4 MCM Packaging Alternatives	11
	1.2 Finding Your Way	15
	1.3 The Importance of Materials	17
	1.4 The Importance of Manufacturing Processes	21
	1.5 The Importance of Industry Infrastructure	22

X CONTENTS

	1.6 Decision-Making as a Process	25
	1.6.1 Determining the Application:	
	Possible MCM Markets	26
	1.6.2 Determining the MCM Technology:	
	Business Decisions	28
	1.6.3 Designing the Product:	
	Multidisciplinary Engineering	30
	1.7 Overall Prospects for MCMs	30
	1.8 Summary	32
	General References	33
	References	35
2	MCM Package Selection:	
	A Materials and Manufacturing Perspective	37
	2.1 Introduction	37
	2.2 Package Body and Substrate Base Choices	40
	2.2.1 Ceramics	40
	2.2.2 Organic Laminates	43
	2.2.3 Metals	43
	2.3 Signal Interconnect and MCM Substrate Ch	01ces 43
	2.3.1 MCM-C (Cotified Ceramics)	45
	2.3.2 MCM-L (Organic Laminates)	48
	2.3.3 MCM-D (Deposited Dielectric)	51
	2.5.4 MCM-D/C (Deposited Dielectric	53
	2.3.5 MCM Si (Inorganic Thin Film)	54
	2.3.5 MCM-SI (morganic Timi Finit) 2.3.6 Thick Film Hybrid MCM	55
	2.5.0 Thick Find Hybrid Meth	56
	2.4 Chip Mounting Choices	58
	2.4.2. TAB	50 60
	2.4.3 Flip TAB	63
	2.4.4 Flip Chin	63
	2.5 Module Level Connection Choices	66
	2.5.1 Peripheral I/O	66
	2.5.2 Pin Grid Array	67
	2.5.3 Pad Array Carrier (PAC)	68
	2.6 Global Materials and Manufacturing Consid	lerations 69
	2.6.1 Cost	69
	2.6.2 Electrical Performance	70
	2.6.3 Thermal Path	75
	2.6.4 Rework	76

	2.7 Module Design Examples	76
	2.7.1 Mainframe Computer Module	76
	2.7.2 Workstation Module	77
	2.7.3 Low Cost Module	78
	2.7.4 Data Communications Module	78
	2.8 Future Trends in MCM Materials Manufacturing	81
	2.8.1 Substrates	82
	2.8.2 Optical Multichip Modules (MCM-O)	83
	2.8.3 Test	83
	2.8.4 Thermal Control	83
	2.8.5 Environmental Concerns	84
	References	85
3	MCM Package Selection: A Systems Need Perspective	87
	3.1 Introduction	87
	3.2 System Design Process	88
	3.3 The Packaging Hierarchy	
	3.4 Packaging Performance Factors	
	3.4.1 Size and Weight	93
	3.4.2 Interconnection Capacity Within Each Level	94
	3.4.3 Connection Capacity Between Packaging Levels	100
	3.4.4 Delay and Electrical Noise	102
	3.4.5 Power Consumption	107
	3.4.6 Heat Dissipation	109
	3.4.7 Performance Tradeoffs	111
	3.5 Packaging Cost Factors	112
	3.5.1 Production Cost	112
	3.5.2 Post Production Costs	117
	3.5.3 Design and Prototyping Costs	118
	3.5.4 Time-to-Market	119
	3.5.5 Cost Tradeoffs	120
	3.6 Packaging Decisions and the System Design Process	
	3.7 Determining System Requirements and Goals	
	3.8 Determining and Evaluating Packaging Alternatives	
	3.9 Impact of Semiconductor Technology	
	3.10 Example of the System Design Process	124
	3.11 Summary	128
	References	130

xii CONTENTS

4	MC	M Package Selection: Cost	Issues	133
	4.1	Introduction		133
		4.1.1 The Importance of Co	ost	134
	4.2	Techniques for Cost Analys	is	135
		4.2.1 Traditional Cost Anal	ysis	135
		4.2.2 Activity-Based Cost A	Analysis	137
		4.2.3 Technical Cost Mode	ling	138
	4.3	Technical Cost Modeling		139
		4.3.1 Principles of Technica	al Cost Modeling	139
		4.3.2 Applications of Techn	nical Cost Modeling	141
	4.4	Results of Technical Cost M	Iodeling	143
		4.4.1 Printed Wiring Board	S	143
		4.4.2 Thick Film Substrates	\$	146
		4.4.3 Cofired Multilayer Su	bstrates	148
		4.4.4 Thin Film MCMs		150
		4.4.5 Cost of MCM Assem	bly	151
	4.5	Applications to Substrate Se	election	153
	4.6	Design Activity-Based Cost	Modeling	157
	4.7	Summary		161
	Ref	erences		163
Pa	rt B:	The Basics		165
5	Lai	ninate-Based Technologies	for Multichip Modules	169
	5.1	Introduction	-	169
		5.1.1 MCM-L Amid the Sp	ectrum of	
		MCM Substrate Tech	nologies	170
		5.1.2 MCM-L Attributes		171
	5.2	Standard MCM-L Construct	ion Process	172
		5.2.1 Dielectric Material		174
		5.2.2 Copper Foil (Conduct	or) Processing	174
		5.2.3 Inner Layer Photolith	ographic Processing	174
		5.2.4 Blind and Buried Via	Formation	175
		5.2.5 Lamination		176
		5.2.6 Drilling		176
		5.2.7 Plating of Drilled Ho	les	178
		5.2.8 Processing of Surface	Layers	178
	5.3	Material Considerations		180
		5.3.1 Dielectric Layers		181
		5.3.2 Polymers for Dielectr	ic Layers	185
		533 Reinforcements for D	ielectric Lavers	186

	5.3.4 Copper Conductors	186
	5.4 Flexible ("Flex") Circuits	187
	5.4.1 Flex Circuit and Connector Integration	188
	5.5 Advanced MCM-L Material and Process Technology	189
	5.5.1 Integral Termination Resistor Technology	189
	5.5.2 Additive Conductor Processing	190
	5.5.3 Advanced Reinforcement, Rigidifying Materials	192
	5.6 IC Connection and Repair: Laminate Implications	195
	5.6.1 Wire Bonding to MCM-L Substrates	196
	5.6.2 TAB Bonding on MCM-L Substrates	197
	5.6.3 Flip Chip Bonding on MCM-L Substrates	197
	5.7 Versions of Assembled MCM-L Systems	198
	5.7.1 Substrate I/O Configurations	198
	5.8 MCM-L Extendibility and Cost Issues	209
	References	211
_		
6	Thick Film and Ceramic Technologies	
	for Hybrid Multichip Modules	215
	6.1 Introduction	215
	6.1.1 Definitions of MCM-C Technologies	215
	6.1.2 General Comparison of MCM-C Technologies	219
	6.2 Material Considerations	225
	6.2.1 Thick Film Technology	225
	6.2.2 High Temperature Coffred Ceramic Technology	234
	6.2.3 Low Temperature Confed Ceramic Technology	235
	6.2.1 Thick Films	240
	6.3.2 High Temperature Cofired Ceromics	240
	6.3.3 Low Temperature Cofficed Ceramics	242
	6.4 Design Rules	243
	6.5 Applications	244
	6.6 Future Trends	243
	6.7 Engineering Choices	240
	References	253
		255
7	Thin Film Multilayer Interconnection	
	Technologies for Multichip Modules	255
	7.1 Introduction	255
	7.2 Characteristics, Benefits of TFML Interconnections	256
	7.2.1 Packaging Structures Using	
	TFML Interconnections	257

		7.2.2 Signal Line Characteristics	259
		7.2.3 Interconnect Design Rules	262
		7.2.4 Comparison With Alternative	
		Interconnection Technologies	264
	7.3	Materials for Thin Film Interconnection Systems	267
		7.3.1 Substrate Materials	267
		7.3.2 Conductor Materials	272
		7.3.3 Dielectric Materials	274
	7.4	Thin Film Multilayer Processing	276
		7.4.1 Conductor Deposition and Patterning Processes	277
		7.4.2 Dielectric Deposition and Patterning Processes	282
		7.4.3 Basic Process Approaches: Additive and Subtractive	289
		7.4.4 Unique Process Requirements	294
	7.5	Design Strategies for Thin Film Interconnections	298
		7.5.1. General Design Strategies	298
		7.5.2 Implementations of Thin Film Interconnections	300
	7.6	Applications, Growth of Thin Film Interconnections	304
		7.6.1 Interconnection Technology Selection	304
		7.6.2 Evolution of TFML Applications	305
		7.6.3 Future Applications	306
	Ref	erences	307
8	Sel	ection Criteria For Multichip Module Dielectrics	311
	8.1	Introduction	311
	8.2	Behavior and Function of Dielectrics	312
	8.3	Multilevel Thin Film Structures	314
		8.3.1 Major Technical Challenges	
		• •	314
		8.3.2 MCM Fabrication Processes	314 316
		8.3.2 MCM Fabrication Processes8.3.3 Dielectric Processing	314 316 318
	8.4	8.3.2 MCM Fabrication Processes8.3.3 Dielectric ProcessingPolymer Properties	314 316 318 319
	8.4	8.3.2 MCM Fabrication Processes8.3.3 Dielectric ProcessingPolymer Properties8.4.1 Some Polymer Specific Terms	314 316 318 319 319
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 	314 316 318 319 319 320
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 	314 316 318 319 319 320 321
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 	314 316 318 319 319 320 321 322
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 	314 316 318 319 319 320 321 322 323
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 	314 316 318 319 319 320 321 322 323 326
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 8.4.7 Failure Mechanisms 	314 316 318 319 319 320 321 322 323 326 329
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 8.4.7 Failure Mechanisms 8.4.8 Chemical Resistance 	314 316 318 319 320 321 322 323 326 329 331
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 8.4.7 Failure Mechanisms 8.4.8 Chemical Resistance 8.4.9 Adhesion 	314 316 318 319 320 321 322 323 326 329 331 332
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 8.4.7 Failure Mechanisms 8.4.8 Chemical Resistance 8.4.9 Adhesion 8.4.10 Water Uptake 	314 316 318 319 320 321 322 323 326 329 331 332 334
	8.4	 8.3.2 MCM Fabrication Processes 8.3.3 Dielectric Processing Polymer Properties 8.4.1 Some Polymer Specific Terms 8.4.2 Dielectric Constant 8.4.3 Thermal Stability 8.4.4 The Glass Transition Temperature 8.4.5 Coefficient of Thermal Expansion 8.4.6 Mechanical Properties 8.4.7 Failure Mechanisms 8.4.8 Chemical Resistance 8.4.9 Adhesion 8.4.10 Water Uptake 8.4.11 Planarization 	314 316 318 319 320 321 322 323 326 329 331 332 334 335

	8.5 Polymer Materials	337
	8.5.1 Polyimides	338
	8.5.2 Polyphenylquinoxaline	346
	8.5.3 Fluorocarbons	346
	8.5.4 Benzocyclobutenes	346
	8.6 Summary	346
	References	347
9	Chip-to-Substrate (First Level) Connection	
	Technology Options	349
	9.1 Introduction	349
	9.1.1 Chip Connection Technologies	350
	9.1.2 Electrical Design	351
	9.1.3 Mechanical Design	351
	9.1.4 Technology Comparisons	352
	9.2 Die Bonding and Physical Attachment	354
	9.2.1 Die Attach Material Choices	354
	9.2.2 Die Attach Processes And Process Control	358
	9.2.3 Die Attach Equipment	363
	9.2.4 Issues for MCMs	363
	9.3 Wire Bonding	368
	9.3.1 Wire Bonding Methods and Procedures	368
	9.3.2 Wire Bonding Processes and Configurations:	
	Geometry, Density and Package Design	
	Considerations	375
	9.3.3 Wire Bonding Equipment	379
	9.3.4 Electrical Performance	384
	9.3.5 Reliability as Applied to MCMs	385
	9.3.6 Yield and Repairability as Applied to MCMs	386
	9.3.7 Wire Bond Process Development	387
	9.3.8 Wire Bond Process Costs	389
	9.3.9 Comparison to Other Connection Techniques	391
	9.3.10 Summary	392
	9.4 Tape Automated Bonding	394
	9.4.1 Introduction	394
	9.4.2 Basic Process Flow for TAB Packaging	396
	9.4.3 TAB Tape Considerations	401
	9.4.4 Inner Lead Bonding	406
	9.4.5 Outer Lead Bonding	410
	9.4.6 Single Chip and MCM Implementations of TAB	416
	9.4.7 Thermal and Electrical Performance	418

	9.4.8 Reliability	420
	9.4.9 Reworkability	423
	9.4.10 Manufacturing Issues and Costs	425
	9.4.11 Comparison with Other Connection Technologies	427
	9.4.12 Summary	428
	9.5 Flip Chip Connection Technology	429
	9.5.1 Introduction	429
	9.5.2 The Basics	429
	9.5.3 Connection Medium (I): Solder Bumps	434
	9.5.4 Connection Medium (II): Conductive Polymers	446
	9.5.5 The Whole Picture	448
	9.6 Flip Chip Solder Bump (FCSB) Technology:	
	An Example	450
	9.6.1 Introduction	450
	9.6.2 Fabrication, Process Flow, Tools and Hardware	451
	9.6.3 Manufacturability	457
	9.6.4 Rework	459
	9.6.5 Reliability	462
	9.6.6 Performance	473
	9.6.7 Summary	476
	9.7 Summary	477
	9.7.1 Cost	477
	9.7.2 Performance	478
	9.7.3 Tradeoffs: Making a Choice	480
	References	481
10	MCM-to-Printed Wiring Board (Second Level) Connection	
	Technology Options	487
	10.1 Introduction	487
	10.1.1 Second Level Connection Alternatives	488
	10.2 Basic Issues Affecting Connection Choice	491
	10.2.1 Mechanical and Materials Issues with	
	Permanent Soldered Connections	491
	10.2.2 Separable Connector Interface Physics	492
	10.2.3 Electrical Performance Issues	497
	10.3 Basic Approaches to MCM Level Two Connections	499
	10.3.1 Direct Attachment Through Soldering	500
	10.3.2 Chip-on-Board Connections	502
	10.3.3 Separable Connections (Sockets)	502
	10.3.4 Sockets for Leadless MCM Substrates	503
	10.3.5 Sockets for Leaded MCM Substrates	513

	CONTENTS	xvii
	10.4 Standards Activities	518
	10.4.1 Supporting Groups for Standards	520
	10.5 Summary	521
	10.5.1 Recommendations	521
	10.5.2. Future Trends	521
	References	522
11	Electrical Design of Digital Multichip Modules	525
	11.1 Introduction	525
	11.2 Delay and Noise in Digital Design	526
	11.3 Propagation Delay and Reflection Noise	531
	11.3.1 Reflections	540
	11.3.2 Line Losses	541
	11.3.3 First Incidence Switching	546
	11.3.4 Net Topology	549
	11.3.5 Effect of Loading	550
	11.4 Crosstalk Noise	551
	11.5 Simultaneous Switching Noise	555
	11.6 Other Sources of Noise	560
	11.7 The Electrical Design Process	561
	11.7.1 Technology Selection and System Planning	561
	11.7.2 Modeling, Simulation, and MCM Layout	564
	11.8 Summary	565
	References	567
12	Thermal Design Considerations for	
	Multichip Module Applications	569
	12.1 Introduction	569
	12.2 Thermal Management	570
	12.2.1 Objectives in Thermal Management	570
	12.2.2 Thermal Paths	571
	12.3 Thermal Phenomena in Electronic Enclosures	571
	12.3.1 Heat Transfer Mechanisms	572
	12.3.2 Heat Transfer in Electronic Components (Modules)	575
	12.3.3 The Concept of Thermal Resistance	577
	12.3.4 Heat Transfer On a Board	580
	12.3.5 Thermal Coupling in Electronic Enclosures	581
	12.4 Thermal Management of MCMs	583
	12.4.1 Alternate Thermal Control Methods for MCMs	583
	12.4.2 Cooling Methods - Cost Impact of	
	Thermal Management Techniques	592
	12.4.3 Parameters Impacting MCM Thermal Performance	593

	12.5 Tools	for Thermal Design	595
	12.5.1	Overview of Design Analysis Tools	595
	12.5.2	Analysis Tools	596
	12.5.3	Solution Procedure	598
	12.5.4	Analytical Modeling - Integral Method	598
	12.5.5	Computer Based Tools - Numerical Method	604
	12.5.6	Experimentation - Why, When and How	607
	12.6 Sumr	nary	610
	References	·	611
13	Electrical '	Testing of Multichip Modules	615
	13.1 Introd	luction	615
	13.2 Subst	rate Test	616
	13.2.1	Introduction	616
	13.2.2	Fixed Probe Array Testing	619
	13.2.3	Single Probe Testing	621
	13.2.4	Two Probe Testing	625
	13.2.5	Electron Beam Probing	628
	13.2.6	Developing a Substrate Test Strategy	630
	13.2.7	Summary	630
	13.3 IC Te	est	631
	13.3.1	Requirements for MCM Modules	631
	13.3.2	Introduction to IC Test	632
	13.3.3	Test Generation	633
	13.3.4	Boundary Scan and Built-In Self Test	634
	13.3.5	Parametric Testing	637
	13.3.6	Wafer Probing	638
	13.3.7	Die Carriers and Packages	638
	13.3.8	Final Test	640
	13.3.9	Burn-In	641
	13.3.10) Summary	644
	13.4 Asser	mbled Module Test	644
	13.4.1	Introduction	644
	13.4.2	Testing Strategies	645
	13.4.3	Design for Testability	649
	13.4.4	Test Generation	651
	13.4.5	Test Equipment	654
	13.4.6	Bring-Up	655
	13.4.7	Production Test	656
	13.4.8	Summary	657
	13.5 Conc	lusion	658

	со	NTENTS	xix
Par	rt C: Case Studies		661
14	The Development of Unisys Multichip Modules		665
	14.1 Introduction		665
	14.2 The Driving Forces Behind Multichip Packaging		666
	14.3 Advances in Silicon Delay Multichip Modules		667
	14.4 Single Chip Packages have Limitations		668
	14.5 Basic Packaging Goals		668
	14.6 Multichip Module Development Process		669
	14.6.1 Designing Through Technology Change		669
	14.6.2 Develop a Multichip Packaging Strategy First		670
	14.6.3 Technical Decision Making in an MCM Program		670
	14.6.4 Making Multichip Module Tradeoffs		
	in a Disciplined Way		671
	14.6.5 Temperature Hierarchy Management		676
	14.6.6 Problems Encountered and Solutions Developed		676
	14.6.7 Good News for the MCM Developer		678
	14.6.8 Verify the Reliability of the Multichip Module		679
	14.6.9 Summary of Development Phase Challenges		679
	14.7 Unisys Multichip Module Implementations		680
	14.7.1 SCAMP		680
	14.7.2 A16/A19		682
	14.7.3 2200/900 Double Sided Multichip Module		683
	14.7.4 Limits of Cofired MCMs		685
	14.7.5 Thin Film MCMs		686
	14.8 Multichip Module Industry Issues		690
	References		693
15	High Performance Aerospace Multichip Module		
	Technology Development at Hughes		695
	15.1 MCMs Meet the Needs of Systems Evolution		695
	15.1.1 Examples of Trends in Weapons		696
	15.1.2 Measuring the Trends		701
	15.2 New Packaging Roles		706
	15.2.1 A High Density Multichip Interconnection Program	L '	706
	15.2.2 A Company Wide Electronics Packaging Program		706
	15.3 Developing Two Key MCM Technologies		707
	15.3.1 An Evolutionary Approach	,	708
	15.3.2 Technology Comparisons	,	709
	15.3.3 HDMI Technology	,	711
	15.3.4 Low Temperature Cofired Ceramic (LTCC) Techno	ology '	718

XX	CONTENTS

	15.4 Comprehensive Design/Manufacturing Services	721
	15.4.1 Design	722
	15.4.2 Interconnection Structures	727
	15.4.3 Pretesting of Packaging System Elements	727
	15.4.4 Component Attachment and Connections	729
	15.4.5 Electrical Testing	731
	15.4.6 Inspection, Quality and Reliability	732
	15.4.7 Encapsulation	732
	15.5 Summary	734
	References	734
16	Silicon-Based Multichip Modules	737
	16.1 Introduction	737
	16.2 Materials	738
	16.2.1 Silicon Substrates	738
	16.2.2 Conductors	743
	16.2.3 Dielectrics	744
	16.3 MCM Examples	747
	16.3.1 Prototype Silicon Substrates	748
	16.3.2 Custom Silicon Substrates	752
	16.4 Summary	761
	References	766
17	The Technology and Manufacture of the	
	VAX-9000 Multichip Unit	769
	17.1 Introduction	769
	17.2 MCM System Perspective	770
	17.3 VAX-9000 MCU Development	772
	17.3.1 VAX-9000 Performance Goals	772
	17.3.2 Analysis of Alternative Physical Technologies	772
	17.3.3 VAX-9000 MCU Strategy	773
	17.3.4 MCU Manufacturability Goals	774
	17.3.5 Concurrent MCU Engineering and Manufacturing	774
	17.3.6 MCU Quality Engineering	775
	17.4 VAX-9000 MCU Technology	776
	17.4.1 MCU Overview	776
	17.4.2 MCU Integrated Circuits	778
	17.4.3 HDSC Interconnect Technology	780
	17.4.4 MCU Structure	782
	17.4.5 MCU Electrical Characteristics	784
	1/.4.6 MCU Thermal Management	785

	17.4.7 MCU Test Technology	786
	17.5 VAX-9000 MCU Manufacturing	795
	17.5.1 Overview of Manufacturing Process	795
	17.5.2 HDSC Signal Core Fabrication Process	797
	17.5.3 HDSC Power Core Fabrication Process	800
	17.5.4 HDSC Assembly Process	801
	17.5.5 MCU Subassembly Process	803
	17.5.6 MCU Assembly Process	807
	17.6 Future Evolution of MCM Technology	808
	17.6.1 Strategy for Migration of MCM Technology	
	from Mainframe to Low Power Applications	809
	References	813
Par	rt D: Closing the Loop	815
18	Complementing Technologies for MCM Success	817
	18.1 Introduction	817
	18.2 Separable Connectors and the Packaging Hierarchy	818
	18.3 Effect of Semiconductor Type on Package Design	824
	18.4 System Performance Level (Cycle Time)	826
	18.5 Level of Chip Integration, Function Size	
	and Chip I/O	828
	18.5.1 Memory MCMs	828
	18.6 Chip Joining Interface, Burn-in, Testability and Repair	830
	18.6.1 Known Good Die	830
	18.6.2 Chip Joining Interface	831
	18.6.3 Flip TAB	831
	18.6.4 Flip Chip	831
	18.7 Power Density and Cooling Methods	832
	18.8 Timeliness of Introduction	834
	18.9 Alternatives to MCMs	834
	18.9.1 Future Growth in Single Chip Packaging	836
	18.10 Inexpensive, Low Complexity MCMs	838
	18.11 Summary	840
	References	840
Ep	Epilogue	
Ab	out the Authors	845
Ina	index	

FOREWORD

Far from being the passive containers for semiconductor devices of the past, the packages in today's high performance computers pose numerous challenges in interconnecting, powering, cooling and protecting devices. While semiconductor circuit performance measured in picoseconds continues to improve, computer performance is expected to be in nanoseconds for the rest of this century - a factor of 1000 difference between on-chip and off-chip performance which is attributable to losses associated with the package. Thus the package, which interconnects all the chips to form a particular function such as a central processor, is likely to set the limits on how far computers can evolve.

Multichip packaging, which can relax these limits and also improve the reliability and cost at the systems level, is expected to be the basis of all advanced computers in the future. In addition, since this technology allows chips to be spaced more closely, in less space and with less weight, it has the added advantage of being useful in portable consumer electronics as well as in medical, aerospace, automotive and telecommunications products. The multichip technologies with which these applications can be addressed are many. They range from ceramics to polymer-metal thin films to printed wiring boards for interconnections; flip chip, TAB or wire bond for chip-to-substrate connections; and air or water cooling for the removal of heat.

msgalica@mintz.com

While there are several books now on packaging, these books deal with the subject of multichip modules as part of packaging in general, or they treat a particular multichip module technology or they are at an advanced level. What is needed, therefore, is a *comprehensive book at the basic level*, structured so that anyone entering the field can quickly learn about the technologies, understand the tradeoffs, review the product examples, and make systems level decisions.

Such a book has been provided by Daryl Ann Doane and Paul D. Franzon. They have worked with an outstanding team of packaging experts from industry and universities. Together they have produced **Multichip Module Technologies and Alternatives: The Basics**, an outstanding book for both industry and university use. It is equally appropriate as an introduction to the multichip module technologies for those just entering the field, and as an up-to-date basic technical book for those currently practicing in it.

The books deals with the subject of multichip modules along three parts: systems level perspectives including packaging technology options and costs, the basics of ceramic, thin film and printed wiring board technologies as well as chip and module level connections; thermal and electrical design considerations including electrical testing; and finally product examples illustrating how multichip modules have been useful.

The basic and integrated nature of the book clearly reflects the dedication and the hard work of the editors and the authors.

Rao R. Tummala IBM Fellow

PREFACE

Welcome!

Welcome to our book. We feel it is a unique book in the field of packaging and we hope you find it both useful and enjoyable. We (editors and authors) have certainly enjoyed bringing it to you!

Uniqueness of the Book

This is a very unique book! Its uniqueness comes about in two ways: first in the approach to the subject, and second, in the approach to the writing of the book.

First, we feel that this book helps define a turning point in the discipline of packaging. The "bottleneck" to increased systems performance is now more often the package than the chip. One effect of this is that suddenly a whole "breed" of engineers need to gain an understanding of how package design affects their systems performance and cost goals. Another effect is the widespread recognition that multichip module packaging technologies are possible solutions to this performance limitation. The attendant explosive growth in the MCM technology alternatives available is a testimony to this recognition.

Elm Exhibit 2162, Page 23

msgalica@mintz.com

Until recently, packaging was mainly the domain of mechanical and materials specialists, and furthermore was rarely taught at universities. This has changed. Almost overnight, the pressure of high on-chip speeds and high transistor counts meant that packaging became a subject that must be understood by just about any engineer involved in designing a system. When these engineers tried to establish their understanding they found themselves in a virtual "Tower of Babel." First, often the same terms were used for different things or, just as bad, different terms were used for the same thing. Second, as this was the domain of specialists, significant background was required in each discipline in order to understand it. This book turns the discipline of packaging into a subject accessible to the generalist rather than just one accessible to the specialist. Common terms are defined and crosslinked to the terms in current usage. Each discipline is presented in such a way that is both understandable to all and is useful. In providing this book, we help turn packaging into a discipline in which anyone can participate. We also present a book highly suited for teaching within a university.

The book also is unique in how it was created! Packaging is a multidisciplinary subject. We realized that no two people can claim mastery of all the disciplines needed and involved in the subjects we wished to cover. We also realized that rapid writing was important. Thus an edited text was called for. But edited texts are often poorly lacking in terms of understandability and flow. Usually an editor relies on selecting appropriate experts and then accepting what each expert writes with minor alterations.

This book is very different. It could be called a "closely edited" text. Each chapter has had the heavy hand of development of both editors in it. This happened in several forms. Often we spent many hours with individual authors defining what we thought was appropriate and guiding them in the actual writing. In all cases, as well as seeking outside reviewers, we wrote "anonymous" reviews ourselves. In some cases, we directly adjusted the text after it was submitted. Thus we are responsible for the final product as much as each chapter author is.

The result, we feel, is a book that has the authority that comes with a book written by a team of experts, but has the understandability, completeness and flow of a book written by a single author, At least we hope that the book comes as close to that ideal as possible!

Audience

For whom is the book intended? This book is for everyone! The emphasis in the book is on understanding the fundamentals and the reporting of real

experiences rather than including a huge amount of data. It is intended for those who need a broad exposure to the concepts underlying the design, fabrication, packaging, assembly, manufacturing of multichip modules and the costs associated with alternative packaging technologies. The book is intended for applications, manufacturing and design engineers as well as for technical decision makers and managers who are confused about MCM issues but who wish to understand the fundamentals and basics. Specifically, it will be useful to

- engineers in design, processing, fabrication, manufacturing, assembly and test who need to choose a packaging technology for specific product and application goals;
- managers determining technology alternatives for new systems needs;
- marketing; sales technologists who need a working knowledge of the alternative MCM technologies.

The book also is a suitable text for advanced undergraduate and graduate students in design, electrical, mechanical and systems engineering as well as for students in the applied sciences as discussed further on in this Preface.

Philosophy of the Book

What are the key decisions needed when considering using MCM technologies? There are four perspectives: materials, manufacturing, systems performance and cost requirements. The book presents the basics of MCM technologies from these four perspectives with an emphasis on decision-making. How do you choose a packaging technology from a systems performance, cost perspective? What are the fundamental materials and manufacturing issues that need to be considered? Is the packaging strategy appropriate to the design goal?

This book is applications-oriented in the sense that it discusses

- Examples of MCMs and the products and systems in which they have been used,
- Examples of equipment and processes used to design, build and test MCMs,
- Actual case studies and insights provided by leading companies themselves rather than a summary of reported results.

XXVIII PREFACE

Organization of the Book

The book is divided into four parts which we call

- Part A The Framework (Making Decisions: The Big Picture)
- Part B The Basics
- Part C Case Studies
- Part D Closing the Loop

Each Part begins with what we call a "frame" in which we summarize the purpose or goals of the Part. The frames alert the reader to what theme dominates in the Part, and what can be learned from it. The frames are set apart from the rest of the book by the border around their pages.

For example, in Part A, the goal is to present a basis for understanding multichip module technology possibilities for packaging. We call this a *framework for understanding*. The basic definitions needed for understanding are presented. Some alternative packaging approaches are discussed, and the themes of decision-making and the design process that run through the book are also introduced. The reader is alerted to look for the global picture in Part A, an overview that a breadth of knowledge is required for understanding, and that an awareness of the basic concepts introduced is Part A will enable him or her to follow the detailed technical information in Part B.

Part B provides an understanding of the technical fundamentals in the design, fabrication and testing of multichip modules. The tutorials in Part B are unique in the published literature. For example, there is detailed coverage of MCM-L, the fabrication technology based on laminate structures. Another special feature of Part B is the discussion of multichip module-to-printed wiring board (second level) connections in Chapter 10. The test chapter (Chapter 13) provides specific guidelines on how to design MCM products for testing and how to reduce test cost and effort. In each Chapter, the reader is alerted to the options associated with the technologies presented so that these options can be related to product applications.

Part C consists of "reports" from some companies that have created and are selling successful MCM products. The development of their MCM technologies, internally from existing expertise, is described. Here the reader will benefit from the insights shared by these companies.

Finally, Part D "closes the loop." It focuses on what aspects of all the technologies presented are likely to have the greatest impact on meeting future needs from a systems perspective. It also presents a *forward view* by describing

some open, unsolved problems, open challenges that must be met if future systems are not to be constrained in performance by limitations in packaging technologies.

Use as a Textbook

We feel that this book is highly usable as a packaging textbook. It covers this interdisciplinary topic at a level that is accessible to all while providing concrete learning material. This book emphasizes decision-making and design, and thus, relates directly to what practicing engineers do for a living. It would be useful in both graduate courses and advanced undergraduate courses. It has the tightness and continuity of a textbook in contrast to a typical edited book. In our opinion, this book is superior to any current packaging "textbook" for use as a text mainly because of the style of presentation.

Despite its title, it could be used for many courses that do not feature only MCMs. Throughout this book, single chip packaging is presented as an alternative to be considered. The design chapters apply equally well to single chip and multichip package design. The only single chip package elements that are missing are a discussion of how plastic packages are made and a specific discussion of how boards are assembled. Chapter 9 covers some of these related points. Board manufacturing is covered in Chapter 5. Single chip ceramic package manufacturing uses similar processes as those for making ceramic MCMs which are discussed in Chapter 6.

There are several types of courses that can use this book. One of us (PDF) uses the book in a semester long graduate course on packaging design for electrical engineers. That course starts with an overview of packaging alternatives (Chapters 1 and 2, with 5 to 10 as references), discusses systems level decision-making (Chapter 3 and parts of the Case Studies), and briefly discusses cost (parts of Chapter 4). Almost half of the course is concerned with electrical design of packaging (Chapter 11). A small amount of time is spent on thermal design (Chapter 12), to the level of detail needed by an electrical engineer, and on the appropriate impact of testing on design (Chapter 13).

The book also would be useful for a more technology and manufacturing oriented course. There Chapters 1 and 2 would be used to introduce the course. Establishing the reasons for the multiple packaging alternatives available and how to choose among them would be covered with the aid of Chapter 3. Chapter 4 provides a detailed look into the impact of manufacturing costs. Chapters 5 through 10 would be used to teach the bulk of the course. Brief references to Chapters 11 through 13 would provide technologists with an

XXX PREFACE

understanding of how their decisions affect design. Chapter 17 would offer a detailed manufacturing viewpoint.

Finally, this book also could be useful to mechanical and chemical engineers and materials science-based courses on packaging.

Daryl Ann Doane Paul D. Franzon

msgalica@mintz.com

ACKNOWLEDGMENTS

With the publication of this book, it is a special pleasure for one of us (DAD) to express gratitude and appreciation to Paul B. Wesling, Tandem Computers. His longstanding and patient role as mentor and resource person has provided an immeasurable educational opportunity and insight regarding the value of technical publications. He currently serves as Vice President of Publications for the IEEE Components, Hybrids and Manufacturing Technology (CHMT) Society. He has been a *pioneer* in the area of technical publications related to peerreviewed journals, reprint as well as conventional books and Conference Proceedings. His innovative ideas in publications have been adopted by Societies outside of the IEEE, and also are reflected in the value and vision for this book. Thank you, Paul!

Several people advised us on technical and organizational topics. They generously contributed their time and valuable expert knowledge in extended discussions. In particular, Dr. Robert C. Sundahl, Intel, Dr. Walter J. Bertram, Alcoa Electronic Packaging and Professor R. Wayne Johnson, Auburn University, shared their ideas on what was needed in the book, as well as providing detailed reviews for several of the chapters. We wish to thank them.

John H. Lau, Hewlett-Packard was helpful in sharing his experience as an editor of three books for Van Nostrand Reinhold. He never tired of answering questions; we appreciate his patience and helpfulness very much.

msgalica@mintz.com

XXXII ACKNOWLEDGMENTS

John Nelson, Unisys, is very special! A faithful colleague and friend, he graciously responded positively to so many requests, providing the value of his longstanding experience and insight in the packaging industry in answering questions pertaining to the book. It has been a wonderful opportunity to discuss the technical aspects of the book, as well as its human side, with him. Many of the unique aspects of the book are due to his suggestions. We thank him for everything!

Each chapter in the book was reviewed by at least three individuals whose areas of expertise covered the chapter topic. We also deliberately asked individuals who worked in areas peripheral to a chapter topic to review the chapters for comprehension and clarity. Although we cannot list these more than 54 individuals, we are very appreciative of their efforts, thoroughness and helpful comments. We took them all very seriously.

Still, with all the technical expertise brought to form the contents of this book, there would have been no way to complete a book of this scope, size and complexity had it not been for the unique combinations of dedication and perseverance of the staff and management at Van Nostrand Reinhold (VNR), our publisher. We are very grateful for their efforts. Especially noteworthy was the standard of inspiration set by Mr. Stephen Chapman, Electrical Engineering Editor. He shared our "vision" for the book and persevered on our behalf for the conventional as well as the nonconventional requests we made. He arranged for an accelerated publication schedule so the book could be available when requested. Without his dedication to the book it would not have appeared in such a timely manner.

Sue Alexander of Seaborn Enterprises, Chandler AZ, was responsible for producing the galleys with artwork for the entire book. She demonstrated a high level of skill and innovation in adapting busy tables and complex equations to a text already filled with an unusual amount of photos and artwork. All the while she was faced with unprecedented time constraints. We are very much in awe of the final excellence of her creation in the layout and appearance of the book.

Most of all, we would like to thank those who are close to us. One of us (DAD) wishes to express special thanks to Dr. John L. Doane, Timothy A. ("TAD") Doane and Stevie Doane for their love, support and patience. Their faithful encouragement and sustaining, as well as helpful, assistance contributed significantly to the existence of the book. The other of us (PDF) gives loving thanks to Debra L. Ray in acknowledgment of her forebearance during the many long evenings and weekends when he was not there.

Part A—The Framework Making Decisions: The Big Picture

"Cheshire Puss," she began, rather timidly, "Would you tell me, please, which way I ought to walk from here?" "That depends a good deal on where you want to get to," said the Cat.

> Alice in Wonderland by Lewis Carroll

In the first part of this book, we provide a *framework (or basis) for understanding* multichip module packaging possibilities. Construction of our framework begins by defining the alternative approaches available with respect to multichip module structures and the packaging strategies involving such structures. It also includes such issues as manufacturability, performance, cost and time-to-market common to any packaging strategy - single chip or multichip. Finally, in the construction of the framework, we suggest a design methodology or decision-making process by which a packaging strategy can be coupled to a particular product goal. There is no single "right" or "wrong" packaging technology, but only relative to a specific application goal.

By constructing our framework of understanding, it is possible to cover a broad base of information in a short time. Such an overview, or global picture, also emphasizes the multidisciplinary nature of the multichip module technologies involved. Because of the breadth of understanding required in the packaging "sciences," there is some confusion about multichip module packaging as a viable, practical technology alternative. Suddenly people who did not have to think at all about packaging must consider complex packaging alternatives. Managers and engineers are being forced to make decisions without a complete understanding of packaging alternatives. In Part A, we bring together the multiple disciplines involved in such decision

msgalica@mintz.com

making considerations in a way understandable to the nonspecialist and to the student, as well as informative to a worker in the field.

In Chapter 1 of Part A, for example, the multichip module structure is defined and the alternative multichip module technologies are described. The themes of decision-making and the design process that run through the book also are introduced. The challenges presented to growth in the use of MCM systems, such as their performance, cost and associated infrastructure, as well as single chip alternatives, are discussed.

Chapter 2 provides an overview of MCM alternatives concentrating on materials and manufacturing issues. Specific examples of how some multichip modules have been manufactured are discussed.

We also consider (in Chapter 3) the relationships between the "needs" of the system being packaged and the choices of packaging technologies in terms of performance and cost, needs being defined in terms of application or product goals.

Packaging decisions are very sensitive to both direct and indirect costs which are discussed in all the chapters. Specifically, Chapter 4 of Part A focuses on cost modeling, at the component level, and from a manufacturing and design perspective.

In the Framework, the factors that influence high level packaging decisions are explored. Thus, Part A is particularly useful to the technical manager who needs to make decisions on the best applications of this new technology to company products, but does not need to understand the detail required to design these products. The practicing engineer will gain the breadth of knowledge required to understand the full impact of the new technology. The basic concepts introduced will enable an understanding of the detailed technical information provided in Part B. Marketing and sales people also will find this part useful in communicating to customers how the company packaging decisions help meet customer needs.

INTRODUCTION

Daryl Ann Doane

1.1 BACKGROUND AND DEFINITIONS

1.1.1 Purpose and Perspective of the Book

This is a book about multichip modules (MCMs). And the strategies involved in implementing them *successfully*! The book discusses the technology alternatives appropriate for MCMs, and the decision-making processes required for choosing the "best" technology for a particular application. The book helps to answer the question: *Is the technology applicable (or appropriate) to the design goal?*

Multichip modules are not new! The basic science and technologies involved in the design, materials, fabrication and manufacturing of MCMs have been known generally for over 30 years. These packaging technologies span a wide range of technical disciplines, such as chemistry, physics, materials science, electrical and mechanical engineering. The broad range of technologies which now must be understood by an MCM designer, for example, has expanded beyond a conventional experience base. One goal of this book is to provide information so that such an individual can learn about the various available technologies, and how and which one to choose for his or her particular packaging needs.

msgalica@mintz.com

4 INTRODUCTION

In the past 30 years, there has been an exponential growth in transistor speed and in the number of transistors per chip. For example, both have doubled over the last three years. Chip interconnection speeds and chip I/O count have not increased at the same rate. As a result, chip interconnections play a more dominant and limiting role in determining overall system speed and performance. Packaging of the chips has become a more significant factor in performance. Systems level performance improvements are now being limited more by the packaging and interconnection technologies, and less by the chip technology itself.

Advanced packaging and interconnection technologies have been mainly the domain of high performance computers whose performance otherwise would have been limited by the package. MCMs are viewed by some as the *only suitable technology* for packaging such systems. Soon the performance of many more systems will be limited by the package, so it is anticipated that MCMs will appear in a wider range of systems in the future.

Some fundamental advantages of MCMs are:

- Increased system speed
- Reduced overall size
- Ability to handle chips with large numbers of input/outputs (I/Os)
- Increased number of interconnections in a given area
- Reduced number of external connections, for a given system function

Every external package connection degrades the system performance somewhat by its parasitics (undesired circuit elements). Each package lead has some parasitic capacitance and inductance that distort the shape of the signals passing through it. Reducing the number of external connections not only reduces these parasitics, but also generally contributes to system reliability. Thus, one way MCMs would reduce the parasitic effects is through the reduction in the number of second level connections they afford.

This book discusses selected technologies and alternatives for MCM design and fabrication. A multidisciplinary approach is used to evaluate the interactions between materials, processes, designs and assembly techniques. Contributions by different authors provide perspectives from many different viewpoints. The emphasis throughout the book is on making decisions among the technology alternatives for specific applications and products.

This chapter has two main purposes. First it provides the definitions needed to understand the book and relates these definitions to terms used in other chapters. These terms are common "jargon" in various disciplines, technologies or corporate cultures, and thus, may vary from chapter to chapter depending on the perspective. By relating these terms to each other and to the basic concepts, we hope to remove some stumbling blocks in the way of the reader's understanding.

The second purpose of this chapter is to introduce some of the themes and perspectives presented in the book. Major themes include decision-making as it relates to designing MCM-based products and as it relates to implementing MCMs in the marketplace, MCM structures and technologies, and MCM manufacturing. These themes are introduced so that the reader can start relating to the multiple disciplines and approaches that are concerned with packaging right from the start of the book.

1.1.2 ARCHITECTURE - Building a Multichip Module Structure

In this section we define what is meant by a multichip module, and introduce the alternative technologies used to build a MCM structure.

What is a Multichip Module?

In this book we define a multichip module (MCM) as a structure consisting of two or more integrated circuit chips electrically connected to a *common circuit base* and interconnected by conductors in that base. The conductors in the base are usually patterned in multiple layers separated by a suitable insulating dielectric material, with vias for interconnections between layers. The base also provides the required mechanical support for the chips. Figure 1-1 shows schematically an MCM structure.



Figure 1-1 MCM architecture (schematic).

6 INTRODUCTION

Based on this definition of an MCM structure, *multichip modules are not new!* In the very early days, MCMs were called hybrid circuits. Hybrid circuit assemblies are still in use, and usually contain lower lead count analog and digital integrated circuits (ICs), as well as active or passive discrete components. They may contain many layers of metallic conductors and ceramic dielectrics, but they classically have been formed one layer at a time. Chip-on-board (COB) assemblies, where bare chips are wire bonded to conventional printed wiring boards, also have been in use for some time.

Another definition for MCMs is based on the interconnection technologies that can be used in their structures. In particular, the IPC (Institute for Interconnecting and Packaging Electronic Circuits), according to its standard IPC-MC-790 [1], defines three types of technologies that can be used to make MCM structures:

- MCM-L: Modules using advanced forms of printed wiring board (PWB) technologies to form the copper conductors on plastic laminate-based dielectrics.
- MCM-C: Modules constructed on cofired ceramic substrates using thick film (screen printing) technologies to form the conductor patterns. ("Cofired" means that the conductors and the ceramic are all heated in an oven at one time.)
- MCM-D: Modules whose interconnections are formed by the thin film deposition of metals on deposited dielectrics, which may be polymers or inorganic dielectrics.

Some of the authors in this book refer to these structures as laminate MCMs, cofired ceramic or thick film MCMs, and thin film MCMs, respectively. All of these structures satisfy the definition of containing multiple chips electrically connected to, and interconnected on, a single base structure that is eventually connected to a printed wiring board.

A third approach to defining an MCM is based on the packaging efficiency achieved by the technology. Packaging efficiency (or "silicon density") is defined as the percentage of area occupied by silicon ICs [2]. Using this approach, an MCM can be defined as follows: An MCM is a structure in which a packaging efficiency of greater than 30% is achieved. This definition implies a particular degree of technology which allows the chips to be packed closely together. It excludes many hybrid and COB designs. For the purposes of this book, we seek to understand the broader construction of a multichip module structure without the 30% packaging efficiency restriction.
What is the MCM Architecture?

The basic idea of MCMs is to decrease the average spacing between ICs in an electronic system. The fundamental aspect of MCM technology is therefore the chip interconnection technology. This technology includes the method of connecting the I/O conductors on the chips to the common circuit base, and also the method of sending signals through that base between the chips. The main goals are higher performance resulting from reduced signal delays between chips, reduced overall size and reduced numbers of external connections.

The elements that make up an MCM structure, or architecture, are shown in Figure 1-1. These elements are the chips, the first level connections, the common circuit base, the second level connections and the package. The remainder of this section functionally defines these elements and introduces related terminology. Succeeding sections briefly present some of the available technology alternatives.

We use the term *common circuit base*, because it is common to all the chips, because it contains the signal interconnection circuits and the power and ground distribution circuits, and because it provides a mechanical support base for all the chips. Sometimes the top of the common circuit base is called the "substrate" or the "MCM substrate". See, for example, Figure 2-1. In that figure, the lower "substrate base" is that part of the common circuit base that provides mechanical stiffness to the MCM. In Figure 1-2, the "substrate/package circuitry" and the "substrate" correspond, respectively, to the "MCM substrate" and the "substrate base" of Figure 2-1.

Sometimes the substrate and the substrate base are the same. For example, in an MCM-C, the interconnections are formed in a solid ceramic. On the other hand, in an MCM-D, the interconnections are formed in a plastic-like polyimide dielectric. For mechanical support this substrate then needs to be attached to a substrate base, typically ceramic or silicon. (In other places, a silicon substrate base may be called "the substrate," based on its similar function in an IC.)

The first level connections consist of the signal and power/ground wires provided between the chips and the common circuit base. Not shown in Figure



Figure 1-2 Common types of first level connections. Chip to common circuit base. (Courtesy E. Larson).

msgalica@mintz.com

1-1 is the die attach. The terms "die attach," "chip attach" or "chip connect" are used to describe the physical anchoring of the die to the common circuit base. The wires in the first level connection usually form only part of the die attach. Often the chips are bonded to the common circuit base as well.

Note the distinction between the use of the words "connection" and "interconnection." In this book, the term "connection" is used to refer to the electrical connection between two levels of the packaging architecture in an electronic system. The term "interconnection" (signal interconnection, interconnection structure, or interconnection circuit) refers to the conductors provided within the common circuit base.

The second level connections are the connections between the MCM and the PWB on which it is mounted. Again, these paths refer to the current carrying conductors. Separate structures are sometimes used to fix the MCM physically to the PWB. Second level connection alternatives are explored in Chapter 10.

The package is the final part of an MCM, which can be seen, touched and felt. It provides the following functions:

- Physical protection from environmental (corrosion, humidity) and mechanical (vibrations, shock) stresses and from handling by automatic machinery used during manufacturing processes
- Part of the second level connection elements for the distribution of signals, power, and ground
- A means for removing the heat dissipated in the chips
- A space transformation (fanout) from the fine conductor spacings on chips to the coarser spacings on PWBs

Thus, the electronic package is an electromechanical structure that supports, protects and connects (electrically and thermally) the devices contained within it, independent of the actual number of chips involved - single chip or multichip. The package serves as the link between the component or circuit level functions (such as ICs, discrete devices, resistors, capacitors) and system level functions (such as circuit board assembly and board to board connections). In addition, it often allows testing and burn-in of its contained parts prior to further assembly.

In Table 1-1, we summarize the basic MCM architecture outlined above, and provide some examples of functions and technologies for each level in the architecture. (Sometimes this architecture is called the "hierarchy.") Notice that the order of the levels from top to bottom in Table 1-1 is the same as the physical order, shown schematically in Figure 1-1. In the next sections we define in more detail some of the technologies shown in that Table.

LEVEL	FUNCTIONS	TECHNOLOGIES
Chips	Digital Analog	Si: CMOS, bipolar GaAs
1st level connection	Conductor connection from chips to common circuit base	Peripheral: wire bond, TAB, flip TAB Area: flip chip solder bump
Common circuit base	Signal interconnection Power and ground conductors	Hybrid circuits MCM-L, MCM-C, MCM-D MCM-Si, MCM-D/C
MCM package	Hermeticity Heat removal Physical protection Conductors	Peripheral conductors: DIP, QFP Area array conductors: PGA, PAC
2nd level connection	Conductor connection to PWB	Plated through-hole Surface mount

Table 1-1 Basic MCM Architecture.

1.1.3 First Level Connection and Common Circuit Base Alternatives

Most of the methods for connecting the I/O conductors on the chips to the common circuit base also are used within single chip packages. See Figure 1-2. Wire bonding and tape automated bonding (TAB), for example, are used to make the connections between peripheral conductors on single chips and their packages. With "flip chip" arrangements, such as flip TAB, the chip is flipped over so that its conductors are on the bottom, facing the common circuit base. Flip chip solder bump (FCSB) connections are analogous to the connections beneath single chip surface mount area array packages to conductors on PWBs.

Physical attachment ("chip attach" or "die attach") of the chips to the common circuit base is a distinct operation from wire bonding or TAB conductor connections, but occurs simultaneously with the solder connections in FCSB. With flip TAB, there is no metallurgical bond between the chip and the common circuit base. Sometimes epoxy is used, and sometimes the chip is held in place by the connected TAB leads.

The common circuit base is that part of the MCM structure that defines the MCM technology used: MCM-L, MCM-C or MCM-D. However, MCM signal

interconnection technologies themselves have developed from other technologies, such as patterning ICs on silicon, packaging ICs and mounting ICs on PWBs. The names for these technologies are relatively new [1], but the basic technologies are fairly mature.

For example, decreasing the sizes of conductor lines and spaces on laminate PWBs leads to the MCM-L technology ("L" stands for laminate). MCM-L technology also is an outgrowth of the COB technology, since the packages around individual chips are omitted. The laminate normally includes several layers of patterned metal conductors pressed together (laminated) with interspersed polymer dielectric insulating layers. The adhesive joining of all these layers is done at a relatively low temperature.

MCM-C technology ("C" stands for cofired ceramic, but also may include thick film) is a development of traditional hybrid circuit technology, where printed thick film conductors are used to provide the signal interconnections. To increase the circuit density, multiple thick film layers of conductor and dielectric are used, just as in PWBs. The layers originally had to be built up with sequential ceramic firing operations, since the dielectric was applied as a thick film of ink. Later developments allowed conductors and ceramic layers of alumina to be fired all at once, provided this was done at a higher temperature ("high temperature cofired ceramic" technology or HTCC). One early application of this cofired MCM-C structure, with over 30 ceramic layers, was for mainframe IBM computers [3].

The glass-ceramic dielectrics used originally in ink form were later made into tapes. (The layers of unfired ceramic generically are called "green tape." "Green" is a ceramicist's term for the unfired material, even though the sheets of tape actually may be other colors.) Simultaneous firing of all the layers was then possible at the lower temperatures used in original hybrid circuit technology. This is called "low temperature cofired ceramic" technology or LTCC.

To obtain the highest density MCM signal interconnection circuits, other technologies use techniques such as photolithography developed for patterning circuits on silicon chips. With MCM-D ("D" stands for deposited), both the conductors and the insulating dielectric layers that separate them are deposited. The dielectric is usually an advanced (organic) polymer such as polyimide, although (inorganic) silicon dioxide (SiO₂) is sometimes used. Multilayer conductors are deposited as thin films, such as are used on single layer hybrid circuits for resistors and bonding pads. Vacuum metal deposition processes such as sputtering, polymer deposition processes such as spin and spray coating, chemical vapor deposition (CVD) of SiO₂, and patterning processes such as wet chemical, or plasma (dry) etching and laser etching are familiar chip processing techniques.

The dielectrics used in MCM-D usually are not strong enough to stand alone. Thus, in this technology a separate, different material is used for mechanical stiffness and support. It may be silicon, metal, ceramic containing no conductors, or a PWB or multilayer cofired ceramic that contains conductors. The identification of this material is sometimes included in the technology label. For example, when MCM-D technology is used with a multilayer cofired ceramic substrate base containing the MCM power and ground circuits and signal I/O vias, the technology is called "MCM-D/C." When silicon is used, the MCM technology is called "MCM-Si." MCM-Si technologies may use either thermally grown SiO₂ or CVD SiO₂.

1.1.4 MCM Packaging Alternatives

Packaging technologies involve a consideration of the materials to be used, the geometry of the package, as well as the thermal and electrical design parameters. This section begins by considering the basic functions that a package must perform. Since most MCM packages have been, until now, similar or identical to single chip packages, the classifications of these packages also are summarized. Then, some features of packages specific to MCMs are examined.

Entire books [G1] have been written on microelectronics packaging. For details, the reader is referred to these general references at the end of the chapter. The intent in this section is to provide a working knowledge of the language of packaging and some aspects of packaging that will be useful and relevant to MCM design and technology selection.

Classifications of Packages

Packages can be classified by their main material, the second level connection technique, the means used to remove heat, the method and degree of protection (encapsulation) of the chips, first level connections, and common circuit base. In this part, we explore certain aspects of these classifications. Second level connection classifications are explained more fully in Chapter 10. Available options for heat removal are discussed in Chapter 12.

The most common materials used for the package body are ceramic and plastic. If ceramic is used, it is made using the same approaches used to make ceramic substrates, and then a metal lid is brazed to the package. Plastic bodies usually are made through a process called injection molding, where the chip is placed in a mold into which plastic is injected.

Packages may contain internal cavities, or recesses, for mounting chips or MCM substrates. These cavities provide room for many connections to the package leads.



Figure 1-3 Common through-hole and surface mount packages. (Courtesy P. Franzon.)

Some of the single chip types of package also used for MCMs are sketched in Figure 1-3. One way to classify these packages is in terms of the method of mounting to the next level in system assembly: through-hole mounting or surface mounting. The conductors are called leads or (straight) pins; in the case of surface mounted packages they may be pads.

Plated through-holes (PTH) in a printed wiring board (PWB) refer to precision holes drilled through the board and plated with copper. These copper-plated holes form electrical interconnections between bond pads on the top of the board and the conductors within the board. They also provide interconnections between the different conductor planes within the board.

Current PWB design rules and manufacturing capability limit the use of through-hole packages with high pin counts. For example, current PWB designs can accommodate holes on a 0.100" (2.5 mm) grid. Each package requires one hole for every package pin; each pin is typically 0.035" (0.9 mm) in diameter. Advanced PWBs can accommodate holes with 0.050" spacing. Surface mounted packages, on the other hand, can be connected to pads of the surface of PWBs with 0.025" spacing.

Packages can be grouped further according to physical arrangement of the leads (peripheral or area array), and package lead geometries, such as J-shaped and gull wing. The SOIC (small outline IC) and QFP (quad flat pack) shown in Figure 1-3, for example, have gull wing leads.

In Table 1-2, we classify the common packages, provide their abbreviations, and indicate typical lead (conductor) count and spacing.

The progression in Table 1-2 is from the simple to the advanced. The upper package types in this table all have peripheral leads. We start with dual in-line packages (DIPs), which have leads on only two sides and which are mounted in PTHs. Of the surface mounted packages with peripheral leads, SOJ stands for a small outline package with J- shaped leads, TSOP stands for thin small outline package, and PLCC stands for plastic leaded chip carrier. Next come the area array packages, with PTH mounted pin grid arrays (PGAs) first. If the PGA pins are shortened as shown in Figure 1-3, then PGAs can be surface mounted. Since surface mount pads on a PWB can be spaced more closely than PTHs, the possible lead pitch is reduced.

Area array pad array carriers (PACs) with solder bumps on the underside may be soldered directly to the PWB in a process analogous to connection of the solder bump for flipped chips (FCSB) to their MCM substrate. The process sometimes is called C5 (Controlled Collapse Chip Carrier Connection) for the PAC, and C4 (Controlled Collapse Chip Connection) for the FCSB. PACs are presently in the prototype stage; future lead spacings may drop to 0.02". A PAC is similar to a PGA with zero-length leads. Such a PGA is called a land grid array (LGA).

Properties of MCM Packages

Table 1-2 is only a starting point for MCM packaging: there are many alternatives.

MCM packages often look like regular-sized versions or large versions of well known single chip packages with pins or leads emerging from them. Sometimes the MCM has a highly customized package that does not resemble an existing single chip package. Often the substrate base (mechanical support

14 INTRODUCTION

PACKAGE TYPE	BC	DDY	LEAD		TYPICAL LEAD	
	Plastic	Ceramic	Arrange	Connect	Pitch (in.)	Count
Dual In-Line	PDIP	Cer/DIP	Peripheral	РТН	0.05 - 0.1	64
Small Outline	SOIC SOJ TSOP	Flat Pack	Peripheral	Surface	0.025 - 0.05	28
Chip Carrier	PLCC		Peripheral	Surface	0.05	84
Quad Flat Pack	PQFP		Peripheral	Surface	0.01 - 0.05	256
Pin Grid Array	PPGA	CPGA	Area	РТН	0.05 - 0.1	144; 299
Short Lead PGA	yes	yes	Area	Surface	0.05	to 500
Pad Array Carrier = Land Grid Array	yes	yes	Area	Surface	0.02 - 0.05	to 1000

Table 1-2 Common Package Classifications.

structure of the common circuit base) is used to form part of the package. In other cases, a package housing is not used. Instead, leads or pins are attached to the common circuit base for second level connection, and the exposed chips and wires are sealed against moisture by using epoxy or gel.

MCM packages must be mated carefully to the total MCM architecture. For example, the MCM common circuit base technology determines the possible lead geometry of the package. If there are electrical vias through the substrate bases for MCM-D or MCM-Si, the package leads then must be peripheral (as shown schematically in Figure 1-1). To allow area array conductors under the package with MCM-D, a multilayer ceramic package can be added (as in MCM-D/C). The first level connection determines the possible thermal paths. In the case of flip chip mounting, much of the heat can be taken through the top of the package. Then rather poor thermal conductivity in the MCM substrate can be tolerated. Since there are no electrically sensitive conductors on the backsides of the chips, the chips may be connected through thermal gel or elastic spacers directly to the package top, to which a heatsink or cold plate can be attached.

Hermeticity describes the relative effectiveness, or extent, of sealing of a package. Hermetic packages have very low leak rates in helium testing $(< 10^{-7})$ atm-cc/sec typically). Hermeticity or near-hermeticity is important in packaging, since moisture can cause electrical circuit failures. Water absorption in organic dielectrics results in swelling that can cause separation of conductors, and other electrochemical activity such as copper plating, for example. Also, water may induce migration of silver from conductors, leading to short circuits. HTCC MCM-Cs may be used without a separate package, since the fired ceramics are As discussed in Chapter 6, most LTCC ceramics probably are hermetic. hermetic also. The lowest cost MCM-Ls have simple epoxy encapsulation for protection, without a separate package [4], and thus are not hermetic. Recently, it has been realized that many nonhermetic sealing approaches, including plastic packaging, epoxy and gel encapsulation, are sufficiently resistant against contaminants and moisture to provide good reliability in many, even military, applications. This subject is currently under investigation. See Figure 1-4 for some possible encapsulation options.

The performance of a package is discussed further in Chapter 3.

1.2 FINDING YOUR WAY

The other chapters in **Part A - The Framework** provide further perspective on MCMs. They also provide bases for choices between MCM technologies from the point of view of manufacturability, performance and cost, in Chapters 2, 3 and 4, respectively. Chapter 2 provides an overview related to the materials and the manufacturability of MCMs. Specific examples of how some MCMs have been manufactured are presented. Chapter 3 covers the relationships between the needs of the system being packaged and the alternative technologies that exist ("need" being defined in terms of application or product goals, and being coupled with performance and cost as tradeoffs).

Chapter 4 focuses on MCM cost at the component level in keeping with the book's emphasis on MCM components. Two types of cost models are described. One is manufacturing activity-based (called technical cost modeling), useful when in-house manufacturing details are known. The other is a design activity-based model useful when the only manufacturing-related information available to the designer are vendor quoted prices, for example. Although many companies

16 INTRODUCTION



Figure 1-4 Hermetic and nonhermetic package encapsulation options. (Courtesy P. Franzon).

pursue a technology based primarily on previous experience, these chapters may provide the incentive to examine less familiar technologies.

In **Part B** - The Basics, MCM-L, MCM-C, and MCM-D signal interconnection technologies are discussed in some depth in Chapters 5, 6 and 7, respectively. Chapter 8 examines the dielectrics used with MCM-D. The several parts of Chapter 9 treat the physical die attach methods as well as the conductor connection technologies such as wire bonding, TAB, and flip chip methods similar to FCSB for the interior of the MCM. External connection of MCMs to PWBs is considered in Chapter 10.

The maximization of electrical (signal propagation) performance and thermal performance is considered in Chapters 11 and 12, respectively. MCM electrical testing issues are a big factor in overall MCM costs and are considered in Chapter 13.

The Case Studies in Part C consider various developments of actual products based on particular MCM technologies. The Unisys experience with MCM-C and MCM-D is examined in Chapter 14. Hughes MCM-C and MCM-D products are covered in Chapter 15. Experience with MCM-Si is considered in Chapter 16, and the DEC MCM-D experience is considered in Chapter 17. For examples of MCM-L products, the end of Chapter 5 may be consulted.

1.3 THE IMPORTANCE OF MATERIALS

Even a cursory reading of the various chapters in Parts A and B reveals the central importance of materials properties for MCM technologies. Many of the most important properties are highlighted in Table 1-3, which also shows some of the symbols and units. We mention some examples briefly in this section.

Numerical values for many of these properties are given in Chapters 5 through 8. For example, Table 5-1 lists the properties of dielectrics used in MCM-L, Tables 6-5, 6-7 and 6-8 show the properties of MCM-C materials, including HTCC and LTCC materials, and Table 7-3 gives the properties of dielectrics and conductors used in MCM-D. Tables 8-2, 8-4, 8-5 and 8-6 list important properties for various MCM-D thin film dielectrics, and Table 8-13 summarizes these properties for polyimides.

The electrical properties are most important for high frequency signal interconnections, where they affect the signal delay between chips and the spacing between conductors. These properties are introduced in Chapters 2 and 3, and are discussed in considerable detail in Chapter 11. For example, the speed of signal propagation in a dielectric is inversely proportional to $\sqrt{\epsilon_r}$, so higher dielectric constants, ϵ_r , cause longer delays.

Performance is degraded not only by slower propagation between chips, but also by distortion of square-shaped signal pulses. Major sources of this degradation are unintentional resistances, inductances and capacitances in the circuits between chips. These unintentional elements are called parasitic circuit elements. Some of these parasitics are caused by the first level connections; others are related to the materials in the common circuit base; still others are caused by second level connections.

Parasitic resistances, for example, are associated with losses in the metal conductors and in the dielectric of the common circuit base. These resistances increase the rise time of the pulse, which effectively increases the signal delay. Dielectric losses are proportional to the dielectric loss tangent, tan δ . The loss tangent (dimensionless) is sometimes called the dissipation factor, in which case it is normally given in percent.

Parasitic capacitance between conductors is due to the dielectric, and can cause crosstalk coupling of one signal into another. Parasitic mutual inductance is associated with the magnetic fields around current-carrying conductors and can

Table 1-3 Principal MCM Materials Properties.

ELECTRICAL PROPERTIES	THERMAL PROPERTIES	MECHANICAL PROPERTIES	PHYSICAL PROPERTIES	CHEMICAL PROPERTIES
Dielectric constant, ϵ_r	Coefficient of thermal expansion CTE (ppm ^o C)	Young's modulus E (GPa or kpsi)	Microstructure (grain size)	Metal oxidation
			Flatness and	Metal migration
T ace tangent	Thermol conductivity	Doitocar's antic 1	Planarization	Reactivity
tan ô	(W/m-K)	roisson s raulo v	Hermeticity	Adhesion
Resistivity (O-cm)	Shrinkace	Elongation	Melting point	Toxicity
	Thermal stability	Flexural strength (GPa)	Glass transition temperature, T _g	Environmental

also cause crosstalk. To keep the crosstalk acceptably low in high frequency interconnections, the conductor spacing must increase with increased ε_r . These factors are explained further in Chapters 3 and 11.

Thermal properties are important during the processing that forms the MCM and/or during the thermal cycling that occurs in MCM operation. Ceramic shrinkage after processing ("firing") and the thermal stability of organic dielectrics at processing temperatures are examples of dielectric properties important mostly for process design. A high thermal stability means that there is no significant outgassing or change in the mechanical dimensions that could lead to delamination.

Thermal conductivity is most important with high power single and multichip module operation. The thermal conductivity of each path from the chip to the outside world must be considered to maintain the junction temperatures on the chip at acceptable levels during operation. Poor heat transfer through materials with low thermal conductivity can be overcome by placing special conductors ("thermal vias") through the material, or by sending most of the heat through another path, such as a conductor attached to the "backside" (side away from the I/Os) of a flip chip connected die. Optimal design of these paths and also of heat transfer to cooling media such as air and water is examined in Chapter 12.

Differences in the coefficient of thermal expansion (CTE) between chips, common circuit base or package can lead to breakage either during cool down from a processing step or after extended thermal cycling. Whenever the temperature changes, connected parts with different CTEs expand or contract by different amounts, resulting in strain on the interface between parts. With large temperature changes, such as occur during processing, this might result in immediate failure, such as conductor breaks, or conductor peeling from a dielectric. Repeated smaller temperature changes, such as on/off operation in the field, eventually might result in thermal fatigue failures.

The units of CTE are parts per million per °C (ppm/°C). For example, with a CTE of 100 ppm/°C, the length of an unconstrained dielectric originally 10 mm long increases by 0.1 mm in a temperature rise of 100°C. Note particularly that the CTE itself can be a strong function of temperature, and can also be anisotropic (direction dependent). The CTEs (also called the thermal coefficient of expansion, or TCE) for some ceramics and organic dielectrics are tabulated in Chapters 6 and 8.

Some mechanical properties are important for their role in determining the response of materials to temperature cycling. Consider, for example, the bowing or camber of a silicon substrate beneath a thin dielectric. Excessive bowing following cool down from processing can make photolithographic conductor patterning difficult in MCM-Si. This bowing results from the different CTEs of

the substrate and dielectric, and is proportional to the ratio of the Young's modulus of the dielectric to that of the substrate. A large Poisson's ratio corresponding to an almost incompressible dielectric can also cause large bowing. These issues are discussed Chapter 7.

Conductors with high elongation and ceramics with high flexural strength can make a design more robust to thermal and mechanical cycling. In copper, the substantial plastic flow or elongation before breakage is a definite advantage. Similarly, solders that can plastically deform to accommodate CTE mismatch between chips and the MCM substrate can make flip chip solder bump (FCSB) technology more reliable.

Several units of stress for quantities such as Young's (elastic) modulus and flexural strength are in common use. Some equivalences are: 1 GPa = 1000 N/mm² = 145 kpsi, where GPa is gigapascal, N is newtons, and kpsi is 1000 pounds per square inch.

The physical and chemical properties listed in Table 1-3 are important mainly in processing. For example, a smooth and flat surface is important for high resolution conductor patterning. A ceramic with a large grain size may have unacceptably large surface roughness. In a multilayer structure, a polymer with good planarization can smooth out the features, or height variations, of layers that it covers.

Adhesion between metal conductors and polymer dielectrics is important in MCM-L and MCM-D processing. The glass transition temperature, T_g , of a glassy polymer is its softening temperature. Generally a lower T_g is beneficial for adhesion of the polymer to other layers and for stress relaxation, but is detrimental to dimensional stability. See the discussion in Chapter 8.

Copper reacts with polyimides and oxidizes at high temperatures. Therefore, copper in multilayer structures often is protected with another metal, such as nickel or chromium, which in turn should not react with the neighboring dielectric. Good conductors, such as copper, silver and gold, melt at temperatures required in HTCC processing, so lower conductivity metals must be used in HTCC structures. These and many other processing issues are covered in detail for each of the MCM signal interconnection technologies in Chapters 5, 6 and 7.

Hermeticity or near-hermeticity is important for long term reliability. This is particularly important for MCM-C devices that may have no separate package. Verification of LTCC ceramic hermeticity allows the use of relatively inexpensive silver conductors, which are susceptible to metal migration when exposed to moisture. Absorption of moisture in polymers may cause swelling and consequent breakage of attached conductors.

Toxicity of compounds of beryllium and alloys of lead under certain conditions, and environmental concerns over chlorinated fluorocarbons and some solvents, has led to searches for alternatives to these materials.

New materials also can make new high performance and/or cheaper technologies practical. For example, porous cordierite ceramic was developed for the interior of MCM-L laminates because it has a relatively high thermal conductivity and Young's modulus, and a relatively low dielectric constant and CTE. Some other new materials possibilities have been touched upon in this short review. Hopefully, readers of this book will be inspired to find new materials solutions to problems that impede the introduction of present and future MCM technologies.

1.4 THE IMPORTANCE OF MANUFACTURING PROCESSES

New MCM manufacturing processes to reduce costs are under active investigation, as can be seen from a reading of the chapters in Part B. These include blurring of the various MCM technology definitions described above.

For example, MCM-L may approach the thin film capability of MCM-D to pattern fine lines by using additive conductor processing instead of the subtractive etch processing traditionally used to pattern conductors on PWBs. The common circuit base is still a low cost laminate. Similarly, new developments of thick film screen printing emulsions and wires may allow fine conductor lines to be patterned on LTCC dielectrics, while at the same time new LTCC dielectrics have been developed with dielectric constants as low as those of the organic dielectrics used in MCM-D. In these ways, performance may approach that of MCM-D but at lower cost.

On the other hand, MCM-D costs can be reduced by increasing yields and using larger substrates. For example, patterning of somewhat wider lines and spaces on MCM-D substrates can lead to higher yields, since there are fewer large defects that can cause shorts and opens than there are small defects. Larger wafers (for MCM-Si) can become useable also. This is a simple example of improved manufacturability, where the yield is less sensitive to process imperfections. This type of manufacturability improvement is important particularly because it reduces costs even without the reduction in process imperfections that normally occur in proceeding up the learning curve to high volume production.

Just as with IC patterning, transfer of a new technology from a prototype lab to the production floor is critical. The production environment offers new challenges to high yields. Superior R&D efforts by U.S. companies have not been matched with equal efforts in transferring technology to production. Fast turnaround and accurate monitoring of factory production flow are essential.

High performance with high yields and accompanying low costs then requires creative and competent people from process design engineers to

22 INTRODUCTION

technology transfer experts to production technicians. An adequate supply of such people presupposes that manufacturing is viewed as an attractive alternative for students making career choices. Manufacturing competence is critical to a healthy economy [5], and we hope that readers of this book will even find that manufacturing can be glamorous and rewarding!

1.5 THE IMPORTANCE OF INDUSTRY INFRASTRUCTURE

In order for a product (the "final" product) to be designed and produced successfully, a number of intermediate products and services must be available. For MCMs, for example, these include the substrates, the chips, CAD tools and training in how to use them, test machines, etc. In principle, a company could provide such intermediate products and services from internal resources. Such companies are referred to as "vertically integrated companies." However, for most companies this would be inefficient. Thus, a portion of those intermediate products and services provide to the industry as a whole are referred to as the "industry infrastructure."

Naturally, the infrastructure needs of any one company depend on what it does not wish to provide from its internal resources. The most common case is the company whose only asset is design expertise. Such a company relies on others to manufacture the ICs, MCMs, PWBs and other assemblies that make up their designs, and to provide the information and tools necessary to carry out their designs. In order to implement high performance MCM-based products successfully, such a company needs the following infrastructure elements:

- One or, preferably, two or more suppliers who can manufacture the same "unpopulated" (without chips) MCM in quantity with high yields and with sufficient manufacturing capability to be cost competitive. Two suppliers are preferred to reduce risk of product starvation if one supplier has a temporary problem, for example. This is referred to as "multi-sourcing." True multi-sourcing is rare now, but most MCM designers are coping adequately with single suppliers. MCMs need to be delivered fully tested and guaranteed.
- One, preferably two or more suppliers of qualified bare die suitable for MCM mounting. By "qualified," it is meant tested, burnt in, and guaranteed to the degree that packaged die are. Generally, it is more efficient for the chip manufacturer to perform these tests than for the chip user. By "suitable," it is meant that die are provided in the following forms:

- Bare, suitable for wire bonding
- With a pad layout suitable for attachment to a standard TAB frame, or with a TAB frame
- With solder bumps, suitable for flip connection. Since die are easiest to solder bump in wafer form, this should be done before dicing and shipping.

One side effect of this requirement is that multiple suppliers provide chips with the same size, pad layout and electrical specifications. One supplier might suffice as long as the extra risk is acceptable. This is rare for chips, however. Chip production is very susceptible to production problems.

- At least two suppliers of the other required package components, such as TAB frames (if not provided by the chip supplier) and heatsinks.
- Manufacturers who can assemble all of these components, determine if they work correctly, and repair them, if needed. Providers of unpopulated MCMs also could offer this service. Standards are required so that automated assembly equipment can be developed.
- Computer-aided design (CAD) tools. These tools produce the MCM physical layout (locations of the chips and the interconnecting circuits, for example) and turn this layout into a suitable format for feeding to production machines. Ideally these tools automatically produce a design that functions correctly.
- Computer-aided engineering (CAE) tools to help in early design decisions, such as which MCM technology to use and to verify that the design will work. The verification step requires simulators to verify the logic design, the electrical properties of the interconnection and package structures, the thermal design, and the test plan. Other verification tools check that the manufacturing design rules have been followed and that the design will have no yield or other manufacturing problems.

All of these tools should be seamless in that the designer should not have to re-enter information that is described already in another tool. For example, the designer should be able to simulate an interconnection structure by "clicking" on it in the layout; he or she should not have to type in a simulation file.

• Computer-based libraries. These libraries should provide physical information, including die size and pad location, as well as information

24 INTRODUCTION

needed to simulate the design, such as logic models of the circuits and package structures, thermal models of heat dissipation and package properties, test models of the chip, and the MCM design rules. Again, these libraries should be seamless. The designer should not need to type in or move data between programs in order to do the design. For example, clicking of the interconnection to be simulated should allow appropriate models to be pulled from the libraries automatically.

• Information and training. This book is an element of the infrastructure!

As of today (mid-1992), not all of this infrastructure is in place, particularly in the United States and Europe. (Japanese companies tend to be large and vertically integrated, or to have such close relationships with supplier companies that they are effectively vertically integrated.) This picture is changing very rapidly. Next we summarize the current infrastructure and how it is changing.

Though many MCM foundries exist, only a few can manufacture MCMs in high volume. Those that can often are using lines that were previously internal to vertically integrated companies. In the future, more of the smaller companies need to be able to provide volume manufacturing of MCMs once the need for such high volume exists. The work of standards committees to develop standard MCM sizes and second level packages hopefully will enable a designer to obtain the same MCM package from multiple sources. This activity also should enable equipment manufacturers to produce equipment that can handle and assemble these MCMs automatically.

Until very recently, guaranteed MCM-suitable die were available only at a high price premium over the equivalent packaged part. Several service companies are providing these parts, buying wafers from the chip manufacturers, and testing them. One chip manufacturer recently announced the availability of tested, guaranteed bare die, at the same prices as the equivalent packaged part [6]. Hopefully, as demand increases and the cost of at-speed bare die testing and burn-in decreases, more manufacturers will offer this service at a price *less than* the equivalent packaged part.

Currently, obtaining multiple sources of bare die with the same pad locations and in the same size is difficult. Manufacturers may need to collaborate. For TABed (or TABable) die to become available, for example, standard pad sizes and locations are needed, as well as standard TAB frame sizes (see Chapter 9). For solder bumped die to become pervasive, more chip companies will need to license or develop a solder bump technology. One helpful development would be for a foundry to add solder bumps to die in wafer form obtained from chip suppliers.

CAD and CAE vendors provide many good computer tools. However, seamless operation with each other and with libraries is not quite there. More

agreement between the tool vendors is needed. Part of this agreement could be in the form of common frameworks for information exchange between tools. A small number of frameworks are emerging that provide this commonality. Seamless libraries require that the CAE vendors and the library providers (the chip and MCM manufacturers) agree to common formats for this information.

Achievement of much of the above infrastructure requires agreement in a competitive environment. Most companies recognize that this agreement is necessary for their own commercial success. A number of forums exist for the required consensus building effort. These forums have been provided through professional and industry service organizations, such as the IEEE, ISHM, IEPS, IPC, EIA, JEDEC, SRC; by the government (particularly DARPA, DOD-Air Force, Army); and by Consortia (particularly MCC and MCNC).

MCM vendors may copy profitably the approach of vendors of ASICs (Application Specific Integrated Circuits). Such vendors provide standard packages, high level CAD descriptions of their gate arrays, and quick turnaround. MCM equivalents would include: industry standards for new MCM packages; high level CAD descriptions of microprocessors, logic and memory chips; CAD aids for analyzing propagation delay, reflections, crosstalk, and final noise margin; and CAD thermal design tools. Much of the engineering can be done in advance of specific customer requirements, not only to produce faster turnaround, but also to improve reliability. Increased volume relative to total custom products also will provide a cost advantage.

Last but not least, training and information is needed. This book fulfills part of that mission. Conferences and seminars also provide part of this information. Universities have been slow in providing interdisciplinary courses in packaging technologies. This book will help overcome this by serving as a possible course text. Training courses for faculty, with government and industry-provided support to attend such courses, have been successful in the past for other important emerging technologies. The chip design industry, in particular, has benefited from this activity in the past. Now it is the turn of the electronics packaging industry!

1.6 DECISION-MAKING AS A PROCESS

With the introduction of MCMs, the number of packaging alternatives available to the designer, and the number of decisions that must be made, have increased substantially. The case studies presented in Part C of this book provide unique examples of the decision-making process. The descriptions there of the decision-making process emphasize the decision to introduce an MCM technology and then designing into it rather than just describing the reported results. In this section, we describe some general features of the decision-making process.

Engineering, managerial and marketing personnel together have to decide on the packaging alternative that best meets the need of their customer. An MCM technology provides a significant performance premium for the customer, but possibly at additional cost and schedule risk. Together with engineering staff, the manager needs to determine the impact of the different packaging technologies on performance and cost. Marketing and sales staff need to be provided with a clear understanding of what these new technologies are providing for their customers.

Knowing what packaging alternatives are available, we first investigate possible applications for MCM technologies by looking at the market.

1.6.1 Determining the Application: Possible MCM Markets

Though any such classification is somewhat arbitrary, it is possible to identify six categories of electronic systems:

- 1. Consumer Products: Included are consumer entertainment products, home appliances, and personal communications products, such as wireless (cellular) telephones
- 2. Aerospace and Military Products: Included are avionics, satellites, and military communications equipment
- 3. **Computers:** Classified broadly into the following subcategories according to applications and relative importance of performance and cost factors:
 - Low-end computers, such as PCs used for general office applications for example, word processing. Low-end computers can be defined as computers designed for users who wish to minimize cost.
 - *Mid-range computers*, such as workstations and servers used mainly for technical applications, for example, circuit simulation. Mid-range computers can be defined as computers intended for users who wish to maximize the ratio of performance to cost.
 - *High-end computers*, including supercomputers and mainframes used respectively for specialized scientific applications such as climate modeling and for applications requiring the rapid processing of large amounts of data such as airline reservation systems. High-end computers can be defined as computers intended for users who wish to maximize performance.

- Portable computing products, including notebooks
- *Peripherals* including input/output devices (scanners, printers, disks, screens)
- *Embedded computers*, such as machinery controllers and automobile computers
- 4. **Biomedical Systems:** Included are items such as office ultrasound machines, large CAT scanners and human-embedded devices
- 5. **Telecommunications:** Included are centrally provided equipment, such as switches, PBXs and their line cards, for example
- 6. Instrumentation: Included are oscilloscopes and test equipment

Each of these categories has somewhat different packaging needs.

The electronics parts usually found in consumer electronics do not have high pin counts (64 pins is a maximum typical count) and do not require high speed interconnection delays. Thus these systems usually use plastic single chip packages or, if small size creates a selling advantage, chip-on-board packages (the cheapest form of a bare chip mounted package). For example, most hand held calculators use a chip-on-board technology.

The majority of aerospace and military electronics systems are either signal processing or communications systems. Several examples are given in Chapter 15. In a signal processing system such as a radar processor, high circuit speeds are not usually required. Instead, additional system performance is achieved by having many chips working on the same signal at the same time. (However, there has been a recent trend to greater use of high speed chips in military electronics.) Though these chips may have only moderate I/O pin counts, the system size and weight can benefit enormously from MCM technology, and thus their use in this domain is common.

To obtain peak performance out of a computer, high I/O count parts must be connected with the shortest possible interconnection delay. (This is discussed in Chapter 3.) For example, the DEC 21064 RISC microprocessor, found in workstation products, is packaged currently in a 431-pin ceramic PGA and runs at 150 MHz. Even small computers tend to have high pin counts. For example, the Intel 386SL microprocessor, found in the current generation of notebook computers, is packaged currently as a 132-pin (LGA) and runs at 25 MHz. To pack these high pin count chips in a small space and to achieve small interconnection delays, most mid-range computer manufacturers are considering the use of MCM technology. Notebook computer manufacturers are seriously considering laminate (MCM-L) technology, as it provides considerable performance improvement, mainly through reduced size. Workstation manufacturers are considering both laminate and thin film (MCM-D)

msgalica@mintz.com

technologies, and several have constructed prototypes. Very low-end computers, such as PCs costing under \$1000 (1992 dollars), seek performance advantages but cost dominates and thus they use plastic single chip packaging.

High-end computer manufacturers have been using MCM technology, mainly ceramic (MCM-C) technology, for many years. As it is not currently possible to manufacture a high speed mainframe processor as a single chip, MCM technology enables high-end computer manufacturers to interconnect many chips and make a "virtual" large chip as an MCM. They require the highest performance out of MCM technology. Examples are given in Chapters 14 and 17.

Embedded computer applications, such as a washing machine controller, a printer controller, a fuel injection controller, or even an automobile navigation computer, tend to be low performance applications. As pin counts and clock speeds are not high, and space is not usually at a premium, advanced MCM technology has little to offer. However, ceramic hybrid circuits have been used for a long time in automobiles as a means to cope with the harsh environmental conditions. Plastic packages provide insufficient protection in this hot, hydrocarbon-filled environment, and hermetic hybrid packages provide a reasonable cost alternative.

Typical biomedical systems would be a computer aided tomography (CAT) scanner, or an implanted defibrillator. Much of the electronics in a CAT scanner is used for signal processing, similar to applications in military systems. However, since size is relatively unimportant, these systems do not need advanced MCM technology. Size is important in implanted systems, but the electronics parts in such systems have been neither high pin count nor high speed. Nevertheless, high pin count electronics parts are appearing in new implanted systems, and hybrid circuit packaging may no longer suffice.

Telecommunications equipment, such as large switches, tend to require both high speeds and large amounts of wiring. The telephone industry is looking very seriously at MCM technology for the provision of these functions.

Instrumentation such as oscilloscopes and test systems places unique demands on its electronics in that it must operate faster than the electronics in the system it is used to observe or test. Hybrid circuits often have been used in such systems and the use of newer MCM technologies is anticipated. The performance and cost considerations associated with these different systems are discussed also in Section 3.3.

1.6.2 Determining the MCM Technology: Business Decisions

Once a particular market application seems attractive for further consideration, many decisions must be made. Fundamental business issues include whether to

develop an internal capability or to use the contract services of another company, whether to try to change the technology with each new product, or to try to identify and develop a technology that will apply to several products.

Customer requirements for electrical performance, package size, unit cost and time-to-market must always be met. Such considerations dictate many of the choices in choosing an MCM technology. At the same time, there are always alternative choices for some aspects of every technology. It is important to recognize the cost and time-to-market advantages gained by exploiting any design or manufacturing infrastructure (and its accompanying expertise) that exists already, and that may be available within a company. For example, in a company with ready access to silicon IC design and fabrication resources, it probably will be cost effective to utilize an MCM-Si architecture. Similarly, assuming that the same desired performance specifications can be met, a company with skills in ceramics and thick film chemistry might initially pursue development of an MCM-C architecture.

On the other hand, if it is desired to do more than demonstrate the MCM concept, or to provide a limited number of application specific MCM prototypes, it is necessary to look at the longer range capabilities of a given MCM design to avoid being locked into an inflexible technology. It is important also to consider outside vertically integrated companies (IBM, DEC, Hughes, HP to name only some) that are willing and able to sell designs, components and functional subsystems. A readily accessible infrastructure greatly eases entry into the MCM business and provides a technology capable of meeting all the initial performance requirements. Such an infrastructure may be ill-suited for scaling up to large volume production or may be incapable of meeting expected future performance requirements. For example, a wire bond MCM-L architecture may represent a good route into the business, but an investment in flip chip MCM-D technology will have to be made eventually if exceptional high frequency performance is a long-range objective.

The importance of automated production using standard parts needs to be emphasized. As long as MCM technology development is focused on highly customized forms for small production runs, MCM costs will be high. Automation and standardization are necessary for cost-effective medium and large MCM product runs. This means the selection and refinement of MCM processes and designs appropriate for high speed automated batch processing, assembly and testing. As far as possible, these MCM technologies also must exploit the generic assembly and testing tools and standards that can be shared across a broad (industry wide) design and manufacturing base. In this case, even MCM-D may become a high volume cost effective alternative to VLSI integration and conventional packaging, and quickly claim its rightful share of the market!

1.6.3 Designing the Product: Multidisciplinary Engineering

With the growing need for aggressive packaging technologies, an interdisciplinary approach is needed for the engineering decision making process. Until recently, electronic package engineering was a discipline concerned mainly with manufacturing and reliability issues. Other engineers, separately, could take the package types offered by the package engineers and design products that met their required specifications. Today unfortunately, this strategy is no longer possible.

Key decisions in packaging involve detailed materials, manufacturing and reliability issues as well as electrical and thermal knowledge. Now design and systems engineers need to understand packaging technologies because package performance has become critical to their design function. Until recently, the choice of packaging technology was obvious once the chips were defined. Now this is not the case.

Even if a strategic direction in packaging technology has been selected, there are other subchoices and decisions still to make. Thermal design has become more intensive in nature. A different perspective is needed in electrical design. Managing the test process, for example, must be given a higher priority than previously. More effort must be spent on understanding materials and manufacturing issues as now their effect on cost, and their interaction with system function, is greater.

"Concurrent engineering" refers to cases where design and manufacturing decisions have to be made early and with the interaction of many different kinds of engineers. MCM product design clearly is one of these cases.

1.7 OVERALL PROSPECTS FOR MCMs

In this book, we consider not only various MCM alternatives, but also the other alternative: "*None of the Above.*" Why MCMs at all? Two major perceived impediments to the widespread production of MCMs are:

- **Design Time**: Designing an MCM generally takes longer than designing the equivalent PWB. Then why bother with the new complications of MCMs if single chips will have the same performance and density by the time an MCM system is finally brought to market?
- **Cost**: Even if the design time can be reduced, will sufficient investment ever be made in U.S. production of MCMs to make their costs competitive?

With regard to design time considerations, a number of points should be made. First, as MCM technology becomes more pervasive, the additional design time overhead will decrease to the point where it is not a serious factor, particularly with proper application of concurrent engineering techniques. Second, many MCMs contain a mix of semiconductor technologies (such as digital, analog, memory, silicon and GaAs) and cannot be replaced by a single chip. Third, large MCMs contain so many transistors that a single chip equivalent will take a long time to develop. Fourth, it does not make sense to design a custom chip for a low volume application. An equivalent MCM will often make sense, however. Finally, design time is not an important factor in many applications where product lifetimes are long, for example, in military applications.

Even in today's commercial markets, however, relatively long life cycles are possible. Systems based on popular microprocessors sometimes fall into this category, and it is encouraging to see that Intel, for example, is beginning to offer bare die of its popular chips [6]. Secondly, even large computer systems are gravitating towards parallel processing based on microprocessors. The size advantage of MCMs is attractive here, and a single modular design might be used over and over to add on processing capability. Another possibility for MCMs lies in designing systems such that the MCMs can be replaced later with improved chips having equivalent performance and I/Os without affecting the rest of the system [8].

With regard to costs, there is serious concern that MCMs will never be commercially viable in the U.S. without the commitment of long term investment. The case of Group 3 Fax technology is cited [9], where investment in the U.S. was withdrawn prior to the development of a large volume market that ultimately drove the costs down. Suggested solutions to this problem include direct government support and management enthusiasm for an initial high volume capability [9], an improved government climate for long term investment and better marketing forecasts [10]. These solutions, however, are not under the control of the individual small or medium size manufacturer.

Not only are initial MCM costs high because the technology is new, but also because there are new design and testing interfaces between chip makers, MCM producers, and systems houses [11]. Long term working relationships need to be initiated within the industry to reduce costs associated with these interfaces [10]-[11]. Costs may be lower in the long run if systems people pick qualified suppliers and give them some confidence in a stable market while helping them develop the needed standards and interfaces. Trying to maintain the usual lowest bidder approach can be counterproductive, because then no supplier may wish to invest in the volume production capability that will drive the cost down. Several chip makers (Intel, Motorola to name only a few) have taken the initiative, without waiting for support from systems houses. They have not only offered bare die of their popular chips, but they have started offering fully tested units ("known good die") [6]. Bare die testing eliminates an uncertainty leading to possible low yield of more expensive MCM units with several chips.

Ramp-up of MCM production may also occur fairly gradually. Since there are different possible systems and different MCM technologies, there is not the all-or-nothing quantum barrier of the Group 3 Fax technology cited above. MCM production may start out meeting market needs for "few chips packages" [12].

MCMs also offer the unique capability for optimized chip technology. No longer must different types of circuits be built on the same chip. Instead a single IC fabrication process can be used for an entire chip. For example, RAM processes can be used for RAMs, logic processes for CPUs, analog processes for CODECs and GaAs, if needed.

A promising niche for commercial MCMs in the short term is in systems where small size is absolutely essential, but the ultimate in performance is not [6]. One possible example is portable communication products [13]. The reduced interconnection complexity associated with MCMs also is attractive in these applications.

High performance systems where the speed is limited by the number of I/Os available on a single chip package are a longer term candidate for MCMs [14]. Reductions in the number of second level connections also should lead to higher reliability in these applications. MCMs have another unique advantage in their huge array contacting capability (see Chapter 18). Finally, MCMs can provide optimal functional performance, if signals only have to travel at high speed between chips within the same MCM package.

High performance alone, however, is not now a sufficient driving force for volume MCMs. Intel recently dropped its production of an advanced MCM for the reason cited at the beginning of this section: the next generation chip device came out with performance comparable to the MCM [6].

Many of these issues are discussed in more detail in Chapter 18. We hope that this cursory view sparks your interest!

1.8 SUMMARY

MCMs offer the potential for increased chip density leading to reduced size of electronic systems. Together with reduced size, MCMs offer a number of advantages. The speed performance is improved due to smaller chip spacings and reduced parasitics. Reliability is improved due to the reduction of the number of second level connections.

The successful use of MCM technology requires careful application. The alternatives and the economic issues that affect its use must be considered. Currently, MCM components are more expensive than the equivalent collection of single chip components and PWBs. Sometimes, an advanced technology custom chip will make sense over a small MCM, particularly a high volume application. If that solution does not make sense, then MCM technology must be considered seriously in any performance-driven application. In any case, a large system built out of MCMs might be less expensive actually than the equivalent single chip package system due to substantial savings in total size.

If MCMs are to be used in an application, full consideration should be given also to the effects of infrastructure, standardization and automated assembly on the part to be used. Careful, concurrently engineered design is required.

The main challenge to keep in mind while reading this book is how to realize these potential advantages of MCMs for applications you may have. Technologies must be developed and chosen so that the MCMs are not hindered from reaching their potential. In particular, an MCM should maintain the performance of its component chips with minimal degradation. The material in Part B of this book will help in developing the required technology. The examples in Part C will help in making the right technology choices. Part D will help to verify whether you have made the right choices. We wish you success in an exciting adventure!

Acknowledgments

The author expresses sincere gratitude to her co-editor, Dr. Paul D. Franzon for his helpful technical discussions and encouragement pertaining to this chapter and to the overall scope of the book. His technical collaboration in the goals of this book has been a most rewarding opportunity! She also acknowledges with sincere appreciation the informative, technical discussions and helpful suggestions of John Nelson, Unisys, John Segelken and T. Dixon Dudderar, AT&T Bell Laboratories and Walter Bertram, Alcoa Electronic Packaging. Their encouragement and support in completing this chapter have been of particular value also.

General References

These following books provide excellent reference material and more detailed information on topics related to packaging and covered in the book. 34 INTRODUCTION

- G1 R. R. Tummala, E. J. Rymaszewski, eds., *Microelectronics Packaging Handbook*, New York: Van Nostrand Reinhold, 1989.
- G2 D. P. Seraphim, R. Lasky, C. Y. Li, *Principles of Electronic Packaging*, New York: McGraw-Hill, 1989.
- G3 R. W. Johnson, R. K. Teng, J. W. Balde, eds., Multichip Modules: System Advantages, Major Construction, and Materials Technologies, New York: IEEE Press Reprint Book, 1991.
- G4 G. Messner, I. Turlik, J. Balde, P. Garrou, eds., Thin Film Multichip Modules, Reston VA: ISHM, 1992.
- G5 J. J. Licari, L. R. Enlow, *Hybrid Microcircuit Technology Handbook*, Park Ridge NJ: Noyes Publications, 1988.
- G6 C. A. Harper, ed., *Electronic Packaging and Interconnection Handbook*, New York: McGraw-Hill, 1991.
- G7 C. A. Harper, ed., *Handbook of Thick Film Microelectronics*, New York: McGraw-Hill, 1982.
- G8 M. L. Dertouzos, R. K. Lester, R. M. Solow, *Made In America: Regaining The Productive Edge*, New York: Harper Perennial, 1989.
- G9 K. B. Clark, T. Fujimoto, Product Development Performance-Strategy, Organization, and Management in the World Auto Industry, Boston MA: Harvard Business School Press, 1991.
- G10 J. W. Dally, Packaging of Electronic Systems (A Mechanical Engineering Approach), New York: McGraw-Hill, 1990.
- G11 H. B. Bakoglu, Circuits, Interconnections, and Packaging for VLSI, Reading MA: Addison-Wesley, 1990.
- G12 J. S. Hwang, Solder Pastes in Electronics Packaging, New York: Van Nostrand Reinhold, 1989.
- G13 J. H. Lau, ed., Solder Joint Reliability: Theory and Applications, New York: Van Nostrand Reinhold, 1991.
- G14 C. F. Coombs, Jr., Printed Circuits Handbook-3rd Ed., New York: McGraw-Hill, 1988.

The Materials Research Society holds week-long symposia on Electronic Packaging Materials Science at some of their meetings (April and November of each year). The papers from each Symposium are published in a hard cover volume. Some recent volumes, for example, are:

- G15 E. D. Littie, P. S. Ho, R. Jaccodine, K. Jackson, eds., *Electronic Packaging Materials Science V*, MRS Symposium Proceedings Nov. 1990, vol. 203, Pittsburgh PA: Materials Research Society, 1991.
- G16 E. D. Littie, R. C. Sundahl, R. Jaccodine, K. A. Jackson, eds., *Electronic Packaging Materials Science IV*, MRS Symposium Proceedings April 1989, vol. 154, Pittsburgh PA: Materials Research Society, 1989.

Elm Exhibit 2162, Page 64

msgalica@mintz.com

References

- 1 Publication IPC-MC-790, "Guideline for Multichip Technology Utilization", IPC (Institute for Interconnecting and Packaging Electronic Circuits), Lincolnwood, IL, 1990.
- J. J. Reche, "High Density Interconnect for Advanced Packaging," Proc. NEPCON West, (Anaheim CA), pp. 1308-1318, Feb. 1989.
 See also:

W. H. Knausenberger and L.W. Schaper, "Interconnection Costs of Various Substrates - The Myth of Cheap Wire," *IEEE Trans. CHMT*, vol. CHMT-3, no. 4, pp. 634-637, Dec. 1980.

3 A. J. Blodgett, Jr., "Microelectronic Packaging," *Scientific American*, vol. 249, pp. 86-96, July 1983.

A. J. Blodgett, Jr., "A Multilayer Ceramic Multichip Module," *IEEE Trans. CHMT*, vol. CHMT-3, no. 4, pp. 634-637, Dec. 1980.

For a more recent article giving the background as well as the latest information on the glass ceramic/copper multilayer substrates for the IBM 390/9000 high performance computers see:

A. H. Kumar, R. R. Tummala, "State-of-the-Art, Glass-Ceramic/Copper Multilayer Substrate for High Performance Computers," *Int. J. Hybrid Microelectronics*, vol. 14, no. 4, pp. 137-150, Dec. 1991.

- 4 L. M. Higgins III, et al., "Glob-Top Encapsulant Suitability for Large Wire Bonded Die on MCM-L," Proc. Int. Conf. Multichip Modules, (Denver CO), pp. 482-484, April 1992.
- 5 M. L. Dertouzos, R. K. Lester, R. M. Solow and The MIT Commission on Industrial Productivity, *Made in America: Regaining the Productive Edge*, New York: Harper Perennial (A Division of Harper Collins Publishers), 1989.
- 6 Electronic Engineering Times, May 4, 1992.
- 7 W. Blood, "ASIC Design Methodology for Multichip Modules," *Hybrid Circuit Techn.*, vol. 8, pp. 21-27, Dec. 1991.
- 8 J. L. Hennessy, "Trends in Processor and System Design and the Interaction with Advanced Packaging," *Proc. IEEE Multichip Module Conference MCMC-92*, (Santa Cruz CA), pp. 1-3, March 1992.
- 9 J. W. Balde, "Crisis in Technology: The Questionable Ability to Make Thin Film Multichip Modules," *Proc. of the IEEE*, Dec. 1992.
- 10 E. Jan Vardaman, "Economic and Political Implications of Disinvestment in the MCM Industry," *Proc. Int. Conf. Multichip Modules*, (Denver CO), pp. 71-73, April 1992.
- 11 F. Bachner, et al., "Defining the Interface between Suppliers and Users in the Multichip Module Marketplace," Proc. Int. Conf. Multichip Modules, (Denver CO), pp. 52-55, April 1992.
- 12 L. M. Higgins III, "Perspectives on Multi-Chip Modules: Substrate Alternative," Proc. IEEE Multichip Module Conference MCMC-92, (Santa Cruz CA), pp. 12-15,

36 INTRODUCTION

March 1992.

- 13 B. Freyman, B. Miles, "A DSP-based Multichip Module employing Advanced Multilayer Printed Circuit Board Technology," *Proc. Int. Conf. Multichip Modules*, (Denver CO), p. 490, April 1992.
- 14 W. M. Siu, "MCM and Monolithic VLSI: Perspectives on Dependencies, Integration, Performance and Economics," *Proc. IEEE Multichip Module Conference MCMC-92*, (Santa Cruz CA), pp. 4-7, March 1992.

MCM PACKAGE SELECTION: A MATERIALS AND MANUFACTURING PERSPECTIVE

Allison Casey Dixon and Edward G. Myszka

2.1 INTRODUCTION

Multichip packaging is receiving increased attention as electronic equipment manufacturers drive toward smaller, faster and less expensive products. By connecting several chips together in a single package:

- Board size can be reduced by up to a factor of 10 or more
- Signal propagation between chips can be up to three times faster
- The number of solder connections in a system can be reduced

Even so, multichip modules (MCMs) will be utilized only where they are the least expensive method of meeting system requirements. The choice of MCM materials and manufacturing processes greatly influences the cost of a multichip module technology in terms of piece part cost, manufacturing yield, manufacturing cycle time and repairability. Materials choices are also dominant factors in the electrical and thermal performance of a module. There is no single "right" choice; rather, different choices are appropriate for different applications.

This chapter presents the technology choices available today for the various parts of an MCM described in Table 2-1 - signal interconnect, substrate base, MCM substrate, package body, chip mounting and module level connection.

msgalica@mintz.com

Elm Exhibit 2162, Page 67

38 MCM PACKAGE SELECTION

Different combinations of these parts are used for different MCM technologies. Some of the typical combinations are sketched in Figure 2-1. Comparisons of manufacturing flows, materials properties, performance and cost are emphasized in this chapter, along with the concept that there are no "right" choices except in light of a specific application.

After the technology choices are described, global material and manufacturing related issues are discussed - cost, performance, thermal path, rework and manufacturability. The chapter concludes with some examples of modules designed for given applications.

PART	DESCRIPTION EXAMPLES		
Chip Mounting	Electrical and mechanical connection of chip to substrate.	Die attach/wire bond, TAB, flip chip	
Package Body	Additional structural support, environmental protection and signal connection to outside world.	Ceramic packages	
Substrate Base	Structural support for the signal interconnect. Signal interconnect may be deposited in the substrate base or the substrate base may be an integral part of the signal interconnect.	Silicon, ceramic, organic laminate, metal	
Signal Interconnect	Metal and dielectric patterns forming the circuitry between chips.	Copper/polyimide, tungsten/alumina, aluminum/silicon dioxide.	
MCM Substrate	Signal interconnect plus the substrate base, may require an additional package.	MCM-C, MCM-L, MCM-D, MCM-D/C, MCM-Si	
Module Level Connection	Electrical and mechanical connection of module to motherboard. Integral part of either the MCM substrate or the package body.	PGA, PAC, gull wing lead	

 Table 2-1
 Multichip Module Parts.



Elm Exhibit 2162, Page 69

msgalica@mintz.com

2.2 PACKAGE BODY AND SUBSTRATE BASE CHOICES

The package body and substrate base are the structurally robust piece parts that form mechanical support for the MCM. The package body is simply a housing for the module. Some module technologies require use of a package body, and others do not. In Figure 2-1, the MCM-D and MCM-Si modules will need a package body to house the fragile module and to provide electrical connection to the outside world. Typically, an MCM-D or MCM-Si module is wire bonded to a ceramic package body. The package body provides mechanical and environmental protection and a means to attach pins or leads to transfer electrical signals to the outside world. In some cases, a package body is not necessary because the substrate base, the structural support for the metal and dielectric patterns forming the circuitry between chips (signal interconnect), provides sufficient mechanical support and a means of electrically connecting the module to the outside world.

The various types of MCMs are sketched in Figure 2-1. Figures 2-2 shows photos of an assembled module and the MCMs packaged parts used to fabricate it. The MCM-L, MCM-C, and MCM-D/C modules make use of integrated MCM substrates that do not require use of a separate substrate base or package body. MCM-D and MCM-Si, on the other hand, utilize all three parts: substrate base, MCM substrate, and package body.

Three general classes of materials may be used for package bodies or substrate bases: ceramics, organic laminates and metals. Important characteristics of these materials are summarized in Table 2-2 and are discussed below [1].

2.2.1 Ceramics

Ceramics used in MCMs have several advantages. Foremost of these is that a properly designed ceramic piece part can serve as the module package body, the MCM substrate, and the module level connection — all in a single integrated part. Ceramics are electrically non-conductive, simplifying module design. They can form part of a hermetic package easily.

Alumina (also known as aluminum oxide or Al_2O_3) is the most common, inexpensive and widely used ceramic substrate material. Alumina may be used in an MCM as either an MCM substrate or as a stand alone package body for MCM-D or MCM-Si type modules. There are two general forms of alumina used – multilayer cofired alumina, which can contain conductive metal traces or planes, and single-layer pressed alumina. Multilayer cofired alumina is made from individual layers of unfired material (called "green tape") which can be patterned with conductive metal traces. Many layers (up to 50 or more) can be stacked together and fired at high temperatures to form a single unit. Further

PACKAGE BODY AND SUBSTRATE BASE CHOICES 41









b

Figure 2-2 MCM-Si package showing (a) chips on silicon substrate forming module, (b) ceramic package base and (c) assembled MCM package.

msgalica@mintz.com Elm Exhibit 2162, Page 71

42 MCM PACKAGE SELECTION

	Formula	Cost	Availability	СТЕ (10 ⁻⁶ /°С)	Thermal Conductivity (W/m-K)
Pressed Alumina	Al ₂ O ₃	Х	Excellent	8.1	30
Aluminum Nitride	AIN	4X	Poor	4.3	260
Silicon Carbide	SiC	2X	Poor	3.7	270
Mullite	3 Al ₂ O ₃ / 2 SiO ₂	Х	Poor	4.5	2-6
Glass Ceramic	Various	х	Fair	3.0-4.2	5
Organic Laminate	Various	0.5X	Excellent	12.0-17.0	0.2-0.3
Metals	Various	0.3X	Good	6.0-20.0	200-400

Table 2-2 Properties of Package Bodies/Substrate Bases.

Note: Silicon, CTE = 3.5×10^{-6} /°C.

processing details are discussed in Section 2.3.1 of this chapter and in Chapter 6. Cofired alumina is a mature technology with highly automated factories in place. Its major disadvantages are shrinkage and warpage. These factors are controlled tightly in the manufacturing process and compensated for in the design process. Shrinkage and warpage limit the density of conductive metal traces on a cofired ceramic MCM due to the additive tolerance of each layer. Shrinkage and warpage also cause problems with mounting high lead count, tight pitch chips in cases where the planarity of connection points is critical. Figure 2-14b (discussed in Section 2.7.2) is an example of a prototype workstation module utilizing cofired ceramic as a substrate base.

Pressed alumina is a single sheet of ceramic material that is independently fired. Pressed alumina does not contain conductive traces or planes and serves strictly as a package body for MCM-D or MCM-Si modules or as a substrate base for MCM-D/C modules.

As shown in Table 2-2, alumina is a poor conductor of heat relative to other ceramics. It is, however, orders of magnitude better than organic laminates. Alumina has a coefficient of thermal expansion (CTE) more than twice that of
silicon. When silicon chips are mounted on alumina, thermal expansion mismatch creates stress that can cause material fatigue, leading to premature failure.

New ceramic materials aimed at improving CTE and thermal conductivity are currently in prototype production. Some of the most promising materials such as aluminum nitride, silicon carbide and mullite, are listed in Table 2-2 for comparison with standard alumina.

2.2.2 Organic Laminates

Organic laminates, commonly referred to as printed circuit board (PCB) or printed wiring board (PWB), are used in many different forms for electronics packaging. In MCMs, organic laminates are used most often as an integrated package body and MCM substrate. In general, sheets of polymer materials (polyimide, FR-4, BT resin plus glass reinforcement) are sandwiched between layers of metal (usually copper) traces. The stack up is then bonded by lamination in a hydraulic press or autoclave under heat and pressure (350°F and 325 psi for standard FR-4). Further processing details for laminate-based MCM technologies are described in Chapter 5.

Like ceramics, organic laminates have the advantage of combining package base and MCM substrate functions into a single piece part. Another advantage of organic laminates is the match of their CTE to the mother board. Any MCM will be mounted eventually on a PWB. The organic laminate-based technology typically has the lowest manufacturing cost of all module technologies because it is mature and employs both batch and parallel processes (see Section 2.3.2). The major disadvantages are very poor thermal conductivity, a CTE four to five times that of silicon, and warpage. Figure 2-3 is a photograph of an organic laminate MCM substrate, consisting of seven copper layers, BT resin dielectric and an embedded leadframe.

2.2.3 Metals

Metals also are used as package bodies for MCMs [2]. Often metals are chosen for high power MCMs because of their excellent thermal conductivity. Metals also provide a very rigid and flat surface for signal interconnect layers within the module. The major disadvantage of using metals as package bodies is that signal connections cannot be made directly through the metal to the outside world.

2.3 SIGNAL INTERCONNECT AND MCM SUBSTRATE CHOICES

Module signal interconnect refers to the pattern of metal and dielectric that forms the circuitry between chips and from the chips to the outside world. As



Figure 2-3 Organic laminate MCM.

discussed in Chapter 1, there are many MCM substrate types available. MCM-C, MCM-L, MCM-D, MCM-D/C and MCM-Si, as well as conventional thick film hybrids, are all available today. The MCM designer is faced with many choices for the fundamental technology, as well as in the selection of materials used within each of these technologies. The selection of the dielectric material, as well as the conductor material, also strongly influences the overall performance of the module. This is true especially for modules containing high speed digital and mixed analog/digital components with switching frequencies in excess of 50 MHz or switching rise times less than 2 ns. This section focuses on the description of the core MCM technologies and processes and the materials which are available to manufacture such MCMs.

There are many materials used as conductors for MCM applications. Copper (Cu) and aluminum (Al) are utilized as conductors for MCM-D, MCM-D/C and MCM-Si applications. Copper is preferred for its lower resistivity. Copper is being used also in experimental MCM-C modules fabricated at low cofiring processing temperatures. Gold (Au) is used occasionally for MCM-D and MCM-D/C, but its relatively high cost prohibits its use in most commercial applications. Refractory metals such as tungsten (W), molybdenum (Mo) and manganese (Mn) are used in high temperature cofired ceramic processing. These metals are selected for their high melting points rather than for their electrical properties. Metals such as W, Mo and Mn are electrical conductors able to withstand the

high (1500°C) firing temperatures required to sinter the ceramic layers together to form a monolithic multilayer MCM-C substrate.

2.3.1 MCM-C (Cofired Ceramics)

Multilayer ceramic structures have been designated as MCM-C. Further details on ceramic-based technologies are given in Chapter 6. This type of MCM can be categorized in two groups: those processed at high temperature (HTCC) and those at low temperature (LTCC). HTCC has been available for decades and is used most commonly [3]. Refractory metals such as W, Mo, and Mn are used as electrical conductors in HTCC processing. These metals are selected because of their inherent high melting point. These metals remain stable and do not decompose during the sintering process where temperature extremes can reach (1500°C). Low temperature cofiring, although fairly new to the industry, has generated a great deal of interest. The ability to use highly conductive noble metals such Ag, Au and, most recently, Cu as the conductor offers many advantages over HTCC modules [4]. Unfortunately, since this technology is quite young and its penetration into the market relatively small, the substrate cost can be as much as 70 - 100% higher than HTCC substrates. As the market penetration increases, this disparity is anticipated to decrease. In the cofire processing for both the high-and low-temperature technologies, a liquid slurry is formed from ceramic particles and organic binders and then cast into a solid sheet. This sheet is often referred to as the "green tape" because of its unfired state. Typical costs for moderate production volumes associated with green tape are \$0.06 and \$0.11 per square inch for HTCC and LTCC, respectively [5].

Next, holes for vias are generated in the green tape. The most commonly used via diameter sizes range from 0.015 - 0.008" with more aggressive designs reaching 0.004". Vias can be drilled, but are formed more commonly by a punching operation. Punching can be accomplished by a single punch head positioned sequentially at each site by computer numerical control (CNC). This often is referred to as "soft tooling." Alternatively, a custom die head that simultaneously punches all the vias on a single layer is referred to as "hard tooling."

Soft Tooling versus Hard Tooling

The nonrecurring engineering cost (NRE) for hard tooling is higher than soft tooling, but the unit price is typically lower. The decision on which type of tooling to be purchased is a function of the anticipated volume. For low volume runs, soft tooling NRE, which

46 MCM PACKAGE SELECTION

can average \$30,000 for an eight layer module, can be cost effective. Conversely, for high volume applications the cost of hard tooling, which can exceed \$60,000, can easily be amortized over the life of the product to produce a cost effective solution. One disadvantage of soft tooling is that there typically is a maximum run rate associated with the tool. That is, there exists a maximum number of modules that are built and shipped each month. This is typically associated with the existing fabrication throughput capabilities.

After the green tape sheets are formed and via holes drilled or punched, a conductive ink (a refractory metal for HTCC MCM substrates or a noble metal for LTCC MCM substrates) is applied to each layer through a screen printing process (see Chapter 6). 10 mil lines are standard and 4 mil lines are readily fabricated using this process. Vacuum can be applied to the underside of the punched sheet to pull the ink into the hole to coat the side walls or, in many cases, to fill the via. Solid vias are advantageous for high density applications because they can be stacked directly on top of each other, otherwise the vias will be staggered and hence require additional area.

After patterns are screened on each layer, the layers are stacked and the assembly is fed into a furnace and fired at temperatures above 1400°C for the high temperature process and approximately 800°C for the low temperature process. The result is a monolithic structure containing all the interconnects.

During the firing process the organic binder decomposes, the ceramic densifies and, unfortunately, the structure reacts by shrinking. This shrinkage becomes a problem during the subsequent assembly of chips where precise positioning of the chip connections is required. This is true especially of high density tape automated bonding (TAB) and flip chip interconnects where the I/Os are fixed. If wire bonding is used as the chip connection method, an automatic wire bonder with look-ahead vision (an optical pattern recognition system that determines the precise location of bonding pads used to make small adjustments to accommodate substrate shrinkage) can be used. Another solution is to deposit a thin film layer of metal or copper/polyimide after firing the ceramic to facilitate chip connection of high I/O devices. This postfire processing can be provided only for the outermost surface of the module.

Thermal solutions in an MCM are accommodated typically by placing vias strategically under the chips (thermal vias) and/or by brazing a Cu/W heatsink to the ceramic structure. Figure 2-4 shows a single chip multitiered cofired ceramic package with an integral Cu/W heat spreader designed to dissipate up to 30 watts. These features can be implemented readily for multichip packaging, but they can come with both cost and electrical performance penalties.



Figure 2-4 Single chip multitiered cofired ceramic pad array carrier (PAC) package.

One of the major disadvantages of cofired alumina is its relatively high dielectric constant (ε_r) which can limit the performance capability of MCMs (Section 2.6.2). Since the performance of a module is influenced heavily by design layout as well as by material selection, no rigid cut off exists for using cofired ceramic technology. Modules have been built and tested at clock frequencies in excess of 50 - 75 MHz. New ceramic materials with lower dielectric constants, such as glassy ceramics (Table 2-2), have been developed for modules operating in excess of this. The next section presents a brief description of several of these materials.

Low Dielectric and Glassy Ceramics

The most well known glassy ceramic was developed by IBM. These ceramics have low cofiring temperatures which make them co-sinterable with either Cu or Au conductors. Additionally, the materials have a low dielectric constant and mechanical properties which may be tailored for a particular application. Others are attempting to modify the material properties by introducing particles suspended in the ceramic matrix. Upon sintering, or through a post annealing

48 MCM PACKAGE SELECTION

treatment, porosity is introduced to the matrix [6]. Since the dielectric constant of the gaseous material produced by the porosity is very low, the effective dielectric constant of the assembly is also low. Experimental data on research grade materials suggest that the effective dielectric constant can be reduced to approximately 2.5. These materials hold great promise but first must prove compatibility with metal films and the ability to withstand the environmental testing required to become a reliable substrate material.

High Thermal Conductivity Ceramics

Attention is being focused also on the development of ceramic materials with higher thermal conductivities than alumina. Besides alumina, beryllia (BeO) is the most widely used ceramic in the microelectronics industry, primarily because of its high thermal conductivity. Although cofired structures have not been developed for this material, it has been used extensively as a heatsink for semiconductor lasers and as a substrate base for high power RF applications. The greatest disadvantage of BeO is in its processing. BeO dust and particulates have been designated as carcinogens and, thus, extreme care must be taken when cutting or grinding this material. As a result, aluminum nitride (AIN) has been introduced as an alternative to BeO for MCMs (Table 2-2). AIN has 80% of the thermal conductivity of BeO and can be cofired in a multilayer structure [7]. With these beneficial properties, it is anticipated that AIN will become a cost effective alternative to BeO in the near future.

2.3.2 MCM-L (Organic Laminates)

The MCM-L structures can be regarded as a laminated PWB scaled to meet the requirements and dimensions of a MCM as discussed in Chapter 1. Fabrication methods vary among manufacturers, but, in practice, most techniques closely follow the infrastructure previously developed by the PWB industry. The following section is intended to provide the reader with a brief description of the most widely used fabrication methods, materials and their associated advantages and disadvantages.

Many organic laminate types are used to manufacture modules for various applications. Epoxy glass (such as FR-4) is most common. For applications where material stability is required at higher temperatures, materials such as Bismaleimide triazine (BT Resin) and polyimides with higher glass transition temperatures (T_g) can be used. Laminates can be double clad, single clad or not clad at all. Cladding is thin copper foil applied to the sheet of dielectric material. It is this foil that is patterned to make the conductors. Double clad and single clad refer to the number of sides of the dielectric material which are coated with copper foil. The clad sheets are called "laminates." Several clad

sheets stacked and bonded together are also called "laminates." Copper foil thickness is measured in ounces. For example, a one ounce copper clad laminate will measure 0.0014" thick. One ounce and one-half ounce copper are typical for the PWB industry and one half, one quarter and even less than one eighth ounce can be found in laminates for the MCM-L industry.

Laminate Sheet Types

Copper clad laminates are available in widths in excess of 24" and are fabricated most commonly by either of two methods. Adhesive laminates rely on the application of a copper foil to a sheet dielectric (bare laminate with no metal) with an adhesive layer (typically acrylic in nature). These materials are clad in a rolling press at very high volumes. Conversely, adhesiveless laminates are fabricated by applying the copper directly to the dielectric with no bonding agent. Adhesiveless laminates can be formed in one of two ways: liquid dielectric may be deposited to the foil and cured, or the metal may be electroplated or vacuum deposited to the fully cured laminate material.

Additive versus Subtractive Processing

Copper conductors are patterned using either a subtractive or an additive process. The subtractive process is used typically with thicker copper films. In the subtractive process the pattern is etched directly into the existing metal. Subtractive processing is used predominantly in the PWB industry. Minimum feature sizes typically are limited to 0.003" - 0.004" due to the isotropic nature of the conductor etching process and the dielectric surface roughness.

Alternatively, in the additive process, a photoresist coating is applied and patterned onto the existing thin metal foil. The actual conductor pattern is electroplated onto this foil, the photoresist is stripped and the thin foil is etched away revealing the electroplated conductor. Refer to Figures 7-12 and 7-13. Typical conductor patterns can range in size from 0.001" - 0.003". It is predicted that additive processing in conjunction with smoother laminate surfaces, will enable conductor features less than 0.001".

After the clad laminates are patterned, vias are drilled on a CNC machine. To increase process efficiency, many laminates are often stacked during the via drilling process on machines with multiple spindles. Currently, minimum via size is approximately 0.004" - 0.006" in diameter and is limited by the

50 MCM PACKAGE SELECTION

availability of reliable drill bits. The copper layers can be patterned sequentially on a single clad laminate or two adjacent layers may be patterned in parallel on a free standing double clad film. The ability to pattern both sides of the laminate simultaneously is a major advantage of MCM-L because it effectively reduces the manufacturing cycle time.

After vias are formed and the Cu layers patterned, each layer can be inspected, an advantage of parallel processing. The inspection of layers prior to final lamination minimizes the possibility of the transfer of defects into the completed module. This early inspection for defects will increase the likelihood of receiving a functionally working substrate. The yielded copper layers are stacked with prepreg (partially cured laminate material) inserted between each layer. The stack is laminated in a press or autoclave with heat and pressure as mentioned previously.

The major disadvantages with organic laminate substrates used today are low routing density, poor thermal conductivity through the substrate and a high CTE, compared to silicon. New materials are being developed to alleviate many of these problems. These materials include advanced polyimides, aramids and fluoropolymers with homogeneous matrices and composite laminates. The electrical (dielectric constant) and mechanical (CTE) properties of many of these materials can be tailored for specific applications [8]-[9]. As the industry progresses to larger ICs with more demanding chip connection methods, such as flip chip, these materials will draw more attention.

For MCM-L applications one noticeable change in the fabrication of the laminate structure is the use of thinner dielectric films. These thin $(25 - 50 \,\mu\text{m})$ films allow controlled impedance, high density modules to be fabricated with transmission line design criteria. These fine line capabilities contribute also to the increase in routing density. However, large metallized pads on each layer are required to aid in the via drilling operation. To truly achieve high density modules novel approaches to via generation also will be required. Processes under consideration include wet chemistry etching, reactive plasma etching and laser ablation techniques. Each of these approaches to the fabrication of small vias is anticipated to break the current 0.004" barrier set by the mechanical drilling process and will increase significantly the routing density of MCM-L substrates. Unfortunately, many of the conventional laminates used today are not compatible with these advanced via processing techniques. For example, when using an excimer laser, since the ablation threshold of the glass weave used in conventional laminates is more than an order of magnitude greater than that for a polyimide matrix, processing of vias is complicated. The glass weave thermally decomposes at the high energy densities required to remove it. Similarly, the acrylic adhesives used to bond copper foils to dielectric sheets also have high ablation thresholds. Laminate suppliers have recognized the limitations of conventional laminate materials and have begun to offer new, more advanced laminates specifically targeted for such applications. Many of these new laminates are adhesiveless and do not contain a glass fiber weave. Instead, these new materials are made up of the pure matrix material or contain small particulates dispersed within the matrix to control its properties and processability. As the industry progresses to more advanced via fabrication techniques, we predict increased use of new laminate materials.

In most cases, laminates are fabricated in a large panel. At the end of production, the layers are cut out of the panel. The ability to fabricate multiple modules in a large panel format is cost effective. However, as the panel size increases, the layer-to-layer registration becomes more difficult. Parallel processing offers the potential to integrate two distinctly different laminate patterning processes: conventional PWB for low density layers (power and ground planes), and advanced fabrication processing, such as laser ablation for high density (signal) layers. Since layer-to-layer registration is less of an issue with low density layers, large panel sizes can be processed. These large panels can be cut into smaller, more manageable panels and mated to panels containing the higher density layers during the lamination process. Additionally, recent advances in the lamination of the multilayer organic structures to leadframes has made it possible to produce leaded MCM-L substrates configured as quad flat packs (QFPs).

2.3.3 MCM-D (Deposited Dielectric)

MCM-Ds are fabricated by a sequential deposition of conductor, typically Cu or Al, and dielectric layers (typically polyimide) on a substrate base made of ceramic, silicon or metal. As the fabrication of MCM-L parallels the PWB industry, MCM-D most closely parallels the processing techniques of the semiconductor industry. Semiconductor type processing provides for very fine lines with high routing densities. However, the semiconductor processing equipment used to manufacture MCM-D substrates is expensive and low volume runs make it difficult to depreciate the high capital costs. MCM-D substrates are fabricated in 4", 5" or 6" round or square "wafers." Fabrication costs are associated primarily with the wafer and not with finished substrate size; that is, a single substrate on a wafer will cost about twice as much as a design where two substrates can be fit onto a single wafer. The cost driver is the number of substrates that can fit onto a given wafer size (number "up").

Another disadvantage of MCM-D technology is that it often requires the use of a separate package body to house the MCM substrate and to make electrical and mechanical connection to the mother board. This results in additional

52 MCM PACKAGE SELECTION



Figure 2-5 Process flow: MCM-D.

assembly operations and higher manufacturing costs. A more cost effective approach to MCM-D is its integration with cofired ceramic. These structures are detailed in Section 2.3.4 on MCM-D/C.

Although MCM-D fabrication processes differ from one manufacturer to another, a common method is presented to give an idea of the processing required to build such a module. Figure 2-5 is a flow chart of a typical process.

A liquid polymer (typically polyimide, but also benzocyclobutene (BCB) and other fluoropolymers) dielectric layer is deposited on the substrate base by a conventional spin coating process. The liquid dielectric material is dispensed onto the center of the substrate and spun until the material spreads uniformly to cover the substrate. Although this is an effective process in terms of uniformity, it is far from efficient. As much as 50 - 80% of the material is spun off and wasted. As typical costs for the polymers can range from \$1.00 to \$2.00 per gram, and up to 4 grams are required to form a 12.5 μ m thick film layer on a 4" × 4" substrate, more efficient, alternative dispensing techniques are needed. Alternative deposition methods under investigation include spraying and extrusion. Dielectric thicknesses of 25 μ m or less are typical, and multiple coats usually are required to obtain this thickness with good planarization.



Figure 2-6 MCM-D via formation techniques.

Vias can be formed by scanning laser ablation, reactive ion etching or wet etching of the dielectric, as well as through the exposure of photosensitive polymers. Figure 2-6 shows the comparison of process steps required to form vias. No one method has been accepted as a standard in the industry [10]. Reactive ion etching of vias is commonly used in the U.S. and wet etching in Japan. These techniques are discussed further in Chapter 7.

After the dielectric layer is deposited and vias formed, metal conductors are formed using either additive or subtractive processing. In additive processing, a bus layer metallization for electroplating is first formed either by sputtering or by electroless copper plating. Photoresist is then spun on, the pattern imaged and the bulk pattern is electroplated. The photoresist then is stripped and the bus layer metallization etched away, leaving the electroplated conductor pattern behind. In subtractive processing, the full surface is coated with metal and the conductor pattern etched. The entire process is repeated for each layer. Both subtractive and additive processing are practiced today.

2.3.4 MCM-D/C (Deposited Dielectric on Cofired Ceramic)

Deposited dielectric on cofired ceramic, MCM-D/C, can cost effectively combine the advantages of MCM-C and MCM-D. This technology utilizes deposited dielectric layers on top of a multilayer cofired ceramic substrate base to produce MCM substrates with all of the high density and high frequency attributes of MCM-D coupled with the flexible physical attributes of MCM-C. This combination provides a single substrate that functions as the signal interconnect, substrate base, package body and module level I/O connection. Vias may be brought out the bottom of the substrate, the package I/Os may be terminated in a pad array carrier (PAC) format or mated with pins to simulate a standard, single chip pin grid array (PGA) package. The cofired substrate is part of the package body, eliminating the cost associated with repackaging of the MCM substrate. The photo in Figure 2-14b is an example of a module that uses a MCM-D/C substrate. In this module there are four layers of polyimide dielectric and thin film copper deposited on a standard 299 pin cofired ceramic substrate.

Another advantage of MCM-D/C technology is that signal lines can be embedded in the low dielectric constant deposited layers to minimize crosstalk, while power and ground planes can be contained in the higher dielectric constant ceramic layers. With the development of high dielectric constant ceramic materials ($\epsilon_r > 200$), it may be possible to incorporate much (if not all) of the decoupling capacitance within the substrate and eliminate the secondary assembly operation required to mount external bypass capacitors.

2.3.5 MCM-Si (Inorganic Thin Film)

Yet another solution to multichip packaging utilizes semiconductor fabrication techniques directly. This technology, often referred to as silicon-on-silicon (MCM-Si), is processed in a similar fashion to conventional silicon ICs. As discussed in Chapter 16, a silicon wafer is used as the substrate base, Al or Cu as the conductor, and silicon dioxide as the inorganic dielectric media. Several microns of the metal conductor and the subsequent silicon dioxide dielectric layers are deposited by a vacuum deposition technique (such as evaporation or sputtering). Layer thicknesses are controlled quite uniformly over areas as large as 6" - 8" diameters. Photoresist is applied and patterned with the support of a wafer spinner and mask aligner, respectively. The metal is etched away by either wet etching or, more typically, by dry etching techniques. Because the silicon wafer substrates are extremely smooth and flat, very fine feature sizes are possible. Although submicron interconnects are possible, typical features of 15 μ m are adequate for MCM interconnects.

MCM-Si offers the advantages of the highest signal interconnect density, excellent CTE match to the silicon die and the utilization of existing semiconductor fabrication infrastructure. The equipment used is readily available and the processes have been established and well characterized by the semiconductor manufacturing community. The utilization of this infrastructure has contributed to a reduction in fabrication cycle time (and learning curve) for many entering the MCM field along with the security of the predictable reliability of conventional ICs. Additionally, the inherent low density of silicon makes it ideal for applications in which light weight is critical, such as in the aerospace and military industries.

However, MCM-Si technology also has some significant disadvantages. Foremost among these are that silicon substrates are not a suitable MCM package base. In some cases, the high resistivity of the aluminum conductor precludes its use in high frequency applications. In addition, as with MCM-D, the equipment costs are high. Batch sputtering systems can range from \$150,000 to \$350,000 and in-line systems from \$1,500,000. In combination with low volume manufacturing, typical module costs will be high. Silicon-based technologies are discussed further in Chapter 16.

Other Inorganic MCM Materials

The majority of the industry has focused on the development of organic dielectric materials (MCM-D and MCM-D/C). For the short term these will be the materials of choice. However, in the long term, some inorganics may offer performance benefits that surpass that of today's organics. Research workers have focused attention on synthetic diamond and diamond-like coatings [11]. Although the development of cost effective deposition methods requires additional work, the superior properties of the diamond-like films warrant investigation as dielectric layers for high density interconnect structures. These materials typically combine desirable properties such as a low dielectric constant, extremely high thermal conductivity (more than four times that of copper) and a low CTE. These properties are desired for high frequency, high power and low residual stress applications, respectively.

2.3.6 Thick Film Hybrid MCM

Ceramics have been used in the microelectronics field for many decades. Various ceramic materials exist, but alumina (aluminum oxide) is most widely used as a substrate for the hybrid microelectronics industry. Some argue that the development of thick film hybrids was the origin of the first MCM-type interconnect. In this type of interconnect system, a flat sheet of alumina is typically used as the substrate material. Each layer is applied sequentially by screen printing, as described in Chapter 6. Although this is not a complicated process, one disadvantage with any sequentially fabricated module is layer-to-layer inspection. Since these layers can only be inspected once they have been applied to the module, there exists a potential for a defect in the final layer causing the entire module to be defective. This is unlike parallel processing (for

MCM-L, MCM-C) where all the layers can be fabricated independently and inspected. Many new processes under development either use parallel processing guidelines or minimize the total number of layers.

The substrate is then placed into a furnace where the ink is dried and fired. Conductor geometries are typically limited by the fabrication of the screens used for the printing process to 0.005" lines and spaces.

The dielectric layer is applied in a similar fashion. No dielectric material is applied in regions where vias lead to the above layers. When the subsequent conductor layer is printed, the ink flows into the openings of the dielectric and make contact to the conductor below. The balance of the module is completed by repeating this process.

Thick film hybrid is a mature technology with good infrastructure. A major advantage of this technology is that a large supply of resistive inks are available that can be screened onto the module to integrate resistors. Additionally, there exists an abundant supply of surface termination inks with metallurgies that facilitate wire bonding, soldering or component attachment through conductive epoxies.

Thick film hybrid technology used for MCMs has some serious disadvantages in the areas of performance and routing density. The dielectric constant of most dielectric inks is as high as 8. While high dielectric constant is not in itself a disadvantage, it can make a desired substrate characteristic impedance difficult or impossible to attain. Thick film hybrids are very limited in routing density - typically 0.005" - 0.010" lines and spaces and via sizes of approximately 0.008". From a module yield point of view, it is advantageous to minimize the number of layers. One way would be to increase the routing density so that all signal interconnects could be made on a minimum number of layers. Recent research and development efforts in transfer printing have claimed results of 0.001" feature sizes. Although this process has yet to be implemented in a production environment, it holds promise for the future.

2.4 CHIP MOUNTING CHOICES

The purpose of any chip mounting technique is to provide a suitable path for electrical signals from chip to MCM substrate and to provide a means of mechanically attaching the chips to the substrate. All chip mounting techniques also provide a path for heat generated in the chip to dissipate. The common of chip mounting choices are wire bond/die attach, TAB, flip TAB, and flip chip. Further details of these chip mounting or chip connecting techniques are presented in Chapter 9.

The method of connecting chips to a substrate is probably the most critical choice in module physical design. This choice largely defines the technical

	Die Attach Wire Bond	ТАВ	Flip TAB	Flip Chip
Cost	х	> 2X	> 2X	0.8X
Chip Availability	Excellent	Fair	Fair	Poor
Reworkability	Poor	Fair/Poor	Fair	Good
Probe Test	DC	AC	AC	AC
Lead Inductance (nH)	2.0 - 3.5	4.0 - 5.0	4.0 - 5.0	< 1.0
Footprint (Chip +)	20 - 100 mil	80 - 600 mil	80 - 600 mil	Clearance
Peripheral Bond Pitch	4 - 7 mil	3 - 4 mil	3 - 4 mil	10 mil
Area Array Bond Pitch	N/A	N/A	N/A	≥ 10 mil
Max I/O Count	300 - 500	500 - 700	500 - 700	> 1,000

Table 2-3 Chip Mounting Choices.

Notes:

- Cost based on high volume production of a nine chip module with over 2000 chip-tosubstrate connections
- Wire bond lead inductance for 1.0 mil Al wire with pad-to-pad spacing from 2.0 4.0 mm.
- Wire bond footprint depends on cavity or no cavity.
- TAB footprint depends on OLB pitch.
- Chip availability may be poor for high performance devices such as DSP, microprocessor and ASIC.

capabilities and product characteristics of the module. The chip mounting technique dictates what thermal path must be provided for the heat generated in the chip and, in many cases, dictates the type of substrate to be used. The chip mounting or chip connection choice also has heavy implications on module cost, chip availability, chip testability, module reworkability, module size and module performance. These issues, compared in Table 2-3, will be discussed for each of the major chip mounting choices.

2.4.1 Die Attach/Wire Bond

Die attach/wire bond chip mounting of semiconductor chips has been used for over 20 years. The chip is attached mechanically to the substrate by a variety of means including organic adhesives (silver filled epoxy is common) and metal eutectics (gold/silicon and various solders are common). The die attach material can be thermally and/or electrically conductive depending on the material selected. Then the chip is connected electrically, wire bonded, to the substrate by gold or aluminum wires. Figure 2-7a shows chips wire bonded in an MCM. Over years of process development, this technique has been thoroughly characterized and refined to its present production limit of about 4.8 mil pitch (a "mil" being a milli-inch and "pitch" referring to the center-to-center spacing of the wires). Figure 2-7b shows a portion of a chip wire bonded with 464 wires on a 4.8 mil pitch using 1.25 mil aluminum wire - a good example of state of the art production.

The die attach/wire bond technique minimizes MCM risk. High volume precision equipment is readily available in existing semiconductor assembly lines and experienced personnel are available. Yields (number of good wire bonds per number of wire bonds attempted), and thus costs, are predictable. All of this adds up to known variables and low risk.

Another unique advantage of die attach/wire bond is availability of ready to use chips. Bare silicon chips with wire bondable aluminum bond pads have been sold for years by semiconductor manufacturers to hybrid manufacturers. Techniques, specifications and applications information for electrical test, visual inspection, packing, shipping and assembling exist in a somewhat established infrastructure. It is important to note, however, that while many thousands of devices are routinely available in chip form, many state of the art devices such as microprocessors, digital signal processors and application specific integrated circuits (ASICs) are not. These devices are fully tested and sold typically in single chip package form - an important issue for MCM manufacturers.

In addition to pitch limitation, the die attach/wire bond technique has two major disadvantages for MCMs. The first is that die attaching a chip to a substrate complicates thermal management within the module. As shown in Figure 2-7a, heat generated by the chip must be conducted through the die attach material to the MCM substrate, to the package body and, finally, to the outside of the package. But most MCM substrates and package bodies are poor conductors of heat. They also may have CTE (the material property defining the of expansion or contraction at different temperatures, usually expressed in ppm/°C) sufficiently different from that of the silicon chip to create mechanical stress problems.



Figure 2-7a Die attach/wire bond.



Figure 2-7b State of the art wire bonding (1.25 mil Al wire, 465 L/O, 4.8 mil pitch).

msgalica@mintz.com Elm Exhibit 2162, Page 89

The second major disadvantage of using the die attach/wire bond technique for MCMs is that bare silicon chips cannot today be tested at-speed (normal operating AC frequency) because the large inductance of probe needles masks the test results. At-speed testing is normally done by the semiconductor manufacturer after chips are packaged. From a MCM manufacturing perspective, the scenario is grim: partially tested chips are assembled together in a module and then the entire module is tested at operating frequency. Any one chip that is not "up to speed" causes the entire module to fail test. Module yield is affected dramatically by the number of chips in the module. Even if the module design and test techniques are clever enough to locate a bad chip, the problem is compounded by the fact that die attach/wire bonded chips are difficult to remove and replace. Thus, the MCM manufacturer is faced with poor yields and little chance of rework. In addition, technology is not commonly available today to burn-in bare silicon chips; burn-in must be done at the module level (see Chapter 13).

The die attach/wire bond technique is often cited for its low cost - about \$0.25 for die attach and approximately \$0.01 per wire bond. But as we have seen, the total module manufacturing costs could be quite high due to low yields. This technique is suited best for modules consisting of a few, low power, inexpensive chips and an inexpensive substrate. Wire bond techniques are discussed in Section 9.3.

2.4.2 TAB

Tape automated bonding (TAB) is a method of electrically and mechanically connecting a chip to the substrate that has many possible advantages over conventional die attach/wire bond techniques for MCMs. The tape is an etchedout piece of metal consisting of tiny beam leads. The end of the leads connecting to the chip are typically on a small pitch and are fanned out to a larger pitch at the end that connects to the substrate. Chips are first bonded to the tape (inner lead bond or ILB) with each chip pad connected up to a beam lead by thermocompression (heat and pressure) bonding. The ILB unit is then excised from the tape and die attached to the substrate using conventional means described in Chapter 9. As a last step, the other end of the beam leads are bonded, using heat and pressure, to the substrate (outer lead bond or OLB). There are many techniques and metallization schemes for accomplishing inner and outer lead bonding. From a manufacturing point of view, the two main types are gang bonding and single point bonding. In gang bonding, all leads are connected at the same time by means of a hot bar thermode. The process is fast and uniform, but lead planarity is a difficult issue. In single point bonding, each lead is connected individually by means of a hot point or laser bonder. Figure 2-8a shows a schematic diagram of chips TAB bonded in a MCM.

TAB has been developed in the United States primarily to handle more chip connections at tighter pitches than wire bonding. TAB works well for high lead count, small pitch chips where wire bond yields become unpredictable and costly. Production TAB ILB bonding is done today at 4 mil pitch and has been demonstrated in the laboratory at 3 mil pitch and less. Figure 2-8b is a photograph of a 385 mil square chip with 360 pads TAB inner lead bonded at a 4 mil pitch. The beam leads are fanned out for outer lead bonding at a pitch of 8 mil. The large fanout is required for two reasons. First, tight pitch bonding requires leads to be very precisely coplanar and this is less likely to be the case by the time the unit reaches the OLB process step. And second, the substrate or board technology limits the pitch capability of pads on the substrate. If an advanced substrate is used, then non-fanout TAB (where the OLB pitch is the same as the ILB pitch) can be used.

TAB offers several advantages when used to bond chips to substrates in MCMs. Foremost is that ILB units can be tested at ac operating speeds. This means that bad chips can be culled and only known good chips need be assembled into the module, thus dramatically increasing module yields. The value in testing chips before module assembly increases with the number of chips in the module. In addition, the type of chip in a module can increase or decrease the need to test them prior to assembly. For example, ASIC chips typically are not speed sorted at final test; they are designed to fall within a range of specified performance. A module containing all ASIC devices can be expected to perform properly even without at-speed testing of the chips prior to assembly. On the other hand, high speed memory components are speed sorted at final test and graded for sale. A module containing these devices could not be expected to yield well at final test unless the chips had been at-speed tested prior to assembly.

A related advantage of TAB is that inner lead bonded chips can actually be burned-in (operated at-speed and sometimes at elevated temperatures for a period of time to cull out bad units) prior to module assembly. Such practice would probably eliminate the need for burn-in at the module level, reducing the cost of scrapping or reworking finished modules.

Although TAB offers the advantages of assembling only known good chips and the capability of bonding large complex chips, the disadvantages of using TAB in MCMs are numerous. Foremost among these is cost. Two factors can make TAB expensive: cost of the TAB tape and the number of difficult manufacturing process steps. Complex double metal layer (one metal layer acts as the ground plane to reduce lead inductance) tape can run as high as \$30 in a 35 mm format. Single metal layer tape is significantly cheaper - approximately



Figure 2-8a TAB.



Figure 2-8b State of the art TAB (360 I/O, 4 mil pitch).

msgalica@mintz.com Elm Exhibit 2162, Page 92

\$4 in a 35 mm format. Once the chip is inner lead bonded, it cannot be reworked or the tape salvaged. A typical TAB manufacturing flow consists of multiple difficult processes: inner lead bonding, testing, burn-in, testing, die attaching and outer lead bonding. Each of these processes can be low yielding on complex components, and together, they can add up to unacceptably low yields and high cost.

Another disadvantage of TAB is that die attaching a chip to a substrate complicates thermal management within the module. The problem here is identical to that discussed in Section 2.4.1.

Finally, TAB technology often means long leads and large component footprints. Even moderate OLB pitch can easily mean long beam leads, with corresponding high inductance. A very long beam lead may require an integrated ground plane to reduce lead inductance. At this level, TAB tape becomes very difficult and expensive to manufacture. Another disadvantage of TAB is that OLB at reasonable pitches can mean very large device footprints. In one case, a 15 mil OLB drove TAB lead length to over 880 mils. Large footprints mean the electrical signal must travel greater distances, something MCMs are designed to avoid.

In general, TAB technology is best suited for large MCMs containing complex and expensive chips and substrates. For this type of application, the advantages of testability and burn-in at the ILB stage probably outweigh the disadvantages.

2.4.3 Flip TAB

Flip TAB, sketched in Figure 2-9a, is a variation on TAB where ILB bonded units are mounted face down on the substrate for OLB. Flip TAB has all the characteristics, positive and negative, of regular TAB, with two major differences. First, in flip TAB, heat generated by the chips can be removed from the backside of the chip to a module lid, thus simplifying substrate design. And second, flip TAB is easier to repair than regular TAB since no die attaching is done. Instead the inner lead bonded chip simply rests on a spacer material and is held in place by the OLBs as shown in Figure 2-9b. The mainframe computer module shown in Figure 2-13 is another good example of non-fanout flip TAB.

2.4.4 Flip Chip

Flip chip mounting offers the best possible electrical connection between chip and substrate because it eliminates leads altogether. As sketched in Figure 2-10, solder bumps are placed in an array pattern across the chip. The chips are

Chip Flip TAB Lead Support Material Substrate

Figure 2-9a Flip TAB chip mounting.



Figure 2-9b State-of-the-art flip TAB.

mounted face down on the substrate and the solder is reflowed (heated to the melting point and then cooled) to form the connection. The chips can be placed as close together as 10 mils with no additional space needed for connections. The technique offers the best performance and smallest footprint at potentially the lowest manufacturing cost.



Figure 2-10 Flip chip mounting.

Flip chip electrical performance is superior to the other techniques because the interconnect length is extremely short - usually only about 50 - 100 μ m, thus, the inductance is extremely low, as shown in Table 2-3. This can be a significant factor if designing a module for high speed applications.

Small chip footprint is also a performance plus for the flip chip technique. Small footprints mean shorter distances between chips on the module and shorter signal propagation delay. (Propagation delay is the time it takes a signal to leave the output buffer of one chip, travel across the substrate and enter the input buffer of another chip). Product miniaturization is one of the market drivers for MCMs. Obviously, small footprints also mean smaller modules.

From a manufacturing point of view, flip chip has several advantages. There are few process steps and those steps are batch rather than individual. Additionally, because the solder bumps can be placed in an array pattern across the entire surface of the chip, the pitch of these bumps need not be extremely small. As the number of signal connections increases on complex chips, wire bond and TAB techniques must continually strive for smaller and smaller pitches. Once developed, the flip chip technique should be the highest yielding and lowest cost of all.

Finally, flip chip, like TAB, offers the ability to probe test chips at-speed prior to mounting in the module. This means that bad chips can be culled and only known good chips need be assembled into the module, dramatically increasing module yields. The value in testing chips before module assembly increases with the number of chips in the module. In addition, the type of chip in a module can increase or decrease the need to test prior to assembly.

Also, like TAB, a flip chip mounted IC can be removed from the module and replaced should a defect be identified. This repair process is accomplished by locally heating the faulty die to reflow the solder and remove the die. The site is then refreshed by either reflowing or removing the existing solder.

Flip chip has been developed and utilized almost exclusively by IBM for over 10 years. The basic technology is well characterized and production

qualified within IBM, and almost universally not characterized or qualified outside of IBM. This fact gives rise to a host of infrastructure issues for the non-IBM module designer selecting flip chip technology. First and foremost is the lack of chips available with solder bump array connections. Basically, they do not exist outside IBM. The outside module designer is faced with procuring standard wire bond compatible chips or wafers and somehow converting them to flip chip compatibility. The problem is compounded by the fact that these chips will have peripheral connection locations, possibly at a very tight pitch. Conversion to a large pitch array pattern of connections, and then bumping them, is a formidable task. Simply asking semiconductor manufacturers to provide flip chip compatible chips may not work. Infrastructure is lacking at this level also. Semiconductor manufacturers do not have computer aided design (CAD) tools or rules for designing flip chips. They lack bumping processes and equipment, probe-test equipment, trained personnel, etc.

The disadvantages of flip chip clearly are related to infrastructure and economics rather than technical. Current efforts at IBM to disseminate the technology to the general industry may eventually alleviate these issues and flip chip could become the preferred interconnect for a wide range of MCMs. For today, the MCM designer must carefully weigh the risks and benefits of using flip chip for a specific application.

2.5 MODULE LEVEL CONNECTION CHOICES

Numerous choices exist for the connection of the MCM to the mother board, but, unlike the vast standards implemented for single chip packages, none presently exist for MCMs. Without industry collaboration, the recent broad activity in MCMs will result in many different package sizes and formats. Fortunately, a cooperative venture has been initiated for the standardization of MCM packaging sizes. A task force operating under the auspices of the IEEE Computer Packaging Committee recently proposed a series of standard MCM package sizes to the EIC/JEDEC Committee [12]. Until this is reconciled, many module sizes will continue to be fabricated. Some of them will adopt the existing single chip packaging formats and benefit from the existing standards implemented by such organizations as JEDEC. The following section provides the basis to better understand the different options that exist and the performance penalties imposed.

2.5.1 Peripheral I/O

Peripheral I/Os are the most commonly used single chip package connection arrangement. The appearance of many MCM designs have been styled to

resemble these single chip packages. By doing so the assembly and testing infrastructure can be utilized with little or no tooling modifications. The user does not know whether the package contains multiple chips or a single IC without dissecting the package. These packages can be implemented with virtually all of the MCM substrate types stated in Section 2.3, but are most prevalent in MCM-L where leadframes can be easily laminated to the module or plated through-holes can be used as castillation joints to form leaded and leadless modules, respectively. If leaded, the leads can protrude from one or all four sides. For most applications, the most efficient way to distribute the I/O warrants the use of all four sides. The leads are trimmed and formed to OFP specifications. If no leads are required, metallization is placed on the underside of the module body to facilitate direct soldering. Although this style of package utilizes less space because the solder joints are on the underside of the package perimeter, inspection of the solder joints by conventional means is difficult. Conversely, the lead adds inductance and contributes to the propagation delay of the package. For high frequency applications, leadless modules are preferred.

In both cases, as the number of I/Os terminating from the module increases, either the lead pitch decreases or the module size increases. As the lead pitch decreases, the ability to reliably solder the module to a circuit board becomes progressively more difficult. As this pitch is reduced, soldering defects, such as bridging (solder flows between adjacent I/O and forms an electrical short), also increases. In contrast, if the package body is increased, more space of the mother board will be required for the attachment process. Thus, the use of perimeter I/Os is self limiting. Typical I/O pitches found are 0.050", 0.025" and 0.020", and I/O counts can reach several hundred. For increased interconnection density, one must consider the use of array I/O connections.

2.5.2 Pin Grid Array

Pin grid arrays (PGAs) can be utilized as a solution to overcome the geometrical constraints imposed by the peripherally terminated modules. By distributing the leads on the full area of the underside of the package, the module is more efficiently used. Figure 2-11 demonstrates the module I/O plot versus the area of the package. For the same pitch, significantly more I/Os exist in the PGA format. One hundred mil, 70 mil staggered and 50 mil pitches are commonly used. Package lead counts from several hundred to over 1000 have been fabricated. These pins can be press fit and soldered onto the MCM, as with plastic packages, or brazed in place, as with ceramic. The approximate cost of adding pins ranges from \$0.01 to \$0.02 per I/O. The PGAs can be through-hole mounted, placed into sockets or the pins may be surface mounted and soldered directly to the board. In many high performance applications, surface mounting



Figure 2-11 Array package efficiency.

will be required. In these cases, as with any leaded module, the inductance of the lead must be taken into consideration. For applications where this inductance is too great, area array formats such as pad array carriers should be considered.

2.5.3 Pad Array Carrier (PAC)

Pad array carriers (PACs) are often referred to as area array packages, land grid arrays (LGAs) or surface mount arrays (SMA). They are similar to PGAs with the exception that the pins are replaced by metallized pads. This package offers all of the efficient I/O termination benefits of the PGA without the negative attributes associated with lead inductance. Additionally, costs are typically less than PGAs since no leads are required. Attachment to the motherboard is through a socket or by a direct soldering process known as C5 (Controlled Collapse Chip Carrier Connection). In this process, solder bumps are deposited onto the metallized pads on the underside of the module and on the mating pads on the motherboard. The PAC then is placed onto the board and reflowed. As with most surface mount applications, small misalignments in component placements are remedied during the reflow cycle by the natural wetting characteristics and surface tension of the solder. This phenomena contributes to a high yielding process.

2.6 GLOBAL MATERIALS AND MANUFACTURING CONSIDERATIONS

2.6.1 Cost

The choice of materials and manufacturing processes greatly influences the final module cost and its suitability for a particular application. Because many MCM technologies are new (such as thin film substrates, flip chip connections and TAB), capital equipment start-up costs can be very high. The immaturity of these technologies also means low initial yields. Although capital costs and initial low yields can easily undermine a start up MCM development effort, this chapter is concerned with cost comparisons based on a full-scale production situation.

The largest contributions to cost for MCM production are module final test yield, substrate cost, and chip cost. Module final test yield is based on the expected yield of the individual components (chips plus substrate) plus an estimate of the damage done by the assembly process. Expected yield of the chip components is a tricky subject, depending heavily on the type of testing done at probe (in wafer form) and at final test for a particular device. For example, a RAM (random access memory) chip is normally tested only for basic functionality at probe. Then the RAM is packaged and sorted at final test for speed and temperature performance. As a last step, the packaged RAM is burned in to cull early life failures. Unless the die is speed and temperature sorted at probe and burned in at wafer level, which is not the normal practice today, the speed, temperature, and burn-in fallout occur at module final test and module burn-in.

If the circuit does not contain redundancy, the module can be considered a "series system" [13]. The expected yield of a series system is the product of the individual expected yields of the components. For example, consider a module containing three different semiconductor chips with expected yields of 95%, 90%, and 97%. The substrate expected yield is 95% and the assembly related yield is estimated at 99%. The expected module final test yield is then:

$$.95 \times .90 \times .97 \times .95 \times .99 = 78\%$$
.

It is easy to see that the module yield quickly deteriorates as a function of the number of components and the expected yield of each of those components. Yield can be greatly improved and module costs dramatically reduced by testing module components - chips and substrate - prior to assembly.

For a given chip set, module yield will dominate total costs for small plastic MCM-L type modules where substrate costs are low and module repair may be difficult or impossible. If a repairable technology is used along with an inexpensive substrate, the silicon cost is the major contributor to total cost. At the other end of the spectrum, substrate cost may be the largest contributor to total cost for large, high end, high power ceramic or MCM-D/C modules where module repair is practiced [14].

Substrate cost is driven by three major factors: capital equipment required to fabricate the substrate, the number of module substrates that can be obtained from a single process unit (such as a ceramic wafer) and substrate yield. More mature technologies such as thick film hybrid and wire bond will have the lowest capital equipment costs and the highest substrate yield. Larger process units (such as $12" \times 18"$ PWB panels) will lead to lowest individual substrate costs. Substrate yield is substantially influenced by basic manufacturing approaches, such as parallel versus sequential processing. In parallel processing (most commonly used for PWB and cofired ceramic type substrates), substrate layers are individually fabricated, inspected and yielded. The layers are then stacked to form a single substrate.

2.6.2 Electrical Performance¹

Electrical performance can be affected significantly by choices made in the areas of signal interconnect and chip, MCM substrate and module level connections. Thermal performance of a package can also affect electrical performance, most likely through degraded performance at excessive junction temperatures. Other than thermal issues, the chief performance factors to be considered during package selection are:

- 1. Material properties of insulator layers and interconnect conductors
- 2. Transmission line parameters such as impedance and crosstalk
- 3. Package parasitics (such as pad capacitance and lead inductance)

Ideally, in any electronic system, the signal leaving the output of some transmitter should arrive, with no changes, at the input of some receiver. However, signals experience attenuation (loss of amplitude) and distortion (noise)

¹ Contributed by Eugene Heimbecher.

as they travel through a packaging system. Loss is caused by material properties and noise is caused by numerous factors broadly broken into transmission line concerns and package parasitics. Noise control considerations generally lead to package designs which are large, difficult to manufacture, and expensive. Manufacturing and cost factors, as well as size and weight considerations, usually lead to package designs with poor electrical performance. Balancing these factors against one another is essential for successful MCM system design.

Insulator and Conductor Material Properties

Insulator layers effect electrical performance through two parameters. These are dielectric constant (ϵ_r or K) and loss tangent (tan δ), also known as dissipation factor (DF). Dielectric constant, ϵ_r , is a key factor in determining most transmission line parameters; both ϵ_r and tan δ contribute to what is called dielectric loss in insulator layers. Dielectric loss, like conductor loss (below), results in signal attenuation. Generally, for MCMs with short line length and frequency content below 1 GHz, dielectric loss is not of concern. For these cases, loss due to resistivity of interconnect metal almost always masks dielectric loss. At frequencies much higher than 1 GHz, certain materials must be avoided because of their high dielectric loss. At microwave frequencies, preferred materials are ceramic or teflon; fiberglass epoxy (FR-4) is to be avoided.

The high dielectric constant of a material in and of itself is not necessarily a problem. It is not an automatic limitation to high frequency usage. There are usually undesirable consequences to using high dielectric constant materials for signal interconnect structures. These are primarily: increased signal propagation delay and increased physical size (for identical transmission line parameters and performance).

Interconnect material (such as copper) affects electrical performance primarily through internal resistance resulting in signal losses. (Voltage drop due to these losses is usually negligible in signal interconnect, but may be significant in power and ground lines. This is one reason for use of wide planes to distribute power and ground). Resistance is generally broken down into two categories. These are standard DC resistance and high frequency AC resistance due to skin effect. Skin effect occurs because alternating currents do not flow evenly throughout the cross section of a conductor, but tend to flow only near the outer surfaces. For example, it is a good approximation to say that, for pure copper, a current with a frequency of 300 MHz only flows in the outer 0.15 mil of the conductor cross section [15].

DC resistance is calculated using standard equations; AC resistance is calculated by a ratio of full cross section to skin effect area [16]. Line resistance is not always important. For example, MCM-L conductors are on the order of 3 - 4 mils wide and about 1 mil thick and both DC and AC losses are usually



Figure 2-12 Effect of line loss.

negligible. On the other hand, MCM-D conductors may be on the order of 0.4 \times 0.15 mil and losses need to be carefully assessed. Measured data (Motorola) of some typical MCM-D substrates show DC resistance on the order of 3 - 5 Ω /cm for copper interconnect and 15 Ω /cm for aluminum.

Large line resistance causes a low pass filter (RC) type roll off on signal edges (Figure 2-12). This roll off is small near the sending end of the line and becomes progressively more pronounced as the signal propagates down the line. The edge degradation is beneficial to noise generation, but can be a significant problem on long lines with critical timing requirements, for example clock distribution lines. Also, the signal rise and fall times may be slowed down so much that CMOS devices become subject to spurious oscillations. The MCM system designer must consider these issues carefully.

Transmission Line Parameters

The understanding of transmission line theory and its application to MCM design is a text in itself [16]. However, it may be sufficient to say that at higher frequencies and faster edge speeds it always is necessary to maintain the best possible controlled impedance environment for both signal lines and power and ground distribution. For optimum performance in high speed, high frequency systems, signal interconnect always needs to be done in a transmission line manner (such as stripline, with good AC ground and reference planes both above and below the signal). Additionally, voltage distribution should always use full planes well coupled to ground references. A near ideal approach is available in the MCM-D/C. Signal lines can be embedded in low dielectric constant deposited stripline layers with optimized signal performance while power and ground return planes can be contained in a high ε_{r} , very thin (and hence highly capacitive) layer in the ceramic base. With sufficient power plane capacitance, it would not be necessary to use discrete decoupling capacitors.

Dielectric constant is important in two ways. It affects both transmission line impedance and speed of signal propagation. As an example, for a given stripline geometry, if the impedance is 50 Ω and the delay is 2.03 ns/foot ($\varepsilon_r =$ 4.0), doubling the dielectric constant without changing any other factor results in impedance decreasing to 35 Ω) while delay increases to 2.87 ns/foot. (Note the square root of two relationship.) The situation is not as clean for microstrip lines or other multimedia line geometries, but the general trend of decreased impedance and increased delay (as ε_r increases) always holds.

For several reasons it is desirable to have transmission line characteristic impedance (Z_0) fall into the general range of 50 - 150 Ω . Line impedance is generally a complex function of several parameters, but it can be increased (from 35 - 50 Ω) by a combination of making line width smaller and/or making the vertical distance between the signal lines and ground planes larger. When dielectric constants are large, as for ceramic, designs must use either thinner line widths or thicker insulator layers than for designs with smaller dielectric constants, such as polyimide. There are two problems with trying to use thinner lines. First is the fact of running into the manufacturing process limit, as discussed in previous sections. Second is a parameter tolerance issue: a line width of 10 mils ±1 is easier to make consistently than a line width of 2 mils ±1. Thus, the usual strategy is to increase the thickness of the insulator.

The chief physical parameters for controlling impedance are: dielectric constant and line width and vertical distance between signal line and ground plane. It is recommended that a full dimensional analysis of all parameters of all layers in a substrate stack up be performed to identify the worst case maximum and minimum impedances to guarantee acceptable electrical performance. Also, it is often desirable to separate or isolate certain signals from other signals. This can result in the need for additional routing layers as compared with a design with no routing restrictions. These extra layers are undesirable from a manufacturing and cost point of view, but are necessary to obtain required electrical performance.

74 MCM PACKAGE SELECTION

Crosstalk is the remaining transmission line parameter to be discussed. Crosstalk is the unwanted coupling of energy from an active (signals occurring) section of some circuit or system to some other unconnected and nominally quiet (no signals) section. This can result in a receiver seeing an input signal that is, in fact, only noise. Crosstalk can arise form a number of sources:

- Electro-magnetic interference (EMI)
- Power and ground plane noise
- · Capacitive affects when signal lines cross over each other
- Mutual coupling when signal lines run parallel to each other

Generally, this last mechanism is considered the most significant source of crosstalk in well designed systems. (But crossover affects can be significant if several or more occur along the length of the victim line.)

For stripline, maximum crosstalk is not affected by the dielectric constant of the insulator material. If ε_r is doubled, all other factors being unchanged, there will be no change in crosstalk. However, if the dielectric thickness is increased (to increase line impedance from 35 - 50 Ω), then crosstalk will increase due to the reason discussed below. Therefore, with higher ε_r materials, to obtain comparable electrical performance, the line separations must be larger and the overall physical size of the design will probably increase.

There are a number of strategies for reducing crosstalk, but the most useful is simply to keep things far apart. Microstrip transmission lines are more prone to crosstalk problems and should be avoided in noise sensitive designs. When microstrip is used, separating two lines by a distance equal to 8H (where H is the vertical distance between the bottom of the signal lines and the top of the ground plane) reduces crosstalk to about zero. Stripline is a more effective approach for lines in a noise sensitive application.

As an example, consider an MCM-D/C design, as mentioned above. If the distance between ground planes is about 40 μ m (1.6 mils), striplines separated by only 80 μ m would have no crosstalk. In practice, it is found that even a separation of 40 μ m is enough to achieve close to zero crosstalk. On the other hand, for an MCM-L design using microstrip, the situation can be much worse. For example, to guarantee no crosstalk in a structure with a dielectric thickness of 4 mils, signal lines would have to be separated by over 30 mils. Thus, there is a 10 to 1 routing density advantage in MCM-D (stripline) as opposed to current MCM-L (microstrip) for identical levels of crosstalk.

Crosstalk does not occur only in MCM substrates. A potential problem in flip chip MCM design is redistribution of I/O pads. Most ICs are not designed for flip chip. Hence, the typical procedure is to take a standard chip, with I/O pads on the periphery, and use an extra metal layer to redistribute the I/O pads over the surface of the chip. This must be done with some care. It is possible to route a very noisy output over the top of some very sensitive input. This is an issue that is not well understood currently.

Crosstalk is an issue to be considered in package selection as well. Three primary factors to reduce chip connection and packaging crosstalk are: to keep leads as far apart as possible, to keep parallel leads as short as possible and to use as many ground leads as possible (mixed in with signal leads) to reduce coupling between signal leads.

2.6.3 Thermal Path

Heat produced within semiconductor chips will dissipate through the MCM package and out to ambient air and to the board on which the module is mounted. If heat is not dissipated efficiently, the chip junction temperature (T_j) rises beyond the reliable operating range and operating life is decreased significantly.

Like single chip electronic packages, module thermal performance is measured in terms of the difference in temperature between the chip junction and ambient air per watt of power produced by the chips:

$$\theta_{ja} = (T_j - T_a) / W.$$

The module must provide thermal pathways that conducts heat expected from the chips mounted in the module. Heat is conducted through a combination of four general paths: chip to substrate to ambient, chip to lid to ambient, chip to leadframe to ambient or chip to substrate to heatsink to ambient. The chip to substrate to ambient path is most common in electronic packaging, but for MCMs this path can be complicated by a substrate with mediocre thermal conductivity and very dense circuitry. Designing structures to improve substrate conductivity, such as thermal vias and metal heat spreaders, usually reduces the routing channels available in the substrate. The designer is faced with a difficult tradeoff between routing density and thermal conductivity. The chip to lid to ambient path has been demonstrated on several high power MCM designs, but is workable only when the backside of the chip faces the module lid - flip chip or flip TAB mounting. Many low cost, single chip packages use the chip to leadframe to ambient approach. This path has limited thermal capability (35°C/W in still air) and may complicate the routing of signal interconnect on the module.

2.6.4 Rework

The choice of materials and manufacturing processes determines whether a module can or should be reworked; that is, a faulty chip removed from the substrate and replaced with a new chip. Rework is both a technical and business issue. The cost of the rework process must be less than the cost of a finished module, otherwise rework does not make economic sense. When considering the cost of rework, one must consider costs associated with designing fault location circuitry, isolating a fault, performing the repair, retesting the part and, perhaps, cycling through another rework iteration.

Module level rework begins with fault detection by either visual or electrical tests. Next, the faulty chip is accessed. The chip is demounted, normally by applying heat to the chip connection points. The substrate connection points are then prepared for a new chip, the new chip is mounted, the module encasement is restored and the module is retested. Materials and manufacturing choices are obviously critical at every step of rework. For example, visual detection may be difficult or impossible with flip chip or flip TAB type chip mounting. Electrical detection is successful only if the module circuitry is designed for such testing, using JTAG boundary scan, for example, as discussed in Chapter 13. Chip accessibility can be difficult or impossible in modules with plastic mold encasement or with certain lid attach schemes. Chip demounting is difficult or impossible in modules where a die attach is employed. Substrate connection point preparation is difficult in modules that contain tight pitch chip connection pads, or in those that rely on solder pads on the substrate.

2.7 MODULE DESIGN EXAMPLES

MCM materials and manufacturing choices should be optimized for a given application. In this section, we describe some real examples of MCMs employing various combinations of technologies to suit specific product requirements.

2.7.1 Mainframe Computer Module

Figure 2-13 is a photograph of an MCM designed and built by Siemens-Nixdorf Information Systems for a high speed mainframe computer. It is 4" on a side and contains bipolar ASIC and memory components. The MCM substrate is a multilayer organic laminate (MCM-L). Substrate manufacturing is additive and sequential. Each layer is visually inspected and repaired as it is fabricated [17]. Chip mounting is flip TAB using thermode gang bonding. An elastic spacer on

MODULE DESIGN EXAMPLES 77



Figure 2-13 Mainframe computer module.

the front or active side of the chip presses the back of the chip directly against an anodized aluminum module lid forming an excellent path for heat transfer. The module can handle more than 500 watts of power. The substrate is mounted in a plastic housing that contains the unique module level I/O, with over 1,000 signals connecting to the mother board. The module I/O is an array of holes in the plastic housing with bifurcated springs designed to accept a matching array of pins mounted on the mother board. The entire module is repairable.

This MCM design is optimal for its intended application of packaging a large number of high speed, high power, expensive bipolar chips. High cost can limit the use of this technology for the general market.

2.7.2 Workstation Module

The MCM sketched in Figure 2-14a is aimed at the performance and density driven CMOS/BiCMOS marketplace, particularly, the departmental computer,

communications, peripherals, workstation and PC segments. It is a cost effective, high density MCM developed to accommodate high speed complex ASIC logic devices, microprocessors and memories.

The package body and MCM substrate are combined in a single substrate made of a ceramic base with deposited thin film copper/polyimide interconnect layers (MCM-D/C) [18]. The substrate provides very high density routing with 20 μ m lines and 30 μ m spaces on two layers. The chip mounting is flip chip, with a thermal gel material providing the heat conduction path between the backside of the chips and the thermally conductive lid. The module I/O is a JEDEC standard 299 pin grid array or a space optimized 50 mil pad array carrier [19] as pictured in Figure 2-14b.

This module is well suited for its intended application. It is capable of connecting 15 to 20 complex chips in a very small area. The power handling capability of 20 - 30 watts is matched to the expected power dissipation of CMOS components. The module offers optimum electrical performance at a cost that serves the market for performance oriented systems.

2.7.3 Low Cost Module

A very inexpensive module commercially available today is pictured in Figure 2-15a. The module, which offers advantages in system miniaturization and system performance, is a direct extension of a single chip QFP package. It utilizes a multilayer PWB substrate (package body and MCM substrate are integrated (MCM-L) with an embedded leadframe that forms the module level peripheral connection (Figure 2-15b). The chips are mounted using standard die attach/wire bond techniques, and the entire assembly is overmolded with plastic.

The major limitations of the module are thermal dissipation and routing density. Heat is primarily conducted from the chip to the copper leadframe The package can dissipate up to 4 watts under forced convection at 500 feet per minute airflow. Routing density is limited primarily by very large vias - 0.8 mm on the internal layers and 0.5 mm on the outer layers.

The technology offers low cost and low risk because it uses a minimum number of inexpensive piece parts and mature processing technologies. It is generally applicable to consumer and communications products where small size and low cost are important.

2.7.4 Data Communications Module

The module shown in Figure 2-16 is a prototype developed for a data communication application. The module consists of two host microprocessors
MODULE DESIGN EXAMPLES 79



Figure 2-14a Schematic of workstation module.



Figure 2-14b Two versions of a six chip C4/C5 workstation module: thin film copper/polyimide on multilayer cofired ceramic pad array carrier (PAC) substrate. Each chip has 463 I/O flip chip connections.

80 MCM PACKAGE SELECTION



Figure 2-15a Low cost module.



Figure 2-15b Schematic of low cost MCM with integral leadframe.



Figure 2-16 Data communications module.

with associated memory, a proprietary chip set and various passive devices. A total of 18 ICs and approximately 45 passive devices are assembled to a $1.8" \times 1.8"$ four layer MCM-D substrate incorporating 75 µm lines and 30 µm vias. Die and wire bond assembly techniques are utilized for chip mounting. In this application MCM processing has made it possible to reduce the size from the original 30.0 square inch PWB to 3.24 square inch.

2.8 FUTURE TRENDS IN MCM MATERIALS MANUFACTURING

MCMs are a solution to the size, parasitic and path length effects of semiconductor packages and their interconnect. The multichip module (MCM) interconnection technology has been maturing as a packaging technology since the first articles were published in the early 80s. The first applications using MCMs used multilayer cofired ceramic substrate technology (HTCC). Low dielectric constant polymers with thin film conductor technology applied to

silicon, metal, alumina and cofired multilayer ceramic came later as more package and or chip I/Os were needed.

During this same period plastic quad flat package (PQFP) technology improved significantly with 0.3 mm pitch and up to 500 I/Os being planned for manufacture by 1993. PWB technology and MCM-L are also improving with 4 mil line and space and 6 mil via technology currently available and manufacturers talking about 2 mil line and space PWBs with 4 mil vias by the mid 90s.

With the exception of the very high end computer and military applications and the very low end consumer market, the use of MCM has been relatively modest. Most of the successful commercial users of MCMs have been vertically integrated mainframe computer companies. The successes have come from products derived from a system viewpoint, where the company has control of the subsystem and the components to achieve performance of the product at an acceptable cost.

If MCM technology is to approach the degree of acceptance in the marketplace to that of surface mount technology (SMT) today, issues relating to die availability, low cost substrate technology and the testing of bare chips and modules must be addressed.

An abundant supply and variety of good bare die must become available at an acceptable cost. Off chip performance will determine whether wire bond, TAB or C4 die configurations are suitable for use. As chip clock frequencies rise, the thrust will be to C4 die attach.

2.8.1 Substrates

Today, typical high density MCM-D substrates can cost anywhere from \$50 to \$100 per square inch, meaning that the substrate cost alone will deter many product designers from considering this packaging approach. A low cost MCM technology must evolve within the next several years if MCMs are to have a major impact across a broad spectrum of products. A high density MCM substrate cost below \$5 per square inch must be achieved. A hybrid version of current MCM-L and MCM-D processing is a potential solution. For example, high density signal layers containing small vias fabricated by advanced processing such as laser ablation can be laminated to low density power and ground planes fabricated by conventional PWB processing. Since there is a cost premium for using the high density layers, utilization should be confined to a minimum of signal layers or for the redistribution from high I/O chips to the circuit board. The cost of such a module is expected to approach that of conventional PWB and current MCM-L technologies.

2.8.2 Optical Multichip Modules (MCM-O)

MCM prototypes intended for use at system clock rates in excess of 3 GHz have been fabricated by conventional copper/polyimide processing. Design guidelines for these high frequency modules have been proposed [20]. A potential next step in the evolution of MCMs that could overcome many of the difficulties associated with high frequency, hard wired MCMs is the use of optics for chipto-chip interconnection. Optical transmission media exhibit terahertz bandwidth, immunity to electromagnetic interference and optical noninteraction. It has been demonstrated that properly designed thin film optical waveguides, physically intersecting with each other and transmitting optical signals, exhibit negligible crosstalk [21]. Therefore, chip-to-chip optical interconnect technology may not require a multilayer system as is required when using conventional, hardwired interconnections. The need for crossovers is eliminated or minimized. However, in practice, it is likely that hardwired and an optical layer will be used together, with the total number of layers significantly reduced from using hardwired alone.

Preliminary results have shown that high speed chip to chip optical interconnects, compatible with conventional MCM substrate processing, are feasible [21]. A semiconductor laser was coupled to a flip chip photodetector through a photolithographically defined polymer optical waveguide (Figure 2-17a). An infrared photograph of the interconnect taken during operation is shown in Figure 2-17b. Optical propagation is from left to right across a 2" ceramic MCM-D substrate. The resulting optoelectronic module is referred to as an optical MCM (MCM-O).

2.8.3 Test

Test methodology and technology to supply functional bare die and assembled MCMs must be developed to a point where testing is not perceived as an impediment to utilizing MCM technology. To assist in this area, standard MCM sizes and footprints (either perimeter or area array) need to be established. In addition, test and burn-in sockets need to be available.

2.8.4 Thermal Control

Thermal solutions for modules which provide for reliable thermal control at reasonable cost need to be developed. For high power applications, C4 die or flip TAB can offer the lowest thermal resistance path (back of die is exposed for heatsinking) without complicating the substrate layout, quite possibly making it the most viable packaging alternative for high power applications [23]. (See Figures 2.9 and 2.14).

84 MCM PACKAGE SELECTION



Figure 2-17a Schematic of a chip to chip optical interconnect.



Figure 2-17b MCM-O operating at 1 Gbit/s NRZ.

2.8.5 Environmental Concerns

Finally, more environmentally sound processes need to be developed. The elimination of many toxic solvents and heavy metals from the fabrication of laminate materials and subsequent assembly processes is becoming an issue. The

move to eliminate solvents, such as methylene dianaline (MDA) from the manufacture of organic laminates [24] and chlorofluorocarbon (CFC) containing materials, has already begun. In addition, federal legislation is pending that limits or eliminates the use of lead from electronic assemblies [25]. OEMs soon may be responsible for recycling all lead containing products from inception to retirement - cradle to grave. As a result the industry needs to focus on developing non-lead bearing solders and conductive z-axis adhesives and cements.

Acknowledgments

The authors acknowledge and appreciate the ongoing support of J. W. Stafford and S. Bowen. We also wish to thank the numerous staff members of the Motorola Corporate Manufacturing Research Center and ASIC Division for review, discussion and contribution to this manuscript.

A special thanks and acknowledgment to Eugene Heimbecher for his contribution of Section 2.6.2.

References

- 1 J. L. Sprague, "Multilayer Ceramic Packaging Alternatives," *IEEE Trans. on Components, Hybrids, and Manuf. Tech.*, vol. 13, no. 2, pp. 390-369, June 1990.
- 2 U. Deshpande, S. Shamouilian, G. Howell, "High Density Interconnect Technology for VAX-9000 System," *Internat. Electr. Packaging Soc.*, IEPS, (Marlborough MA), Sept. 1990.
- 3 Design Guidelines Multilayer Ceramic, Kyocera Corporation.
- 4 Design Guide: Developing Customized Microelectronic Packaging for High-Performance Applications, *Dupont Electronics*.
- 5 M. R. Cox, L. Ng, "An Economic Comparison of High and Low Temperature Cofired Ceramic," *Inside ISHM* pp. 31-33, Jan./Feb. 1992.
- 6 S. J. Stein, R. L. Wahlers, C. Haung, M. Stein, "Interconnection and Packaging of Advanced Electronic Circuitry," *Proc. ISHM*, (Orlando, FL), pp.130-134, Oct. 1991.
- 7 N. Iwase, K. Anzai, K. Shinozaki, "Aluminum Nitride Substrates Having High Thermal Conductivity," *Solid State Techn.*, pp. 135-138, Oct. 1986.
- 8 D. J. Arthur, "Advanced Fluoropolymer Dielectrics for MCM Packaging," First Internat. Conf. on Multichip Modules, (Denver CO), pp. 270-276, April 1992.
- 9 P. Fischer, "Materials for a Parallel Approach to High Density Interconnects," *Proc. IEPS*, (Marlborough MA), p. 103, 1990.
- 10 T. G. Tessier, W. F. Hoffman, J. W. Stafford, "Via Processing Options for MCM-D Fabrication: Excimer Laser Ablation vs. Reactive Ion Etching", Proc. 41st Electronics Components Techn. Conf., (Atlanta GA), pp. 827-834, May 1991.
- 11 R. C. Eden, "Applicability of Diamond Substrates to Multi-Chip Modules," Proc. ISHM, (Orlando FL), pp. 363-367, Oct. 1991.

86 MCM PACKAGE SELECTION

- 12 J. W. Balde, "Proposed MCM Standard Sizes A Report of the IEEE Task Force", *Proc. NEPCON West*, (Anaheim CA), pp. 467-478, Feb. 1992.
- 13 R. A. Dovich, *Reliability Statistics*, Milwaukee, Wisconsin: ASQC Quality Press, pp. 21-35.
- 14 A. Krauter, J. Baumann, M. Becker, "Assembly Process for a New Packaging System," *Siemens Research and Development Reports*, Bd. 17, 1988, No. 5, Springer-Verlag.
- 15 J. Hardy, *High Frequency Circuit Design*, Virginia: Reston Publishing Company, 1979.
- 16 H. B. Bakoglo, Circuits, Interconnects and Packaging for VLSI, New York: Addison-Wesley Publishing Company, 1990.
- 17 H. Brosamle, B. Brabetz, V. Ehrenstein, F. Bachmann, "Technology for a Microwiring Substrate," *Siemens Research and Development Reports*, Bd. 17, 1988, Nr. 5 Springer-Verlag.
- 18 E. G. Myszka, A. Casey, J. Trent, "A Multichip Package for High Speed Logic Die," Proc. 42nd Electronic Components Techn, Conf., (San Diego CA), pp. 991-996, May 1992.
- 19 J. Trent, G. Westbrook, "Fine Pitch Pad Array Carrier Sockets for High Speed Logic Die," *IEEE Multichip Module Conf.*, (Santa Cruz CA), pp. 40-43, March 1992.
- 20 B. K. Gilbert, W. L. Walters, "Design Options for Digital Multichip Modules Operating at High System Clock Rates," *First Internat. Conf. on Multichip Modules*, (Denver CO), pp. 167-173, April 1992.
- 21 C. T. Sullivan, "Optical Waveguide Circuits for Printed Wire-Board Interconnections," Optoelectonic Materials, Devices, Packaging and Interconnects, SPIE vol. 994, pp. 92-100, 1988.
- 22 K. W. Jelley, G. T. Valliath, J. W. Stafford, "1 Gbit/s NRZ Chip-To-Chip Optical Interconnect," *Photon. Tech. Lett.*, to be published.
- 23 R. Darveaux, I. Turlik, "Backside Cooling of Flip Chip Devices in Multichip Modules," *First Internat. Conf. on Multichip Modules*, (Denver CO), pp. 230-241, April 1992.
- 24 F. Klimpl, "Basics of Multilayer Matter," *Electronic Engineering Times*, Issue 681, Feb. 1992.
- 25 C. Melton, A. Skipor, J. Thome, "Manufacturing Process Issues of Non Lead Bearing Solder Pastes," *Internat. Conf. on Solder Fluxes and Pastes*, (Atlanta GA), May 1992.

MCM PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Paul D. Franzon

3.1 INTRODUCTION

There are a large number of packaging alternatives available to the design engineer today. This range is not likely to narrow in the near future. The object of this chapter is to provide a framework of understanding for making packaging decisions with the perspective of how best to satisfy the needs of an electronic system.

The design decisions made in any engineering venture are driven by cost and performance. Fundamentally, packaging acts to limit performance and to increase cost. Recent trends in integrated circuit technology suggest that system performance is being limited increasingly by the package. This has resulted in heightened attention to new, more highly customized forms of packaging used to improve system performance. Multichip modules (MCMs) represent a class of packages used to obtain significant improvements in system performance compared with conventional forms of packaging. Highly customized advanced packages are expensive when compared with off the shelf mass produced single chip packages. It is important to realize when advanced packaging is appropriate and when it is not. A primary aim of this chapter is to identify the nature of the tradeoffs involved and to suggest a decision-making process. The aim is not, however, to provide the reader with complete models for that decision-making. Many of the later chapters provide these models.

88 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Fundamentally, there are three types of silicon found in a digital system. First, we find the very large scale integrated (VLSI) logic chips with over 100,000 transistors. Such chips range from general purpose microprocessors to special purpose application specific integrated circuits (ASICs). ASIC styles range from full custom and semi-custom designs to chips whose function is programmed in the manufacturing line. Second, we find the so called "glue chips," the off the shelf chips that provide functions not integrated into the VLSI chips. Today, more and more of the glue logic is being collected into and thus, is being replaced by ASICs. Typically, the only glue chips found are the drivers required to drive large loads and long lines and the receivers and latches that often are at the other end. Finally, we find the memory chips, which are often the most numerous [1].

The need for advanced packaging is driven by the trends in the design and use of these three types of chips. Today's leading CMOS microprocessor chips often contain over 1,000,000 transistors and are clocked at frequencies in excess of 150 MHz. The transistor count and clock speed are expected to continue to grow at a rapid rate, as shown in Figure 3-1. Leading edge VLSI chips create tremendous demands on the package. These chips have high I/O counts with 500+ I/Os being typical for 1992 RISC microprocessors. As the on-chip circuits become faster, the inter-chip package delay, not speeded up by using faster circuit technology, becomes dominant. One of the requirements for reducing this delay is that the chips be placed closer together. The increasingly fast signals produce lots of electrical noise in the package unless the package is carefully designed. The large number of fast circuits also produces lots of heat that must be removed from the system. For example, the DEC Alpha CPU chip dissipates over 30 W when running at a clock speed of 200 MHz.

The purpose of this chapter is to present a framework for making packaging decisions as part of the system design process. The next two sections set the stage for this by describing the system design process and defining the concept of the packaging hierarchy. Following that, the factors through which packaging decisions affect system performance and cost are presented and discussed. A process is then described showing how these performance and cost factors are used to make packaging decisions. An example is given of this decision process.

3.2 SYSTEM DESIGN PROCESS

The phases that make up the design process are summarized in Figure 3-2. In the system specification phase, the system requirements and goals are determined. How this is done is discussed toward the end of the chapter. Most of this chapter is concerned with the high level design phase, in which the



Figure 3-1 Microprocessor clock speed and the number of transistors per die are both growing at an increasing rate. (Adapted from [16].)

organization of the system, and the technologies to be used, are decided. This is described further in Section 3.6. In the low level design phase, the actual circuits are designed schematically. These are turned into layouts of the chips and packages during the prefabrication stage. A layout describes exactly where each transistor and part is placed and where each wire is run. Aspects of these phases are discussed in **Part B** - **The Basics**.



Figure 3-2 The system design process. (Adapted from a chart prepared by Ken Drake of MCC.)

3.3 THE PACKAGING HIERARCHY

Obviously not all systems fit onto one VLSI chip, one MCM or one printed wiring board (PWB). Thus, multiple levels of packaging are needed so that multiple chip packages can be interconnected. This is referred to as a packaging hierarchy, an example of which is given in Figure 3-3. At the lowest level of the hierarchy, the chips are mounted in single chip packages or MCMs. The next level of the hierarchy usually consists of PWBs (sometimes referred to as cards). The PWBs in Figure 3-3 are then connected together via a backplane (also referred to as a back panel, or sometimes board if the PWB was called a card). Backplane to backplane connections are made via a rack, and the racks then might be connected by some means, and so on. This is a very common hierarchy for larger systems though it is by no means the only one available.

Gate to gate interconnections (nets) might not go through this packaging hierarchy (for example, an on-chip interconnection) or might have to go through one or more levels of packaging. As any one net spans more levels of the hierarchy, the length of that net, and also the signal delay, increases substantially. Ideally, the interconnections and connections provided by the packaging hierarchy would be matched to the interconnection needs of the system. For example, Figure 3-4 shows how an electronic system could be considered as a set of interconnected functional blocks. In this case, the interconnection requirements of the system map naturally onto the packaging hierarchy shown previously in Figure 3-3. The nets that need to be short and fast are kept to the low levels of the hierarchy while nets that can be long span all of the levels of the hierarchy.

This ideal situation is rarely met. Most systems require more short nets than usually is provided by the packaging. The physical limitations of the packaging hierarchy (primarily determined by the desire to keep the sizes of individual packages and connectors down to control cost) force a number of these nets to go through more than one level of packaging. Unfortunately, the number of connections that go between package levels is limited by the capacity of the



Figure 3-3 The packaging hierarchy.

connectors between levels (and high capacity connectors are expensive). If there are not enough paths available through these connectors then it might be necessary for several nets to share the same path (a bus or multiplexed path).

Interconnect Topology:



Map Interconnect Topology onto Packaging Hierarchy:



Figure 3-4 The interconnection needs between functional modules in a system often vary across a system. The packaging hierarchy should be structured so as to match these needs as closely as possible.

Thus, in many systems, a large number of nets take a double hit - they must span multiple levels of package and they must share a path with other nets. The job of the designer is to select the package technologies to minimize the number of nets that must run through the high levels of the package, and then to select

those nets carefully. A tremendous advantage of MCMs over conventional packaging is that they allow more of the nets to be concentrated in the first two levels of the hierarchy, thus improving the system performance.

There are several alternatives to the hierarchy style illustrated in the bottom of Figure 3-3. For example, sometimes the backplane might be bypassed with dedicated cable connections coming from the other side of the board. This can be extended even further to the situation where most of the interboard connections are dedicated wires. For example, this is done in Cray computers. A slight variation on the hierarchy shown in the bottom of Figure 3-3 is to have a relatively large motherboard with smaller daughterboards mounted on it. This is a common hierarchy in desktop and portable computer systems. It also is used to package the IBM 3081 computer central processing unit. In this case, the main board is $60 \text{ cm} \times 70 \text{ cm}$ in size and the daughterboards are MCMs, each containing 100 chip sites [2]. Another hierarchy under active investigation is the use of connections in the third dimension to stack MCMs.

3.4 PACKAGING PERFORMANCE FACTORS

This section describes the factors that relate performance to packaging technology choices. Usually, each performance factor is broken down into the elements that determine it. In some cases, figures of merit used to differentiate package choices are discussed. A figure of merit is a number that attempts to summarize the packaging factor. The usefulness of a figure of merit depends on the accuracy needed to make a decision. Often a full model or evaluation is needed to make a decision.

3.4.1 Size and Weight

Size and weight are often specified as performance goals in many portable and aerospace systems. In the latter this is a primary driving force towards the application of MCMs (see Chapter 15).

A size restriction leading to the use of advanced packaging also might arise artificially. In any system, the size and possibly the weight of each subsystem only is estimated early in the design process. Errors might be made or system goals might change later while detailed design is in progress. In this case it might become necessary to use advanced packaging in a sub-system in order to avoid a complete redesign.

A size limitation might be an area or volume limitation or some mix of the two. For example, in a notebook computer, the height of the computer is limited by the height of the disk drive. Disk drive manufacturers are driven to package the electronics portion of the drive in as small a height as possible, sometimes just a fraction of an inch. On the other hand, a few millimeters of height on the main computer board is not as important as the total area of the board.

If a size limitation is expressed as an area limitation, then a suitable figure of merit for evaluating different packaging approaches is the substrate efficiency. This is defined as the percentage of the substrate covered by silicon [3]. For example, if a 6 cm \times 6 cm MCM has 10 0.8 cm² die placed on it, the substrate efficiency is given as 10 \times 0.8 / (6 \times 6) = 22%. Another useful metric is the number of gates per unit of substrate area. These figures sometimes are evaluated on a volumetric, rather than an area, basis.

Particular attention also must be given to the size and weight of the power supply (battery) and the mechanical structures used to remove heat. This might involve a compromise between the desire to use a small system to house the electronics and the need for a large heat removal structure required by a small system with a high heat density.

By replacing multiple single chip packages with an MCM, substantial size and weight reductions are achieved. The closest comparable single chip package solution would be to use ASICs in surface mount packages on a PWB. The magnitude of the reduction possible with an MCM solution depends, in part, on the interconnection capacity needed.

3.4.2 Interconnection Capacity Within Each Level

Interconnection capacity refers to the total amount of wiring provided within a level of the packaging hierarchy. This wiring is provided by layers within the package devoted to signal interconnections. The physical layout of the wires is determined by a routing Computer Aided Design (CAD) tool. Additional layers are usually used for distributing power and ground. Once the details of the parts to be interconnected are known, the package interconnection (or routing) capacity requirement is expressed in the following form [4]:

Required Interconnection Capacity
$$\approx RN_{net} P/E$$
 (3-1)

with the result expressed in units of length. In this equation, E is the efficiency with which the available interconnect capacity can be used (often called routing efficiency and typically takes a value of around 50% [4]), R is the average length of each net (interconnection) in terms of chip pitch, N_{net} , is the number of nets and P is the average chip pitch, the average distance between chip centers.

The available interconnection capacity is given by:

Available Interconnection Capacity
$$\approx \frac{\left(\sqrt{N_{chip} - 1}\right)^2 P^2 \times Number of signal layers}{Average wire pitch}$$

(3-2)

plus any capacity around the periphery of the board. Here $\mathrm{N}_{\mathrm{chip}}$ is the number of chips.

A figure of merit often used to compare different packaging technologies is the interconnection density:

Average Interconnection Density =
$$\frac{\text{Number of signal layers}}{\text{Average wire pitch}}$$
 (3-3)

Figure 3-5 shows how interconnection density is calculated for a single layer. Figure 3-6 gives a plot of interconnection density and cost per unit area for different interconnection technologies. Wiring capacity is then given by:

Available Interconnection Capacity
$$\approx \left(\sqrt{N_{chip} - 1}\right)^2 P^2 \times Average Interconnection Density.$$

(3-4)

If the available capacity is less than the required capacity then, short of redoing the high level design, there are six choices:

- 1. Share (multiplex) signals onto the same interconnections, for examples, reduce N_{net} . This is often done on backplane busses. If it is likely that several functional modules need the same physical line at the same time then performance degrades.
- 2. Increase the number of signal layers thus increasing weight, size and cost. Typically, the total number of layers increases at a rate almost one and a half times the rate of increase of the number of signal layers. In a high speed system each signal layer has to be next to a power or ground layer (see Section 3.4.4 and Chapter 11). If through-hole components are used on a PWB, the maximum number of layers is limited to 10 or 12 (usually equivalent to a maximum of 8 signal layers).

96 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE



Figure 3-5 How interconnection density is measured.

- 3. Increase the chip pitch, P, and the size of the board or substrate. This then increases weight, size, cost and delay.
- 4. Spend additional design time to try to increase the routing efficiency, E, by manually, rather than automatically, routing the nets.
- 5. Reduce the number of chips by using an ASIC.
- 6. If the via size is larger than the wire width, thus consuming area that could have been used for wires (see Figure 3-7), then reducing the via size will allow an increase in interconnection density. This is common in PWBs and laminate MCMs.

In general, the finer the average wire pitch and the greater the total interconnection density, the less likely that these undesirable steps are required. In particular, thin film MCMs rarely have insufficient interconnection capacity. The average wire pitch is not always the same as the minimum wire pitch, typical values for which are given in Table 3-1. There are two reasons for this. First, in high speed systems, wires need to be pitched further apart to control crosstalk noise. This is discussed in Chapter 11. Second, via size and style can have a dramatic impact on average interconnect density.

If there are insufficient via sites on the board or MCM, it might be difficult for a router to find a via where it needs one. This is referred to as via

PACKAGING PERFORMANCE FACTORS 97



Figure 3-6 Substrate cost (1992) versus interconnection density. (Adapted from [18] with updated prices based on typical 1992 vendor costs.)

starvation. As a rule of thumb, if there are fewer than 1.5 to 2 via sites per signal pin, then the router efficiency, E, decreases [4]. This is particularly a problem with the through hole vias commonly used in PWBs since the number of via sites does not increase as the number of signal layers increases. With high layer count boards and laminate MCMs, buried vias are required to compensate for this. The use of large PWB and laminate MCM vias also can result in via

98 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Table 3-1 Typical Minimum Line Pitches.

TECHNOLOGY	MINIMUM WIRE PITCH	MINIMUM VIA PITCH
Cofired Ceramic MCM	150 µm - 250 µm	250 μm
Thin Film MCM	20 µm - 40 µm	20 μm - 200 μm
Laminate MCM	100 µm - 250 µm	1000 μm - 2500 μm
PWB	250 µm - 300 µm	1000 μm - 2500 μm

starvation. Referring again to Figure 3-7, it can be seen that between every row of vias there are several wires (two or three are typical). This low ratio of wires to vias might also make it difficult for a wire always to find a via where it needs one. In thin film and ceramic technologies the via size can be the same as the conductor size. Buried vias are the norm. Thus, the effects of vias on interconnection properties in these technologies are usually minimal.

Interconnection properties interact with the choice of connection techniques used between levels of the packaging hierarchy. For example, if an edge style of connection is used (such as wire bonding or TAB for the first level connection) then the pads on the chip might have a center to center pitch as small as 75 μ m (though 150 μ m or more is typical). If the pitch of the wires on the substrate (Table 3-1) is larger, then the chip connect function must include fanout for pitch matching, as shown in Figure 3-8. Making room for this fanout increases the minimum possible chip pitch, P. This is one important reason why conventional packaging tends to have poor substrate efficiencies. Providing this fanout to match a 0.35 - 0.5 mm PWB surface pad pitch makes high pin count surface mount packages large (while a small pin count package, such as a memory, is small). Providing fanout often is necessary even on laminate and cofired ceramic MCMs. It is not necessary on a thin film MCM.

On the other hand, with area connection (such as flip chip solder bump), the pad or pin distance on the chip must be equal to or larger than the minimum via spacing in the package. Providing this fanout to match a 0.1" PWB via pitch makes high pin count pin grid arrays (PGAs) large. The signal layers underneath the chip then are used to bring these signals out from underneath (escape) to be routed to other chips at the required pitch. This consumes some of the interconnection resources underneath the chips. It is important to ensure that sufficient interconnection resources remain underneath high pin count PGAs. IBM solves this problem on their flip chip MCMs by assigning special layers (redistribution layers) for the escape function in the Thermal Conduction Module. These layers also perform test and rework functions. Again, the impact is minimized if thin film layers are used.



Figure 3-7 One effect of via size on wire pitch.



Figure 3-8 Fanout often is required to pitch match the chip pads and the package pads.

3.4.3 Connection Capacity Between Packaging Levels

The connection capacity between packaging levels is much smaller than the interconnection capacity within a level. For example, consider an MCM connected to a PWB. The pin pitch of the connector that goes between the two package levels typically is limited to either the wire pitch of the PWB on one layer only, or to the through-hole via pitch of the PWB, usually the latter. The capacity implied by either of these pitches is much less than the capacity, over a similar cross section, of either the PWB wiring or the MCM wiring. Furthermore, the cost of large high pin count connectors can be very high. (See Chapter 10 for a discussion on MCM to PWB connectors generally and Chapter 18 for a discussion on cost.) This is why richly interconnected systems tend to use large MCMs and large PWBs to minimize the use of connectors and backplanes. By flattening the packaging hierarchy in this way, the bottleneck created by the low capacities of the higher levels of hierarchy and the connectors to them is minimized. For example, the IBM 3081 main CPU board contains nine 9 cm \times 9 cm MCMs placed on a 70 cm \times 60 cm PWB. However, a large MCM is more expensive to produce than a collection of smaller MCMs. There is a tradeoff between substrate size and connector capacity.

Performance often is compromised to reduce connector capacity requirements between different levels of the hierarchy. For example, the only significant difference between the Intel 386DX microprocessor and the Intel 386SX microprocessor is that the latter provides fewer pins on the single chip package for the memory interface. This reduces the effective performance of the latter but does make it less expensive.

An important factor determining the connection capacity between levels is whether an edge or area connector is used. For example, Figure 3-9 plots the total number of pads versus pad pitch for both area and edge connectors for the IC to MCM interface. With area connection, more connections are made with reduced manufacturing tolerance than with edge connections. This is a major advantage for flip chip with solder bump technology. It is also a major advantage for PGAs and other area array single chip and multichip packages in comparison to peripheral pinned packages such as surface mount. The subject is discussed in more detail in Chapters 10 and 18.

The above discussion assumes the required connection capacity between levels is known. However, often it is not known and must be estimated. A good estimation of the required signal pin capacity, N, for the signals leaving any level of packaging is given empirically by Rent's rule:

$$N = KM^{p}$$
(3-5)



Figure 3-9 Number of I/O pads for a 1 cm^2 chip with peripheral and edge arrangements.

where K is a constant of proportionality, typically about 2.5, M is the number of circuits contained within the lower level of the hierarchy, and p is Rent's constant (typically 0.5). However, it should be noted that K, M and p are determined empirically. Rent's constant tends to decrease with large increases in the number of circuits. (Consider the number of signal pins leaving a personal computer box versus the number leaving a CPU chip; they are roughly comparable indicating a sharp decrease of p with M.) It should be remembered that N does not include the power and ground pins. In high performance systems there can be almost as many power and ground pins as signal pins to control electrical noise.

102 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

For modest increases in the number of circuits, M, contained within a package, the number of pins required, N, usually increases. Thus, as further transistor miniaturization occurs, there is an increased requirement for interlevel connectivity. Interlevel connections become an important technological constraint on the package performance, as discussed in Chapter 18.

3.4.4 Delay and Electrical Noise

Delay refers to the time required for a signal to travel between the functional circuit blocks in a system. To a first order approximation, packaging-related delay is broken up into the following contributions (Figure 3-10):

 $t_{delay} = t_{buffer} + t_{flight} + t_{rise-time-degradation} + t_{noise-settle}$ (3-6)

where t_{buffer} is the delay incurred within the buffer-amplifier, t_{flight} is the time taken for the signal to travel (fly) along the wire at nearly the speed of light, $t_{rise-time-degradation}$ is the extra delay incurred because of an increase in rise time of the signal (the time for the signal to transition between the two different logic voltage levels) on the rise time as compared to the signal at the input of the buffer. $t_{noise-settle}$ is the extra delay that must be incurred while waiting for electrical noise to settle.

If the buffer is small and weak, then t_{buffer} is small but $t_{rise-time-degradation}$ is large. The delay $t_{rise-time-degradation}$, is determined mainly by the ability of the buffer output to charge and discharge the capacitances associated with the interconnection. Part of this capacitance comes from the chip attach leads and $t_{rise-time-degradation}$ improves as the chip attach leads get smaller. Thus smaller buffers are sometimes used in chips intended solely for MCM use. If the line is lossy (resistive), then $t_{rise-time-degradation}$ is increased further. Thus it is desirable to have low loss lines.

The minimum time for the signal to travel down the line, the time of flight, is given by:

$$t_{\text{flight}} = \frac{l}{c/\sqrt{\varepsilon_{\text{r}}}}$$
(3-7)

where *l* is the length of the interconnect, c is the velocity of light in a vacuum and ε_r is the relative dielectric constant of the insulator. Typical values for



Figure 3-10 Electrical delay.

dielectric constant are given in Chapter 2. In a high speed system, total delay time is limited primarily by the time of flight. This is a driving force behind using packaging types that allow parts to be more closely spaced (reduce l) and for using dielectric materials with lower values for dielectric constant, such as glass-ceramic for cofired ceramic MCMs, polyimide for thin film MCMs and cyanite ester or polyimide for laminate MCMs and PWBs, as discussed in Chapter 1.

Whenever transitions occur on digital signals $(0 \rightarrow 1 \text{ or } 1 \rightarrow 0)$, electrical noise is introduced both on the wire (connection or interconnection) carrying the signal and on the wires around it. This noise contributes to the package parasitics

defined in Chapter 1. In digital systems, noise considerations indirectly affect system performance. For example, Figure 3-10 shows how increased noise is equivalent to an increase in delay on data lines for digital systems. For clock signals and analog signals, excessive noise will directly compromise correct system function. This is discussed further in Chapter 11.

There are three major sources of noise within a digital system: reflection noise, crosstalk noise and simultaneous switching noise. The system also produces noise that affects the operation of systems around it (and itself is susceptible to such noise produced by other systems). This is referred to as electromagnetic interference (EMI) noise. The magnitude of all of these noise sources depends on the rise time, t_{rise} , of the signal. The faster the rise time, the worse the noise.

Reflection noise is a potential problem if the time of flight becomes comparable with the rise time

$$t_{\text{flight}} > t_{\text{rise}}/5 \rightarrow t_{\text{rise}}/3.$$
 (3-8)

When this is the case, the signal edge needs to see a constant impedance as it travels along the line. Whenever the impedance changes, part of the signal is reflected just as part of a light signal is reflected when it encounters a sheet of glass.

The characteristic impedance, Z_0 of a wire is given by:

$$Z_{o} = \sqrt{L/C}$$
(3-9)

where L is the inductance per unit length of the line and C is the capacitance per unit length. Maintaining a constant (termed controlled) characteristic impedance requires that L and C remain constant along the length of the line. Doing this requires that the signal line maintain a constant cross sectional geometric relationship with a reference line or plane, either ground or power. Examples of controlled impedance lines are given in Figure 3-11. The most common approach is to use microstrips and striplines, creating the need for reference planes. Typically, offset striplines are used instead of striplines wherein there are two signal layers between each pair of reference planes. Thus in most digital systems there are typically about half as many reference planes as signal planes (reference layers can be shared by signal layers).



Figure 3-11 Examples of controlled impedance lines.

Also, if a matching terminating resistance, $R = Z_0$ is not placed at one end of the line, then part of the signal is reflected there. This reflection travels up and down the line, interpreted by the receiver as noise. Without a matching

termination, reflection noise can be controlled by keeping the line short (see Chapter 11):

$$t_{flight} < t_{rise}/4.$$
 (3-10)

Crosstalk noise arises whenever signal lines or chip connect leads run parallel to each other. The signal on the active line couples onto the quiet line as noise. The faster the rise time, the greater the coupled noise. Crosstalk noise is controlled by placing the lines further apart than the minimum line spacing and, sometimes, by limiting the coupled length. Thus, the greater the potential impact of crosstalk noise, the lower the interconnection density is. Chapter 11 discusses how crosstalk noise considerations also lead to the use of reference planes and allow lines to be more closely spaced when the dielectric constant is reduced. One advantage of cofired PGA packages over most plastic packages is the provision for these reference planes within the package.

Simultaneous switching noise occurs whenever a large number of off-chip or on-chip drivers switch at the same time, as shown in Figure 3-12. This switching activity causes a large current spike to flow through the ground and V_{CC} connections. When this current spike flows through the inductance associated with the ground and power circuits a noise voltage will appear on the chip's internal power and ground lines. The magnitude of this noise is given approximately as:

$$V_{\text{SSN-noise}} = L_{\text{eff}} N \frac{dI}{dt}$$
 (3-11)

where L_{eff} is the effective inductance of chip ground or V_{CC} connection, N is the number of switching drivers, and dI/dt is the current transient produced by each buffer. The effective inductance, L_{eff} , also can be substantially reduced through the use of shorter chip attach leads and by placing ground planes beneath the leads. As a rule dI/dt increases with decreasing rise time. This noise causes false switching inside the chip, appears as noise on the output leads of any quiet buffers connected to the ground or power rail and increases delay in the switching drivers. One major advantage of MCM technology is that the effective inductance is greatly reduced in comparison to single chip packages. This is particularly true for solder bump and multimetal TAB attachments (TAB with a ground plane).

For the same set of chips, all of these sources of internal noise are usually smaller in an MCM package in comparison to PWB packages. In particular, the shorter connection distances reduce reflection noise as well as the accumulation of crosstalk noise, while the smaller chip connection inductances reduce the amount of simultaneous switching noise. Noise is easier to control in MCMs and its impact on delay is smaller. However, noise usually can be adequately managed in single chip packages. Many packages, particularly ceramic PGAs, provide internal reference planes or ground and power planes for the purposes of controlling noise. Though few plastic packages currently have these planes, some are starting to make limited use of them.

EMI is produced by the package circuits whenever current flows within them. The connections act as antennas producing radiated noise. Meeting required standards for EMI noise can be difficult and often requires the use of metallic screens in the box. One advantage of MCM technology is that it reduces the size of many of these antennas and, potentially, the need for screening.

It is important to note, however, that as transistor speed increases, rise time decreases and noise control becomes more difficult. The subject of electrical delay and noise control is discussed further in Chapter 11.

3.4.5 Power Consumption

Power consumption usually impacts system performance through its indirect impact on size and weight. For example, in a notebook computer, weight considerations dictate the battery size and energy. Thus, to make a computer that lasts four hours on one battery means carefully controlling the power consumption. Similarly, in aerospace systems, increasing power also means increasing the weight of the power supply. In telecommunications systems, power consumption must be controlled so that the batteries required to power the system in the event of a power failure are not too large. Controlling power consumption also reduces power dissipation (the two have the same value), thus reducing the need for complex and large heat removal structures.

Whenever a 0-1-0 transition occurs on an interconnection, the amount of energy consumed in charging and discharging the interconnection capacitance is given by:

$$Energy = CV^2 \tag{3-12}$$

where C is the total capacitance of the interconnection and V is the voltage swing. The power consumed is this energy per transition times the number of

108 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE



Figure 3-12 Simultaneous switching noise in digital systems.

transitions per second. Therefore, in any CMOS circuit where capacitive energy is the main form of power consumption, the power consumption increase is directly proportional to frequency of operation.

As the capacitance, C, is proportional to the length of the interconnection and the size of the chip attach, one advantage of MCM technology is reduced power consumption. For example, let us say that 10% of the system power consumption is due to the interconnect when mounted on a PWB and that migration of the product to an MCM reduces the power consumption due to the interconnect by a factor of five. Then the MCM product either would consume 8% less power than the PWB-based product or could be allowed to run at a faster rate of transitions per second (frequency) with no increase in power consumption.

3.4.6 Heat Dissipation

Consumed power is converted directly to dissipated heat. As the clock frequency and the number of transistors per chip increase, the heat produced by the chip increases. This heat causes the temperature of the chips to rise. With CMOS chips, higher temperatures affect system performance directly by slowing the transistors. In any system, high temperatures decrease the reliability. To maintain reliability, other performance factors might need to be compromised to improve heat removal. The structures used to remove heat might add considerably to the manufacturing cost. Thus, thermal issues are very important in MCM design because the heat density is much higher than in single chip packages, making cooling more difficult.

Heat is removed mainly by conducting it away from the chips and allowing it to convect into a circulating coolant. There are generally two choices for the heat conduction path: through-the-substrate or directly off the back of the chip. Also there are generally two choices for the coolant: air, often forced by a fan(s), or water (or another other liquid such as a fluorocarbon), often forced by a pump. Some possible combinations are shown in Figure 3-13. Unfortunately all of these alternatives involve some compromise in either another performance or cost factor.

Generally it is considered that through-the-substrate forced air cooling is the cheapest alternative. (Hence its very aggressive use by DEC, as described in Chapter 17.) An air cooled system has a number of advantages over a water cooled system. It reduces the need for expensive plumbing and seals. An air cooled system also is more likely to survive a fan failure than a water cooled system would a pump failure. By making the parts more accessible, maintenance and field repair costs are reduced. Through-the-substrate cooling tends to require less precise mechanical engineering than chip backside cooling and further helps in making the parts more accessible for repair.

The use of through-the-substrate forced air cooling might require some performance compromises. First, its use requires a highly conductive thermal path through the substrate. Unfortunately, the materials with the lowest dielectric constant such as glass-ceramic or polyimide tend also to have the lowest thermal conductivity (Tables 7.1 and 12.1, for example). This can be overcome by using either copper slugs, commonly called thermal vias, beneath the chip or sinking the chip into the substrate (Figure 3-14). This substantially reduces the capacity beneath the chip, particularly for laminate MCMs where the slug consumes all

110 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE



Figure 3-13 Some of the alternatives that can be used to cool an MCM.



Figure 3-14 Thermal vias and thermal cutouts.

of the area beneath the chip. (Alternatively, an array of through-hole vias can be used, but their efficiency is limited.) The resulting lack of interconnect capacity might force an increase in chip pitch, P, and thus MCM size. Second, use of through-the-substrate cooling usually means that only edge connectors can be used to the next level of packaging, potentially reducing connection capacity. Third, even with the use of thermal vias, the heat density might be too high for the desired cooling mechanism. It is then necessary either to reduce heat density by forcing the chips further apart or to consider a more aggressive cooling approach.

When individually packaged, high power chips generally must be housed in ceramic or metal packages, or in plastic packages with direct ceramic or metal heat paths to the heatsink. The thermal tradeoffs between single chip and multichip packaging can not be easily summarized. While heat density is higher in the latter, one (larger) heatsink can be shared by multiple chips. The advantages of the multichip packaging relative to single chip packaging depends on the details.

3.4.7 **Performance Tradeoffs**

A number of performance to performance tradeoffs have been identified above. Increased interconnection density is the main driver to increased system

112 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

performance. Increased density allows the chips to be spaced closer together, thus decreasing size, weight and delay. It also allows more chips to be placed in the same area, avoiding the bottleneck posed by the board to backplane connectors. The increased density offered by MCMs primarily benefits systems containing chips with hundreds of I/Os. It is erroneous to think that the main advantage of MCMs is only that they eliminate single chip packages. Repackaging a system with high I/O count chips as bare die on the same PWB results only in a small size reduction due to interconnection density considerations.

Sometimes it is necessary to sacrifice wiring density to satisfy thermal requirements if thermal vias become necessary. This should be weighed against the alternatives of using backside cooling or spreading the chips apart.

3.5 PACKAGING COST FACTORS

In this section, the different factors that contribute to system cost are described and explained. Details about how to model cost are provided in Chapter 4. In this section the intent is to explore their relationship to system level decision making by describing some of the more important cost performance tradeoffs. It must be noted that the unpackaged chip costs also must be included in the production cost. For any one board or module, the chip costs often exceeds the package cost if large, leading edge chips are used. The opposite is true if simple chips are used. When a multiple board system is considered, total packaging cost often exceeds total chip cost (see Chapter 18).

3.5.1 Production Cost

Production cost is the cost involved in getting the product out the factory door to the purchaser. It has two elements, manufacturing cost and the manufacturability cost.

Manufacturing Cost

The manufacturing cost is the cost of materials and process steps associated with the production of each part in the system and their assembly. This includes the chips, packages, connectors, heat removal mechanisms, power distribution features and the final packaging (casing, etc.). It includes the recurrent engineering (RE) cost elements (a cost that recurs for every part made) such as the purchase of the required materials and the labor and energy required to produce each part. It also includes the cost of purchasing the manufacturing equipment, a nonrecurrent engineering (NRE) cost element (a cost that is spread over a number of parts). A technical cost model for these elements is described in Chapter 4.

PACKAGE TYPE	TYPICAL COST
Plastic package Ceramic PGA < 144 pins Ceramic PGA > 144 pins	5¢ - \$5.00 \$5,000 NRE + 10¢ per pin (\$14 or more for 144 pins) \$25,000 NRE = 10¢ per pin (\$50 or more for 500 pins)
Cofired ceramic MCM Thin film MCM Laminate MCM PWB	\$3 per sq. inch per layer (\$30 per sq. inch for 10 layers) \$60 per sq. inch (expected to decrease to ~ \$20) \$3 - \$5 per sq. inch < \$1 per sq. inch
CMOS chip wafers	\$25 - \$150 per sq. inch

Table 3-2 Manufacturing Costs for Packaging and Interconnection Elements.

High volume production allows the cost of purchasing the manufacturing equipment to be distributed over a greater sales volume, minimizing the NRE cost apportioned to each part. The required manufacturing and assembly steps should be as simple as possible and should lend themselves to automation through the use of standard looking parts. For example, the fact that small MCMs often can be mounted in conventional packages (QFPs, PGAs, etc.) gives them an additional cost advantage over large MCMs that tend to require custom packages unusable by standard automatic assembly equipment [5].

Typical vendor prices for some lower level interconnect structures are listed in Table 3-2. It can be seen that high pin count PGAs, high layer count ceramic MCMs and thin film MCMs command a substantial price premium over plastic packages, PWBs and laminate MCMs. Not shown are the cost of connectors. High pin count connectors tend to be costly (see Chapter 18). The higher cost of using high density MCMs at the lower levels of the packaging hierarchy might be compensated for if the need for high pin count connectors at higher levels of the hierarchy were reduced.

Manufacturability Cost

Not all of the parts manufactured function correctly. Parts must be tested, repaired if possible, and retested after repair. The cost of test and repair and the impact of failures is referred to herein as manufacturability cost because they depend, in part, on how well the part was designed for manufacturability.



Figure 3-15 A simplified view of the MCM manufacturing process.
Assuming that this test and repair step is done only once in the manufacturing process (as shown in Figure 3-15), the final unit cost becomes:

Final unit cost = (Manufacturing cost per part + Test cost per part + (1 - Initial yield) × (Repair cost per part + Cost of retest per part)) / Final yield

where each cost component is averaged over all parts, good, bad or repaired.

Yield is defined as that percentage of parts that pass a test phase. Final yield is greater than initial yield if some of the failed parts can be repaired. For a chip, MCM or PWB, yield decreases with size. Thus, the cost premium of increasing a part size by a factor of two might be factor of eight if doubling the size doubles the manufacturing cost and quarters the yield. However, the greater effective interconnection capacity of the larger MCM often compensates for the poorer yield.

For example, consider die yield. The number of dies (chips) manufactured per wafer is given by:

dies per wafer
$$\approx \frac{\pi \times (\text{wafer diameter}/2)^2}{\text{die area}} - \frac{\pi \times \text{wafer diameter}}{\sqrt{2 \times \text{die area}}}$$

- # die sites used for process control

and the die yield is given by:

die yield = wafer yield
$$\frac{1}{(1 + \text{defect density} \times \text{die area}/\alpha)^{\alpha}}$$
 (3-14)

where wafer yield is the percentage of wafers not containing a gross fault affecting every chip on them, and α is the defect clustering parameter, which tends to take on a value between 1 and 3 [6]. The defect density might run anywhere between 0.5 - 2 per cm². Defect density improves with process maturity. For a fixed wafer manufacturing cost (typically \$500 to \$2000), the die cost increases rapidly with chip area. If a large complex chip has a yield of only 10%, then its final cost is at least ten times its manufacturing cost. This is one reason why large chips are not always the best alternative to a small MCM. The presence of defects effectively limits the maximum size of chip that is manufacturable without some method of tolerating the faults produced. As chips usually cannot be repaired, initial yield and final yield are the same.

msgalica@mintz.com Elm Exhibit 2162, Page 145

116 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Testing and rework costs have been identified as critical MCM related cost issues. Many sources attribute one-quarter to one-half the cost of the final MCM to these subcosts. In the normal manufacturing flow (Figure 3-15) individual parts are tested before assembly and the assembled MCM also is tested. If the assembled board fails, then it is necessary to either scrap the board or rework it, that is find and replace the part that failed. Since most MCMs contain at least one high value chip reworking usually is preferred. However, the cost of reworking an MCM is generally high. The best way to minimize rework is to ensure that chips are fully tested before being mounted on the MCM. Currently, this requires that the die be TAB mounted for reasons explained below. TAB mounting itself is expensive and also consumes area on the MCM.

Thus a balance must be struck that optimizes the combined cost of test and the likelihood of rework. This balance is determined by the minimum of the total out the door cost. A simplified expression for this cost (based on the manufacturing process in Figure 3-15) is:

Final cost =
$$\frac{MC + TC + (P(TE) + P(AF)) \times (RC + ATC)}{Final yield}$$
 (3-15)

where MC is the total IC and MCM manufacturing cost, TC is the total IC and bare and populated MCM test cost, P(TE) is the test escape probability, P(AF) is the probability of an assembly fault, RC is the rework cost and ATC is the cost of retesting the assembled substrate.

Test escape refers to an IC that passes its initial test but fails after assembly. This is a potentially significant problem for MCMs. The hardware used to probe the very small chip pads uses long non-controlled impedance leads, making it difficult to pass noise free, fast edge signals to the chip. Good design for test techniques, as discussed in Chapter 13, are needed. At the moment, the easiest way to test a chip at full speed is to mount it first in a package such as a PGA, QFP or TAB. If the chip is not tested at full speed, the test escape probability can be as high as 30% if the IC is a leading edge CMOS chip, but also can be very small if the IC is from a mature line. It also is higher for CMOS parts than for bipolar parts. This is discussed further in Chapters 13 and 18. Depending on the details of this combined cost, there are a number of options that can be considered:

• If the MCM contains no high value parts or has only one high value part likely to fail, then there is no need to fully test the ICs before assembly since the MCM can be scrapped at little expense, if it fails.

- If the MCM has only one high value part, that part can be placed by itself on the MCM where it is easier to test than in bare die form. If the part passes, then the MCM assembly is complete. If the part fails, then the MCM can be scrapped or the single die replaced. This approach reduces the risk that other parts are damaged during rework.
- If the MCM contains a number of high value dies (chips) likely to have high test escape probabilities, then full consideration should be given to properly testing the dies before assembly, perhaps by using TAB chip attach and testing the tape form.
- Sometimes parts can only be tested properly after assembly. For example, a complex microprocessor may be easier to test completely when attached to its memories rather than as a single IC. In that case, the process should be optimized toward inexpensive and easy rework such as flip attach techniques.

3.5.2 Post Production Costs

Post production costs are incurred once the system is in use. They include maintenance, field upgrades and the repair and replacement of failed parts. The sum of production and post production costs is referred to often as the life cycle cost. The goal of a designer is to minimize life cycle cost, of which production cost might only be a fraction.

Reliability depends on the elimination of possible failure mechanisms and the operation of the parts at a sufficiently low temperature. This is discussed throughout the chapters in **Part B** — The Basics.

Repairability relates, in part, to the size of the field replaceable unit. For example, if an MCM is soldered onto a board, the field replaceable unit is the board. If the MCM is socketed, and the field diagnostics located the failed MCM, then the MCM is the field replaceable unit. If that MCM is sealed in epoxy, it must be scrapped. However, if the MCM has a resealable lid, it might be repairable and have further value as a used part. A life cycle cost analysis points at the correct solution (though MCMs are considered to be so reliable that repair is not usually needed).

The down time required to locate the fault and replace the unit also often is important. For example, in military systems, the ability of a system to survive a mission and to be quickly and easily checked and repaired are very important. If liquid cooling is used, then replacing a failed unit takes much longer than when air cooling is used. Also, the greater compactness of an MCM system might make fault location more difficult if the design is not carefully thought out. It is difficult to probe a signal on an MCM in the field.

118 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Later chapters pay particular attention to reliability exposures that come about through the use of specific MCM technologies and reliability results from actual MCM use (see Chapter 17). Particular attention should be given to establishing a quality engineering process so that reliability can be maximized.

3.5.3 Design and Prototyping Costs

Design and prototyping costs include engineer training, engineer time spent in design, purchase of CAE and CAD tools to help in the design, construction and testing of the prototype, followed by design changes that arise from testing. Particularly for small to medium production runs, design and prototyping costs can be a significant part of the final system cost. The following should be considered when estimating the impact of this factor:

- The cost of building and diagnosing faults in an MCM prototype is higher than for a PWB prototype. (In the PWB prototyping, signal lines can be probed easily to test the prototype.) Extra emphasis needs to be placed on using a design approach that results in first pass success of the prototype. This requires extra investment in computer design tools and engineer training. Note, however, this investment is not that much different than the investment required for achieving first pass success for PWB designs operating at similar speeds and power dissipation. Consideration should also be given to using rapid prototyping technology [7].
- Effort should be spent on learning the technology. One technique is to go through the entire design and prototyping cycle with a non-production part. One reason that laminates are currently a popular form of MCM is because of their similarity to the already familiar PWB technology.
- Existing designs and hardware should be reused as much as possible and reasonable. This involves using off the shelf parts (if available and suitable), using programmable or semi-custom logic (if suitable) and reusing portions of existing designs, if applicable, rather than creating new designs. This is very important for low volume commercial parts.
- The decision making process described in this chapter requires that models be built, data obtained and evaluations conducted. This is a design activity. The investment required to carry out this activity

should be balanced against the likelihood that more detailed models, data, etc. lead to better decisions being made.

One possible impact of using existing designs is that, for low volume parts, it might be preferable to gain a performance improvement through repackaging a set of chips as an MCM, rather than redesigning them on one VLSI chip. (In any case, the one chip alternative is not a viable option if the chip set contains a mix of technologies.)

3.5.4 Time-to-Market

Reducing time-to-market often is more important than controlling design, prototype and production costs. One survey showed that being six months late to market resulted in an average 33% profit loss for that product, while a 9% production cost overrun resulted in a 21% loss, and a 50% design development cost overrun resulted in only a 3% profit loss. In a recent survey, engineering managers stated that they would rather have a 100% overrun in design and prototyping costs than be just three months late to market with a product. There are a number of reasons for this. If your competitors beat you to the market with a comparable product, they gain considerable market share and brand name recognition. This is difficult to quantify. One example is the stronger market presence of the Nintendo games over the other newer electronic games simply because it was introduced first. An earlier market entry has more opportunity to improve yield, thereby improving profits. Also, the technology is usually locked into place early in the design process. The design and production must be completed quickly from this point to prevent competitors from introducing a similar or more advanced product at the same time. Finally, design and prototyping costs are an up front investment that must produce a return. The longer investors must wait for their return, the higher that total return must be. (Who would you invest in? The company that could double your investment in two years or the one that would take one year?)

There are many reasons why time-to-market might be longer than necessary. One reason products might be late to market could be organizational, that is, excessive delay in making key decisions and an over reliance on complex design methods. Another possible reason is inexperience. Learning a new technology introduces delays into the design and manufacturing cycles. Remember that you are also inexperienced in pushing the older technology to new limits. It is very difficult to get a conventional 250 pin QFP to run reliably at 150 MHz.

Another potential reason for a long time-to-market, particularly critical for some MCM technologies right now, is the lack of infrastructure, as discussed in

120 PACKAGE SELECTION: A SYSTEMS NEED PERSPECTIVE

Chapter 1. For example, you might need a source of tested, qualified solder bumped die and access to a high volume thin film MCM manufacturing line for success of your product. If both of these are difficult to obtain at the time of manufacture, product release must be delayed while you wait for their availability.

3.5.5 Cost Tradeoffs

The relative weight given to production cost, post production cost, design and prototyping cost and the impact of time-to-market depend on the details of the type of part being produced. For example, a part being produced in high volumes for use in applications where reliability is not critical, emphasizes production cost.

3.6 PACKAGING DECISIONS AND THE SYSTEM DESIGN PROCESS

During the system design process, the design engineering team needs to determine the organization of the system, the packaging hierarchy to be used, the packaging mixture to be used within each level of the hierarchy and the partitioning of the system functions between the chips and packages that comprise the system. The term "organization" means the block diagram for the system - what functional blocks make up the system and how they are connected. The term "packaging style" refers to the details of the package selected, including the type of package (QFP or thin film MCM), the size and layer count. It also includes the details of the connectors used between the packaging levels. The term "partitioning" refers to what functions are assigned to which chips and how the chips are assigned to different packages within the system. As part of the partitioning process, it might be necessary to determine a floor plan describing the approximate placement of components, chips or packages, with respect to each other.

Packaging and partitioning decisions are made by evaluating a set of alternatives. The performance and cost of each alternative is estimated and compared against the requirements and goals of the system. To do this, the requirements and goals of the system must be expressed in terms of the performance and cost factors described above and summarized in Table 3-3. If the performance and cost of each alternative is similarly expressed, then the alternative that best matches the goals of the system is the preferred option. This